UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended March 31, 2011

Commission File Number 001-08106



Florida 65-0829355
(State or other jurisdiction of incorporation or organization) (I.R.S. Employer Identification No.)

800 S. Douglas Road, 12th Floor, Coral Gables, FL

(Address of principal executive offices)

33134 (Zip Code)

Registrant's telephone number, including area code: (305) 599-1800

during the preceding 12 i	mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of months (or for such shorter period that the registrant was required to file such reports), and (2) he so days. Yes \boxtimes No \square	_	4
required to be submitted	mark whether the registrant has submitted electronically and posted on its corporate Web site, if and posted pursuant to Rule 405 of Regulation S-T ($\S232.405$ of this chapter) during the preced was required to submit and post such files). Yes \square No \square	• • •	
	nark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer ccelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Exchange		e
Large accelerated filer		Accelerated filer	X
Non-accelerated filer	☐ (Do not check if a smaller reporting company)	Smaller reporting company	
Indicate by check i	mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act).	Yes □ No ⊠	

As of May 2, 2011, MasTec, Inc. had 79,419,669 shares of common stock, \$0.10 par value, outstanding.

MASTEC, INC. FORM 10-Q QUARTER ENDED MARCH 31, 2011

TABLE OF CONTENTS

PART I. FINANCIAL INFORMATION	3
Item 1. Financial Statements	3
Condensed Unaudited Consolidated Statements of Operations	3
Condensed Unaudited Consolidated Balance Sheets	4
Condensed Unaudited Consolidated Statements of Cash Flows	5
Notes to the Condensed Unaudited Consolidated Financial Statements	6
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	23
Item 3. Quantitative and Qualitative Disclosures about Market Risk	37
Item 4. Controls and Procedures	38
PART II. OTHER INFORMATION	38
Item 1, Legal Proceedings	38
Item 1A. Risk Factors	38
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	40
Item 5. Other Information	40
Item 6. Exhibits	41
SIGNATURES	42

PART 1. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

MASTEC, INC. CONDENSED UNAUDITED CONSOLIDATED STATEMENTS OF OPERATIONS

(in thousands, except per share amounts)

	For the Three I	
	2011	2010
Revenue	\$618,492	\$ 450,231
Costs of revenue, excluding depreciation and amortization	528,553	388,855
Gross profit	89,939	61,376
Depreciation and amortization	15,109	14,175
General and administrative expenses	32,499	27,673
Interest expense, net	7,912	7,376
Other income, net	(165)	(297)
Income before provision for income taxes	\$ 34,584	\$ 12,449
Provision for income taxes	(13,488)	(5,054)
Net income	\$ 21,096	\$ 7,395
Net loss attributable to non-controlling interests	(10)	(16)
Net income attributable to MasTec	\$ 21,106	\$ 7,411
Earnings per share – basic and diluted (See Note 3 - Earnings Per Share)		
Basic net income per share attributable to MasTec	\$ 0.27	\$ 0.10
Basic weighted average common shares outstanding	78,426	75,996
Diluted net income per share attributable to MasTec	\$ 0.26	\$ 0.10
Diluted weighted average common shares outstanding	83,633	84,398

The accompanying notes are an integral part of these condensed unaudited consolidated financial statements.

MASTEC, INC. CONDENSED UNAUDITED CONSOLIDATED BALANCE SHEETS

(in thousands, except share and per share amounts)

	_	March 31, 2011	_ D	ecember 31, 2010
Assets				
Current assets:				
Cash and cash equivalents, including restricted cash of \$18,251 at March 31, 2011 and \$18,000 at December 31, 2010	\$	197,991	\$	177,604
Accounts receivable, costs and earnings in excess of billings and retainage, net of allowance	Ψ	400,992	Ψ	444,363
Securities available for sale		4,142		
Inventories		66,191		57,483
Deferred tax assets, net		6,178		11,990
Prepaid expenses and deposits		16,349		17,176
Other current assets		14,612		13,058
Total current assets	\$	706,455	\$	721,674
Property and equipment, net		206,548		180,786
Goodwill		618,399		613,369
Other intangible assets, net		76,266		78,190
Securities available for sale		14,808		18,997
Investments in unconsolidated entities		13,067		12,922
Other assets		30,316		29,890
Total assets	\$1	1,665,859	\$1	,655,828
Liabilities and Shareholders' Equity				
Current liabilities:				
Current maturities of long-term debt	\$	23,475	\$	18,399
Accounts payable		203,446		196,363
Accrued salaries and wages		37,575		33,079
Accrued taxes payable		12,512		15,269
Accrued insurance		14,320		16,616
Other accrued expenses		20,952		13,329
Acquisition-related contingent consideration, current portion		25,116		35,639
Billings in excess of costs and earnings		84,061		136,070
Other current liabilities	_	21,615		21,780
Total current liabilities	\$	443,072	\$	486,544
Acquisition-related contingent consideration		32,774		34,695
Other liabilities		27,693		24,789
Deferred tax liabilities, net		76,863		62,487
Long-term debt		390,107		394,151
Total liabilities	\$	970,509	\$ 1	,002,666
Commitments and contingencies (See Note 14)				
Shareholders' equity:				
Preferred stock, \$1.00 par value; authorized shares - 5,000,000; issued and outstanding shares - none	\$	_	\$	_
Common stock, \$0.10 par value; authorized shares - 145,000,000; issued and outstanding shares - 79,219,910 and 78,215,189 at March 31, 2011 and December 31, 2010, respectively		7,922		7,822
Capital surplus		684,917		663,927
Accumulated surplus (deficit)		8,594		(12,512)
Accumulated other comprehensive loss		(6,155)		(6,156)
Total MasTec shareholders' equity	\$	695,278	\$	653,081
Non-controlling interests	Φ	72	Ф	81
Total shareholders' equity	\$	695,350	\$	653,162
Total liabilities and shareholders' equity		1,665,859	_	,655,828
Total Habilities and Shareholders equity	Φ.	1,005,059	φI	,033,028

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these condensed unaudited consolidated financial statements.}$

MASTEC, INC. CONDENSED UNAUDITED CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands)

	For the Three Marci	
	2011	2010
Cash flows from operating activities:		
Net income	\$ 21,096	\$ 7,395
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation and amortization	15,109	14,175
Stock-based compensation expense	882	933
Excess tax benefit from stock-based compensation	(2,878)	(39)
Amortization of deferred financing costs	804	807
Accretion of senior convertible note discount and investor fees	921	
Provision for doubtful accounts	510	238
Provision for losses on construction projects, net	160	(288)
Provision for inventory obsolescence	75	74
Earnings from equity method investments	(145)	
Loss (gain) on disposal of assets	118	(252)
Changes in assets and liabilities, net of assets acquired and liabilities assumed:		
Accounts receivable, costs and earnings in excess of billings and retainage	42,861	72,307
Inventories	(8,783)	(11,722)
Deferred tax assets and liabilities, net	13,589	4,843
Other assets, current and non-current portion	(1,595)	(215)
Accounts payable and accrued expenses	19,322	(22,508)
Billings in excess of costs and earnings	(52,009)	(25,846)
Other liabilities, current and non-current portion	284	(5,225)
Net cash provided by operating activities	\$ 50,321	\$ 34,677
Cash flows used in investing activities:		
Cash paid for acquisitions, net, including contingent consideration	(17,474)	(9,345)
Capital expenditures	(14,620)	(5,936)
Investments in life insurance policies	(360)	(115)
Proceeds from sale of assets	1,334	1,429
Net cash used in investing activities	\$ (31,120)	\$(13,967)
Cash flows provided by (used in) financing activities:		
Payments of other borrowings	(2,775)	(3,973)
Payments of capital lease obligations	(4,500)	(3,117)
Proceeds from stock option exercises	6,551	717
Excess tax benefit from stock-based compensation	2,878	39
Payments of financing costs	(1,000)	(47)
Net cash provided by (used in) financing activities	\$ 1,154	\$ (6,381)
Net increase in cash and cash equivalents	20,355	14,329
Net effect of currency translation on cash	32	83
Cash and cash equivalents - beginning of period	177,604	88,521
Cash and cash equivalents - end of period	\$197,991	\$102,933
Supplemental cash flow information:		
Interest paid	\$ 6,999	\$ 7,481
Income taxes paid	955	1,041
Supplemental disclosure of noncash information:		
Equipment acquired under capital lease	\$ 1,774	\$ —
Conversion of leases from operating to capital	24,005	_
Exchange of senior convertible notes	202,322	
Conversion of senior convertible notes to shares	28	_

The accompanying notes are an integral part of these condensed unaudited consolidated financial statements.

MasTec, Inc.

Notes to the Condensed Unaudited Consolidated Financial Statements

Note 1 - Nature of the Business

MasTec, Inc. (collectively with its subsidiaries, "MasTec" or the "Company") is a leading infrastructure construction company operating mainly throughout North America across a range of industries. The Company's activities include the engineering, building, installation, maintenance and upgrade of energy, communication and utility infrastructure, such as: electrical utility transmission and distribution, wind farms, solar farms and other renewable energy, natural gas and petroleum pipeline infrastructure, wireless, wireline and satellite communications, industrial infrastructure and water and sewer systems. MasTec's customers are primarily in the utility, communications and government industries.

Note 2 - Basis of Presentation and Significant Accounting Policies

The accompanying condensed unaudited consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States ("U.S. GAAP") for interim financial information and with the instructions for Form 10-Q and Rule 10-01 of Regulation S-X. Pursuant to these rules and regulations, certain information and footnote disclosures normally included in the annual consolidated financial statements prepared in accordance with U.S. GAAP have been condensed or omitted. The accompanying condensed consolidated balance sheet as of December 31, 2010 is derived from the Company's audited financial statements at that date. Since certain information and footnote disclosures have been condensed or omitted, these condensed unaudited consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto as of and for the year ended December 31, 2010 contained in the Company's most recent Annual Report on Form 10-K. In management's opinion, all normal and recurring adjustments considered necessary for a fair presentation of the financial position, results of operations and cash flows for the periods presented have been included. Interim period operating results do not necessarily indicate the results that may be expected for any other interim period or for the full fiscal year. The Company believes that the disclosures made in these condensed unaudited consolidated financial statements are adequate to make the information not misleading.

Management Estimates

The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. Key estimates include: the recognition of revenue, in particular, on long term construction contracts, including estimates to complete and provisions for contract losses; allowance for doubtful accounts; accrued self-insured claims; estimated fair values of goodwill, acquisition-related contingent consideration, intangible assets, securities available for sale and certain convertible debt obligations; asset lives used in computing depreciation and amortization, including amortization of intangible assets; accounting for income taxes; and range of possible exposure associated with contingencies and ongoing litigation. While management believes that such estimates are fair when considered in conjunction with the consolidated financial position and results of operations taken as a whole, actual results could differ from those estimates and such differences may be material to the condensed unaudited financial statements.

Principles of Consolidation

The accompanying condensed unaudited consolidated financial statements include MasTec, Inc. and its subsidiaries and include the accounts of all majority-owned subsidiaries over which the Company exercises control and, when applicable, entities in which the Company has a controlling financial interest. Other parties' interests in companies in which MasTec exercises control and has a controlling financial interest are reported as non-controlling interests within shareholders' equity. Net income or loss attributable to non-controlling interests is reported as a separate line item below net income. The Company's investments in entities in which the Company does not have a controlling interest, but has the ability to exert significant influence, are accounted for using the equity method of accounting. Equity method investments are recorded as long-term assets in the condensed unaudited consolidated balance sheets. Income or losses from these investments is recorded in other income, net, in the condensed unaudited consolidated statements of operations. The cost method is used for investments in entities over which the Company does not have the ability to exert significant influence. All significant intercompany balances and transactions have been eliminated in consolidation.

Reclassifications

Certain prior year amounts have been reclassified to conform to the current period presentation.

Significant Accounting Policies

Except for adoption of the pronouncements discussed below, there have been no material changes to the significant accounting policies described in the Company's Annual Report on Form 10-K for the year ended December 31, 2010.

Recently Adopted Accounting Pronouncements

In April 2010, the Financial Accounting Standards Board ("FASB") issued ASU 2010-17, *Milestone Method of Revenue Recognition, a consensus of the FASB Emerging Issues Task Force* ("ASU 2010-17"). This update provides guidance on defining a milestone under Topic 605, *Revenue Recognition — Milestone Method*, and determining when it may be appropriate to apply the milestone method of revenue recognition for research or development transactions. Consideration that is contingent on achievement of a milestone in its entirety may be recognized as revenue in the period in which the milestone is achieved only if the milestone is judged to meet certain criteria to be considered substantive. Milestones should be considered substantive in their entirety and may not be bifurcated. An arrangement may contain both substantive and nonsubstantive milestones that should be evaluated individually. ASU 2010-17 is effective on a prospective basis for milestones achieved in fiscal years, and interim periods within those years, beginning on or after June 15, 2010 with early adoption permitted. The adoption of ASU 2010-17 effective January 1, 2011 did not have a material impact on the Company's condensed unaudited consolidated financial statements.

In December 2010, the FASB issued ASU 2010-28, *Intangibles – Goodwill and Other (Topic 350): When to Perform Step 2 of the Goodwill Impairment Test for Reporting Units with Zero or Negative Carrying Amounts* ("ASU 2010-28"). ASU 2010-28 modifies Step 1 of the goodwill impairment test for reporting units with zero or negative carrying amounts. For those reporting units, an entity is required to perform Step 2 of the goodwill impairment test if it is more likely than not that a goodwill impairment exists. In determining whether it is more likely than not that a goodwill impairment exists, an entity should consider whether there are any adverse qualitative factors indicating that an impairment may exist. ASU 2010-28 is effective for fiscal years, and interim periods within those years, beginning after December 15, 2010. The adoption of ASU 2010-28 effective January 1, 2011 did not have a material impact on the Company's condensed unaudited consolidated financial statements.

In December 2010, the FASB issued ASU 2010-29, Business Combinations (Topic 805): Disclosure of Supplementary Pro Forma Information for Business Combinations, a consensus of the FASB Emerging Issues Task Force ("ASU 2010-29"). The objective of ASU 2010-29 is to address diversity in practice relating to the interpretation of pro forma revenue and earnings disclosure requirements for business combinations. Under ASU 2010-29, comparative financial statements should disclose revenue and earnings of the combined entity as if the business combinations that have occurred during the current year had been in effect as of the beginning of the comparable prior annual reporting period only. Additionally, ASU 2010-29 expands the supplemental pro forma disclosures under Topic 805 to include a description of the nature and amount of material, nonrecurring pro forma adjustments directly attributable to the business combinations included in reported pro forma revenue and earnings. ASU 2010-29 is effective prospectively for business combinations occurring on or after the beginning of the first annual reporting period beginning on or after December 15, 2010. The adoption of ASU 2010-29 effective January 1, 2011 did not have a material impact on the Company's condensed unaudited consolidated financial statements.

Recently Issued Accounting Standards Not Adopted as of March 31, 2011

There were no new accounting standards issued or changes in existing accounting standards that were of significance to the Company during the quarter ended March 31, 2011.

Note 3 – Earnings Per Share

Basic earnings per share is computed by dividing earnings available to MasTee's common shareholders by the weighted average number of common shares outstanding for the period. Diluted earnings per share reflects the potential dilution that could occur if stock options and unvested restricted stock ("common stock equivalents"), convertible securities, or other contracts to issue common stock were exercised or converted into common stock or resulted in the issuance of common stock that then shared in the earnings of MasTec.

As described in Note 9 – Debt, the Company had convertible notes outstanding during both the three months ended March 31, 2011 and 2010. During the first quarter of 2011, the Company exchanged \$105.3 million of its original 4.0% senior convertible notes and \$97 million of its original 4.25% senior convertible notes for identical principal amounts of new 4.0% and 4.25% senior convertible notes (the "New 4.0% Notes" and the "New 4.25% Notes," respectively and, collectively, the "New Notes"). The terms of the New Notes are substantially identical to the original notes, except that the New Notes have an optional physical, cash or combination settlement feature and contain certain conditional conversion features. Due to the optional cash settlement feature and management's intent to settle the principal amount thereof in cash, the conversion shares underlying the principal amount of the New Notes are not required to be included in the Company's diluted share count. If, however, the average stock price per share exceeds the \$15.76 conversion price for the New 4.0% Notes, or the \$15.48 conversion price for the New 4.25% Notes, the resulting amount, in shares, of the premium over the principal amount is included in the Company's diluted share count ("premium shares").

The Company's original senior convertible notes are reflected in the calculation of diluted earnings per share for the corresponding periods by application of the "if converted" method to the extent their effect is dilutive. Under the "if-converted" method, in computing the dilutive effect of the Company's convertible notes, net income is adjusted to add back the after-tax amount of interest recognized in the period associated with the convertible notes, and correspondingly, the convertible notes are assumed to have been converted with the resulting common shares added to weighted average shares outstanding. The New Notes are reflected in the calculation of diluted earnings per share to the extent that the Company's average stock price, beginning with the date of issuance through the end of the quarterly period, exceeds the respective conversion prices of the New Notes. If the average stock price for the period exceeds the conversion prices of the New Notes, the resulting equivalent value, in shares, is added to weighted average shares outstanding.

The following table summarizes the principal values of the Company's outstanding convertible notes for the three months ended March 31, 2011 and 2010, including their respective classification within the computation of earnings per share for each period (in millions).

		ntns Ended ch 31,
	2011	2010
Dilutive:		
New 4.0% senior convertible notes (1)	\$ 105.3	\$ —
New 4.25% senior convertible notes (1)	97.0	
Original 4.0% senior convertible notes (2)	9.7	115.0
Original 4.25% senior convertible notes (2)	3.0	
Anti-Dilutive:		
Original 4.25% senior convertible notes	_	100.0
Total principal value, outstanding convertible notes	\$215.0	\$215.0

- (1) Diluted shares associated with the New Notes are attributable to the premium feature.
- (2) Diluted shares associated with the original senior convertible notes are attributable to the underlying principal amounts.

The Company's average stock price for the quarter ended March 31, 2011 exceeded the conversion prices of the New Notes. As a result, premium shares were included in the Company's diluted share count as of March 31, 2011 as follows:

	Three Months Ended March 31, 2011		
	New 4.0%		
	Senior New 4.		
	Convertible Notes	Convertible Notes	
Conversion price	\$ 15.76	\$ 15.48	
Weighted average share price (from date of issuance)	\$ 17.63	\$ 17.51	
Weighted equivalent premium shares (in thousands)	576	640	

The number of premium shares included in MasTec's diluted share count varies with fluctuations in MasTec's actual share price for the related period. Higher actual share prices result in a higher number of equivalent premium shares.

The following table provides a reconciliation of MasTec's basic and diluted net income and weighted average shares outstanding for the periods indicated (in thousands, except per share amounts):

	Three Months Ended March 31,	
	2011	2010
Basic		
Net income attributable to MasTec	\$21,106	\$ 7,411
Weighted average shares outstanding	78,426	75,996
Basic earnings per share attributable to MasTec	\$ 0.27	\$ 0.10
Diluted		
Net income:		
Basic net income attributable to MasTec	\$21,106	\$ 7,411
Interest expense on original 4.0% senior convertible notes, net of tax	154	683
Interest expense on original 4.25% senior convertible notes, net of tax	95	
Diluted net income attributable to MasTec	\$21,355	\$ 8,094
Shares:		
Basic weighted average shares outstanding	78,426	75,996
Dilutive common stock equivalents	1,408	1,105
Dilutive premium shares, new 4.0% senior convertible notes	576	_
Dilutive premium shares, new 4.25% senior convertible notes	640	_
Dilutive shares, original 4.0% senior convertible notes	1,623	7,297
Dilutive shares, original 4.25% senior convertible notes	960	_
Diluted weighted average shares outstanding	83,633	84,398
Diluted earnings per share attributable to MasTec	\$ 0.26	\$ 0.10

Common stock equivalents are measured under the treasury stock method. For the three months ended March 31, 2011, approximately 167,000 restricted shares, or approximately 41,000 shares on a weighted average basis, had average market prices for the period in excess of their grant date prices. These shares were not included in the computation of diluted shares outstanding because the effect would have been anti-dilutive.

Note 4 - Acquisitions and Other Investments

DirectStar/Funraisers - Amendment of Purchase Option Agreement

In February 2011, the Company amended its purchase option agreement with Red Ventures LLC ("Red Ventures"), the same seller that sold to MasTec both DirectStar TV LLC ("DirectStar") and 100% of the membership interests of Funraisers PR, LLC ("Funraisers"). DirectStar, together with its subsidiaries, including Funraisers, is referred to as the "DirectStar Business." The amended and restated purchase option agreement grants to Red Ventures an option to purchase the DirectStar Business from MasTec at any time from March 1, 2012 to November 30, 2012 for an amount equal to the sum of: (i) the shareholders' equity of the DirectStar Business as of May 31, 2010, (ii) five percent (5%) of adjusted net income (generally, the net income (loss) before provision for income taxes) of the DirectStar Business from January 1, 2010 until the last day of the month immediately prior to the date of the sale and (iii) \$25,600,000. The purchase option also allows Red Ventures to pay up to 35% of the option purchase price in the form of a secured note with a one year term. In connection with the amendment to the purchase option, MasTec agreed to increase certain commissions that DirectStar presently pays to Red Ventures for managing the DirectStar Business. DirectStar, in support of the DIRECTV® installation business, provides marketing and sales services on behalf of DIRECTV®. Should Red Ventures execute its purchase option, the Company's revenues and profits from DIRECTV® would be materially reduced.

As of March 31, 2011 and December 31, 2010, the estimated fair value of the purchase option was \$0, determined using a probability-weighted market-based approach including Level 3 inputs such as projected earnings before interest, taxes, depreciation and amortization ("EBITDA") and EBITDA multiples.

Equity Method and Other Investments

In November 2010, MasTec entered into a membership interest purchase agreement and invested \$10 million to purchase a 33% voting interest in EC Source Services LLC ("EC Source") and acquired a two-year option (the "EC Source Merger Option") that grants MasTec the right, but not the obligation, to acquire EC Source via merger. EC Source is an extra high voltage specialty transmission contractor, and MasTec's investment will be used by EC Source to support working capital and bonding requirements as well as equipment purchases, as necessary, to meet the demands of EC Source's existing and future projects. See Note 17 – Subsequent Events in the notes to the condensed unaudited consolidated financial statements regarding the Company's exercise of the EC Source Merger Option on April 29, 2011.

Through a 60%-owned consolidated subsidiary, MasTec has a 34% interest in a rock extraction business in Panama (for a net beneficial ownership interest of 20.4%), for which the Company invested \$1.6 million. This investment is accounted for under the equity method of accounting. During the three months ended March 31, 2011, MasTec performed construction services for this investee and recognized revenue of approximately \$0.7 million. As of March 31, 2011 and December 31, 2010, approximately \$2.2 million and \$2.1 million, respectively, is included in costs and earnings in excess of billings related to this customer.

The Company has certain other cost and equity method investments, which are not material individually or in the aggregate, as of March 31, 2011 and December 31, 2010.

Note 5 - Goodwill and Other Intangible Assets

The following table sets forth information for MasTec's goodwill and intangible assets as of the dates indicated (in millions):

	March 31, 2011	December 31, 2010
Amortizing intangible assets: (1)		
Gross carrying amount	\$ 75.2	\$ 75.2
Less: accumulated amortization	(40.8)	(38.9)
Amortizing intangible assets, net	\$ 34.4	\$ 36.3
Non-amortizing intangible assets:		
Goodwill	\$ 618.4	\$ 613.4
Trade name	35.3	35.3
Other	6.6	6.6
Non-amortizing intangible assets	\$ 660.3	\$ 655.3
Goodwill and other intangible assets	\$694.7	\$ 691.6

(1) Consists principally of customer relationships, tradenames and non-compete agreements with finite lives.

The following table provides a reconciliation of changes in goodwill and other intangible assets from December 31, 2010 to March 31, 2011 (in millions):

	Other Intangible Assets			
	Goodwill	Non-amortizing	Amortizing	Total
Balance at December 31, 2010	\$613.4	\$ 41.9	\$ 36.3	\$691.6
Accruals of acquisition-related contingent consideration	5.0	_		5.0
Amortization expense	<u> </u>		(1.9)	(1.9)
Balance at March 31, 2011	\$618.4	\$ 41.9	\$ 34.4	\$ 694.7

Information about estimated expense associated with amortizing intangible assets in future periods, for assets recorded as of March 31, 2011, is summarized in the following table (in millions):

	Amortization	
	Expense	
Remainder of 2011	\$	5.7
2012		4.7
2013		3.7
2014		3.0
2015		2.6
Thereafter		14.7
Total	\$	34.4

Note 6 - Fair Value of Financial Instruments

The Company's financial instruments include cash and cash equivalents, accounts and notes receivable, cash collateral deposited with insurance carriers, cash surrender value of life insurance policies, auction rate securities, cost and equity method investments, deferred compensation plan assets and liabilities, accounts payable and other current liabilities, acquisition-related contingent consideration and debt obligations.

Fair value is the price that would be received to sell an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. The fair value guidance establishes a valuation hierarchy, which requires maximizing the use of observable inputs when measuring fair value. The three levels of inputs that may be used are:

Level

1 Quoted market prices in active markets for identical assets or liabilities.

Level

2 Observable market based inputs or other observable inputs corroborated by market data.

Level 3 - Significant unobservable inputs that are not corroborated by observable market data. These values are generally determined using valuation models incorporating management's estimates of market participant assumptions.

Carrying amounts and estimated fair values of financial instruments as of March 31, 2011 and December 31, 2010 are as follows (in millions):

	March 31, 2011		December 31	, 2010
	Carrying Amou	<u>ıt Fair Value</u>	Carrying Amount	Fair Value
Assets:				
Cash surrender value of life insurance policies	\$ 8.	5 \$ 8.5	\$ 8.3	\$ 8.3
Auction rate securities	19.	0 19.0	19.0	19.0
Liabilities:				
Acquisition-related contingent consideration	\$ 45.	0 \$ 45.0	\$ 45.0	\$ 45.0
7.625% senior notes	150.	0 151.5	150.0	151.9
Original 4.0% senior convertible notes	9.	7 14.4	115.0	139.2
New 4.0% senior convertible notes	96.	4 98.5	_	_
Original 4.25% senior convertible notes	3	0 4.6	100.0	120.4
New 4.25% senior convertible notes	88.	5 89.9	_	

The following methods and assumptions were used to estimate the fair values of financial instruments:

Cash Surrender Value of Life Insurance Policies. Cash surrender values of life insurance policies are based on current cash surrender values as quoted by insurance carriers, a Level 1 input. Life insurance policies support the Company's split dollar agreements and deferred compensation plan assets.

Auction Rate Securities. The fair value of the Company's auction rate securities was estimated by an independent valuation firm, Houlihan Capital Advisors, LLC, using a probability weighted discounted cash flow model, a Level 3 input. See Note 7 - Securities Available for Sale.

Acquisition-Related Contingent Consideration. Acquisition-related contingent consideration in the table above represents the estimated fair value of additional future earn-outs payable for acquisitions of businesses that closed after December 15, 2008. The fair value of such acquisition-related contingent consideration is based on management's estimates and entity-specific assumptions, Level 3 inputs.

Debt. The estimated fair values of the Company's 7.625% senior notes and original 4.0% and 4.25% senior convertible notes are based on quoted market prices, Level 1 inputs. The estimated fair value of the debt component of the Company's New Notes is calculated using an income approach, based on a discounted cash flow model. This method is based on management's estimates of the Company's market interest rate for a similar nonconvertible instrument, which is a Level 3 input. See Note 9 - Debt.

Assets and Liabilities Measured at Fair Value on a Recurring Basis

As of March 31, 2011, the Company held certain assets and liabilities that are required to be measured at fair value on a recurring basis. The fair values of financial assets and liabilities measured on a recurring basis were determined using the following inputs as of March 31, 2011 and December 31, 2010 (in millions):

			Usi	ing Inputs	Considered as	Signific	cant
	Ma	alue as of rch 31, 011	in A Mark Identica	l Prices ctive ets for al Assets (el 1)	Significant Other Observable Inputs (Level 2)	Unob Ir	nificant eservable aputs evel 3)
Assets:							
Cash surrender value of life insurance policies	\$	8.5	\$	8.5	_		_
Auction rate securities	\$	19.0		_	_	\$	19.0
Liabilities:							
Acquisition-related contingent consideration	\$	45.0		_	_	\$	45.0

Fair Value Measurements

Fair Value Measurements Using Inputs

					dered as Signifi		P
	Dece	Value as of mber 31, 2010	in Mar Identi	ed Prices Active kets for cal Assets evel 1)	Significant Other Observable Inputs (Level 2)	Unol Iı	nificant oservable nputs evel 3)
Assets:							
Cash surrender value of life insurance policies	\$	8.3	\$	8.3	_		—
Auction rate securities	\$	19.0		_	_	\$	19.0
Liabilities:							
Acquisition-related contingent consideration	\$	45.0		_	_	\$	45.0

The following tables provide a reconciliation between the beginning and ending balances of items measured at fair value on a recurring basis using significant unobservable inputs (Level 3 inputs) for the three months ended March 31, 2011 and 2010:

Three Months Ended March 31, 2011	Stuc	Auction	Stru Fi	Securition Securities	Total	Co	ition-Related ntingent sideration
Balances at December 31, 2010	\$	16.4	\$	2.6	\$19.0	\$	45.0
Changes in unrealized (losses)/gains included in other comprehensive income		(0.1)		0.1			
Balances at March 31, 2011	\$	16.3	\$	2.7	\$19.0	\$	45.0
Three Months Ended March 31, 2010	Stuc	Auction	Stru Fi	Securition octured nance curities	<u>Total</u>	Co	ition-Related ntingent sideration
Balances at December 31, 2009	\$	16.4	\$	8.1	\$24.5	\$	40.3
Changes in unrealized losses included in other comprehensive income		(0.2)			(0.2)		
Balances at March 31, 2010	\$	16.2	\$	8.1	\$ 24.3	\$	40.3

Assets and Liabilities Measured at Fair Value on a Nonrecurring Basis

Assets and liabilities recognized or disclosed at fair value in the financial statements on a nonrecurring basis include items such as goodwill and long lived assets that are measured at fair value resulting from impairment, if deemed necessary. For the three months ended March 31, 2011 and March 31, 2010, the Company had no significant assets or liabilities required to be measured at fair value on a nonrecurring basis subsequent to their initial recognition.

Note 7 – Securities Available For Sale

The Company's securities available for sale consist of auction-rate securities, which represent interests in pools of student loans guaranteed by the U.S. government under the Federal Family Education Loan Program and a structured finance security. The structured finance security has an attached credit default swap under which the principal value would be partially or fully forfeited at net default rates on the underlying corporate debt obligations ranging from 8% to 9%. The actual net default rate as of March 31, 2011 was 4.89%. Both the structured finance security and the credit default swap are collateralized by investment grade credit-linked notes made up of floating rate international bank notes or credit card receivable notes.

Due to liquidity issues associated with auction rate securities, there was insufficient observable market data to determine the fair value of the Company's auction rate securities as of March 31, 2011. Therefore, their fair value was estimated by an independent valuation firm, Houlihan Capital Advisors, LLC, using a probability weighted discounted cash flow model. This valuation is sensitive to market conditions and management's judgment and can change significantly based on the assumptions used. The following tables set forth the fair value of the Company's auction rate securities by type of security and underlying credit rating as of March 31, 2011 and December 31, 2010 (in millions):

As of March 31, 2011	Underlying Credit Rating (1)		
	A3/A-	CCC	Total
Underlying security:			
Student loans	\$16.3	\$ —	\$16.3
Structured finance securities		2.7	2.7
Total auction rate securities	\$16.3	\$ 2.7	\$19.0
As of December 31, 2010	Under	lying Credit Rat	ing (1)
As of December 31, 2010	Under	lying Credit Rat	ing (1) Total
As of December 31, 2010 Underlying security:			
Underlying security:	\$16.4	CCC	**Total
Underlying security: Student loans	A3/A-	<u>ccc</u> \$—	Total

 The Company's auction rate securities maintain split ratings. For purposes of this table, securities are categorized according to their lowest rating.

As of March 31, 2011, the yields on the Company's auction-rate securities ranged from 1.74% to 2.51%. These yields represent the predetermined "maximum" reset rates that occur upon auction failures according to the specific terms within each security's prospectus. The issuers have been making interest payments when due. As of March 31, 2011, the weighted average yield for the Company's auction rate securities was 2.35%. Total interest earned on the Company's auction rate securities for the three months ended March 31, 2011 and 2010 was \$0.1 million and \$0.2 million, respectively.

Auction Rate Securities - Changes in Fair Value

Management evaluates whether unrealized losses on investments in securities are other-than-temporary, and if the unrealized losses are believed to be other-than-temporary, an impairment charge is recorded. There were no other-than-temporary impairment losses recognized for the three months ended March 31, 2011 or 2010.

Structured Finance Auction Rate Securities

In the fourth quarter of 2009, the Company deemed its structured finance auction rate securities to be other-than-temporarily impaired due to ratings agency downgrades of the corresponding securities. During 2010, the Company sold two of its three structured finance auction rate securities. The Company's remaining structured finance security as of March 31, 2011 has a par value of \$5.0 million and a cost basis of \$2.3 million. Cumulative credit losses of \$2.7 million have been recognized in prior years in connection with this security. Following the sale in 2010 of two of the Company's structured finance securities, management determined that the Company no longer met the criteria for intent to hold with respect to its remaining structured finance security. Accordingly, further declines in this security's carrying value will be recognized in earnings, whereas increases in its carrying value will be recorded in other comprehensive income, net of applicable taxes. Cumulative gross unrealized gains in other comprehensive income associated with this security total \$0.4 million, or \$0.2 million net of applicable income taxes, as of March 31, 2011 and \$0.3 million, or \$0.2 million net of applicable income taxes, as of December 31, 2010.

Student Loan Auction Rate Securities

Cumulative gross unrealized losses on the Company's student loan auction rate securities, which are included as a component of other comprehensive income, total \$1.2 million, or \$0.7 million, net of applicable income taxes as of March 31, 2011, and \$1.1 million, or \$0.7 million net of applicable income taxes, as of December 31, 2010.

In April 2011, the issuer of one of the Company's student loan auction rate securities redeemed its security at its par value of \$4.6 million. The estimated fair value of this security as of March 31, 2011 was reclassified to current assets. In connection with the redemption, \$0.5 million of cumulative unrealized losses and the corresponding tax impact of \$0.2 million were eliminated from other comprehensive income in April 2011.

Auction Rate Securities - Reconciliation of Cost Basis to Fair Value

The cost basis, gross cumulative unrealized (losses) gains and estimated fair values of the Company's auction rate securities for the periods indicated are as follows (in millions):

		March 31, 2011			
	Adjusted Cost Basis (1)	Gross Cumulative Unrealized (Losses)/Gains Fair Value			
Auction rate securities – student loans	\$ 17.5	\$ (1.2) \$ 16.3			
Auction rate securities – structured finance securities	2.3	0.4 2.7			
Total auction rate securities	\$ 19.8	\$ (0.8) \$ 19.0			
	1	December 31, 2010			
	Adjusted Cost Basis (1)	Gross Cumulative Unrealized (Losses)/Gains Fair Value			
Auction rate securities – student loans	\$ 17.5	\$ (1.1) \$ 16.4			
A satisfactory and the satisfactory of the sat					
Auction rate securities – structured finance securities	2.3	0.3 2.6			

(1) Adjusted cost basis reflects adjustments for credit and other losses recognized in earnings. Adjustments for credit and other losses on securities held as of March 31, 2011 totaled \$2.7 million. Par value of securities held as of both March 31, 2011 and December 31, 2010 was \$22.5 million.

As of March 31, 2011, contractual maturities of the Company's student loan auction rate securities range from 17 to 37 years, and for the structured finance security, 6 years.

Note 8 - Accounts Receivable, Costs and Earnings in Excess of Billings, and Retainage, Net of Allowance

Accounts receivable, classified as current, consists of the following (in millions):

	March 31, 	Dec	cember 31, 2010
Contract billings	\$ 219.0	\$	309.8
Retainage	31.8		32.7
Costs and earnings in excess of billings	159.0		110.4
	\$ 409.8	\$	452.9
Less allowance for doubtful accounts	(8.8)		(8.5)
Accounts receivable, net	\$ 401.0	\$	444.4

Retainage, which has been billed but is not due until completion of performance and acceptance by customers, is expected to be collected within one year. Any receivables, including retainage, expected to be collected beyond one year are recorded in other long-term assets.

The Company maintains an allowance for doubtful accounts for estimated losses resulting from the inability of customers to make required payments. The allowance for doubtful accounts, both for specific customers and as a reserve against other past due balances, was \$8.8 million and \$8.5 million as of March 31, 2011 and December 31, 2010, respectively.

The Company has a trade receivable for a "pay-when-paid" project, which provides for payment in quarterly installments through June 2015. The receivable has been recorded at its net present value, with the non-current portion recorded within other long-term assets in the condensed unaudited consolidated balance sheets. Imputed interest is recognized as interest income as earned. As of March 31, 2011, \$5.0 million is outstanding, with \$3.6 million recorded in other long-term assets. During the three months ended March 31, 2011, the Company recognized an immaterial amount of interest income related to this receivable.

Note 9 – Debt

The carrying value of debt is composed of the following at March 31, 2011 and December 31, 2010 (in millions):

Description	Maturity Date	March 31, 2011	December 31, 2010
7.625% senior notes	February 2017	\$ 150.0	\$ 150.0
New 4.0% senior convertible notes, \$105.3 million principal amount	June 2014	96.4	_
New 4.25% senior convertible notes, \$97.0 million principal amount	December 2014	88.5	_
Original 4.0% senior convertible notes	June 2014	9.7	115.0
Original 4.25% senior convertible notes	December 2014	3.0	100.0
3.5276% equipment term loan	In installments through 2013	11.7	12.9
Capital lease obligations, weighted average interest rate of 3.8%	In installments through		
	2016	45.8	24.5
Notes payable for equipment, weighted average interest rate of 5.2%	In installments through 2014	8.5	10.2
Total debt		\$ 413.6	\$ 412.6
Less current maturities		(23.5)	(18.4)
Long-term debt		\$ 390.1	\$ 394.2

Senior Convertible Note Exchange

During the first quarter of 2011, the Company exchanged \$105.3 million of its original 4.0% senior convertible notes and \$97 million of its original 4.25% senior convertible notes (together, the "Original Notes") for identical principal amounts of new 4.0% and 4.25% senior convertible notes (the "New Notes") for an exchange fee of approximately 50 basis points, or 0.5% of the aggregate exchanged principal amount of Original Notes. The terms of the New Notes are substantially identical to those of the Original Notes, except that the New Notes have an optional physical, cash or combination settlement feature and contain certain conditional conversion features. Exchange fees of approximately \$1.0 million will be treated as a component of interest expense and recognized over the remaining life of the New Notes. Approximately \$0.1 million and \$0.5 million of transaction costs were recognized in connection with the exchange within selling, general and administrative expense for the three months ended March 31, 2011 and the year ended December 31, 2010, respectively. Following the exchange, \$9.7 million and \$3.0 million of the original 4.0% and 4.25% notes, respectively, remain outstanding.

In accordance with the accounting guidance for convertible debt instruments, issuers with convertible debt instruments that may be settled in cash or other assets on conversion are required to separately account for the liability component of the instrument in a manner that reflects the entity's estimated market interest rate for a similar nonconvertible instrument as of the instrument's issuance date. As a result, the

Company separated the principal balance of the New Notes between the fair value of the debt component and the fair value of the common stock conversion feature. Using an income approach, management discounted the values of the New Notes at an estimated rate of 6.73%, which represents the estimated market interest rate for a similar nonconvertible instrument as of the date of the exchange. The total debt discount of \$17.4 million will be accreted to interest expense over the remaining terms of the New Notes. This will increase interest expense during the term of the New Notes above the 4.0% and 4.25% cash coupon interest rates to an effective interest rate of 6.73%. The fair value of the common stock conversion feature is recorded as a component of stockholders' equity.

The carrying values of the debt and equity components of the New Notes as of March 31, 2011 are as follows (in millions):

	March 31, 2011			
		New 4.25%		
	New 4.0% Senior	Senior		
	Convertible Notes	Convertible Notes		
Principal amount	\$ 105.3	\$ 97.0		
Unamortized debt discount (1)	(8.4)	(8.1)		
Unamortized balance of investor fees (2)	(0.5)	(0.4)		
Net carrying amount of debt component	\$ 96.4	\$ 88.5		
Carrying amount of equity component	\$ 8.9	\$ 8.5		

- Represents the unamortized portion of the discount created from separation of the debt portion of the New Notes from the equity portion. This discount will be accreted to interest expense over the term of the New Notes.
- (2) Represents the unamortized portion of exchange fees paid in connection with the debt exchange. These fees will be accreted to interest expense over the term of the New Notes.

The New Notes are guaranteed by the Company's subsidiaries that guarantee the Original Notes.

Other Debt Activity

In January 2011, the Company modified the terms of certain of its equipment operating leases. As a result of the modifications, these leases were required to be classified as capital leases as of the effective date of the lease modifications. Accordingly, \$24.0 million of capital lease assets and corresponding capital lease obligations were recorded in the Company's condensed unaudited consolidated balance sheet as of January 1, 2011.

Debt Guarantees and Covenants

The subsidiary guarantees related to the Company's senior convertible notes and senior notes are full and unconditional and joint and several, and there are generally no contractual restrictions on the Company's ability to obtain funds from its subsidiaries. Also, MasTec, Inc. is a holding company with no independent assets or operations, and the Company's subsidiaries that do not guarantee the senior convertible notes and senior notes are minor, individually, and in the aggregate, as such term is defined under the rules and regulations of the SEC. Accordingly, condensed unaudited consolidating financial information for MasTec, Inc. and its subsidiaries is not presented.

MasTec was in compliance with all debt covenants as of March 31, 2011 and December 31, 2010.

Interest Expense, Net

Interest expense for the three months ended March 31, 2011 includes \$0.9 million of debt discount and investor fee accretion associated with the New Notes. Details of interest expense, net, for the periods indicated is as follows:

	Three Mon March	
	2011	2010
Interest expense:		
Contractual interest payments	\$ 5.8	\$ 6.1
Senior convertible note discount and related investor fee accretion	0.9	_
Amortization of deferred financing costs	0.8	0.8
Other interest expense	0.6	0.8
Total interest expense	\$ 8.1	\$ 7.7
Interest income	(0.2)	(0.3)
Interest expense, net	\$ 7.9	\$ 7.4

Note 10 - Stock-Based Compensation Plans

Share-based payments, to the extent they are compensatory, are recognized based on their grant date fair values and the estimated number of shares ultimately expected to vest. The Company records a deferred tax asset, or future tax benefit, based on the amount of share-based compensation recognized in the financial statements over the vesting period of share-based awards. For stock options, upon exercise, or for restricted share awards, upon vesting, if the fair value of the share-based award exceeds its grant date fair value, the tax effect of this difference ("excess tax benefit") is recorded as an increase to additional paid-in capital ("APIC"), creating an "APIC Pool." If the exercise date/vesting date fair value of a share-based award is less than its grant date fair value, the tax effect of this difference would reduce the APIC Pool. If the APIC Pool is reduced to zero, subsequent shortfalls would increase tax expense.

The Company has three stock-based compensation plans with stock options and restricted share grants outstanding as of March 31, 2011: the 1994 Stock Option Plan for Non-Employee Directors; the 2003 Employee Stock Incentive Plan, as amended; the Amended and Restated 2003 Stock Incentive Plan for Non-Employees, as amended; and individual option and restricted stock agreements. Under these plans, there were a total of 3,572,472 options and/or restricted shares available for grant at March 31, 2011.

Restricted Share Awards

MasTec grants restricted share awards, which are valued based on the market price of MasTec common stock on the date of grant. Through March 31, 2011, approximately 1,941,000 shares of restricted share awards have been issued with a value of approximately \$20.5 million, which is being expensed over vesting periods ranging from the day of issuance to five years. Total unearned compensation related to restricted share awards as of March 31, 2011 is approximately \$4.7 million, which is expected to be recognized over a weighted average period of less than 2 years. The total intrinsic value of restricted share awards vested during the three months ended March 31, 2011 and 2010, which is based on the market price on the date of vesting, was \$5.3 million and \$0.1 million, respectively.

The following is a summary of restricted share award activity during the three months ended March 31, 2011:

	Restricted	Weigl	hted Average
	Shares	Grant I	Date Fair Value
Non-vested restricted shares, December 31, 2010	1,058,720	\$	9.42
Granted	169,566		19.02
Vested	(266,556)		7.71
Non-vested restricted shares, March 31, 2011	961,730	\$	11.59

For certain restricted share awards, the number of shares issued on the vesting date is net of the equivalent number of shares required to satisfy the recipient's minimum statutory tax withholding requirements. The Company then pays the corresponding withholding taxes to the appropriate taxing authorities in cash on behalf of the recipient. Although the withheld shares are not issued, they are treated as common stock repurchases in the consolidated financial statements, as they reduce the number of shares that would have been issued upon vesting. Withheld shares are reflected as an operating activity within the statement of cash flows. During the three months ended March 31, 2011 and 2010, there were no material withholdings of shares for recipient tax withholding requirements.

Stock Options

The Company has granted options to purchase its common stock to employees and members of the Board of Directors and affiliates under various stock option plans at no less than the fair market value of the underlying stock on the date of grant. No stock options have been granted since 2006.

The following is a summary of stock option activity during the three months ended March 31, 2011:

			Weighted Average	Aggregate
		Weighted	Remaining	Intrinsic
	Stock	Average	Contractual Life	Value (1)
	Options	Exercise Price	(years)	(in millions)
Options outstanding December 31, 2010	2,588,170	\$ 10.27	3.87	\$ 11.2
Exercised	(667,417)	\$ 9.42		
Options outstanding March 31, 2011	1,920,753	\$ 10.56	3.76	\$ 19.7
Options exercisable March 31, 2011	1,915,753	\$ 10.55	3.75	\$ 19.6

Amount represents the difference between the exercise price and the market price of the Company's stock on the last trading day of the corresponding period, multiplied by the number of in-the-money options.

As of March 31, 2011, all outstanding stock options were fully vested. The total intrinsic value of options exercised during the three months ended March 31, 2011 and 2010, which is based on the difference between the exercise price and the market price of the Company's stock at the date of exercise was \$6.6 million and \$0.2 million, respectively. Proceeds received from options exercised during the three months ended March 31, 2011 and 2010 totaled \$6.6 million and \$0.7 million, respectively.

Stock Based Compensation Expense and Related Tax Benefit

Stock based compensation expense and related tax benefits for the periods indicated are as follows (in millions):

	Three Months Ended March 31,	
	2011	2010
Stock based compensation expense related to restricted share awards	\$ 0.9	\$ 0.9
Stock based compensation expense related to stock options		
Total stock based compensation expense	\$ 0.9	\$ 0.9
Total income tax benefit from restricted share awards	\$ 1.2	\$ 0.4
Total income tax benefit from stock options	2.0	
Total income tax benefit from stock based compensation	\$ 3.2	\$ 0.4
Excess tax benefit from vested restricted shares (1)	\$ 0.9	\$ —
Excess tax benefit from stock options exercised (1)	2.0	
Total excess tax benefit related to vested restricted shares and stock options exercised	\$ 2.9	\$ —

(1) Excess tax benefits, which represent cash flows from tax deductions in excess of compensation expense recognized for stock options exercised and vested restricted shares, are classified as financing cash flows in the Company's condensed unaudited consolidated statements of cash flows.

Note 11 - Employee Benefit Plans

401(k) Plan. MasTec has a 401(k) plan covering all eligible employees. Subject to certain dollar limits, eligible employees may contribute up to 75% of their pre-tax annual compensation to the 401(k) plan. Effective January 1, 2011, MasTec increased its 100% match of employee contributions from 1% to 2.5% of the employee's salary, capped at \$2,500. Effective January 1, 2011, matching contributions are payable annually, 50% in shares of MasTec common stock and 50% in cash. During the three months ended March 31, 2011 and 2010, MasTec accrued \$0.4 million and \$0.5 million of 401(k) matching contributions, respectively.

Deferred Compensation Plan. MasTec offers a deferred compensation plan to its highly compensated employees. These employees are allowed to contribute a percentage of their pre-tax annual compensation to the deferred compensation plan. Effective January 1, 2011, MasTec increased its 100% match of employee contributions from 1% to 2.5% of the employee's salary, capped at \$2,500. Effective January 1, 2011, matching contributions are payable annually, 50% in shares of MasTec common stock and 50% in cash. During the three months ended March 31, 2011 and 2010, MasTec accrued \$0.4 million and \$0.1 million of deferred compensation plan matching contributions, respectively. Deferred compensation plan assets of approximately \$1.9 million and related liabilities of approximately \$1.7 million are included in other long-term assets and other long-term liabilities, respectively, in the condensed unaudited consolidated balance sheet as of March 31, 2011.

Multi-Employer Pension Plans. Certain of MasTec's subsidiaries contribute amounts to multi-employer pension plans and employee benefit trusts. During the three months ended March 31, 2011 and 2010, total contributions to these plans, and the related number of employees covered by these plans, ranged as follows:

	Number o	f Employees		butions to ployer Plans
Three Months Ended March 31,	Low	High	(in r	nillions)
2011	767	1,111	\$	5.7
2010	254	1.409	\$	4.2

Multi-employer pension plan contributions vary by period, depending upon the location and number of ongoing projects and the need for union resources in connection with those projects.

See Note 14 – Commitments and Contingencies for additional information.

Note 12 - Income Taxes

The Company's consolidated tax rate for the three months ended March 31, 2011 and 2010, was 39% and 40.6%, respectively. In determining the quarterly provision for income taxes, management uses an estimated annual effective tax rate based on forecasted annual pre-tax income and permanent items, statutory tax rates and tax planning opportunities in the various jurisdictions in which the Company operates. The impact of significant discrete items is separately recognized in the quarter in which they occur. Significant factors that could impact the annual effective tax rate include management's assessment of certain tax matters, the location and amount of taxable earnings, changes in certain non-deductible expenses and expected credits.

The Company's subsidiaries file income tax returns in numerous tax jurisdictions, including U.S. federal, most U.S. states and certain foreign jurisdictions. The statute of limitations varies by the various jurisdictions in which the Company operates. The Company's U.S. federal income tax returns for years ending on or after 2007 still remain open for examination. Although management believes its calculations for tax returns are correct and the positions taken thereon are reasonable, the final outcome of tax audits could be materially different from the resolution management currently anticipates, and those differences could result in significant costs or benefits to the Company.

Note 13 – Comprehensive Income

Comprehensive income is a measure of net income and all other changes in equity that result from transactions other than those with shareholders. Comprehensive income consists of net income, unrealized gains and losses from securities available for sale, foreign currency translation adjustments and losses from non-controlling interests.

Comprehensive income for periods indicated consisted of the following (in thousands):

	For the Three Months Ended March 31,		
	2011	2010	
Net income	\$ 21,096	\$ 7,395	
Foreign currency translation gain	32	84	
Unrealized loss from securities available for sale, net of tax	(31)	(127)	
Comprehensive income	\$ 21,097	\$ 7,352	
Comprehensive net loss attributable to non-controlling interests	(10)	(16)	
Comprehensive income attributable to MasTec	\$ 21,107	\$ 7,368	

Accumulated other comprehensive losses of \$6.2 million as of both March 31, 2011 and December 31, 2010 are primarily attributable to unrealized losses and the related income tax effects associated with the Company's investments in securities available for sale. See Note 7 – Securities Available for Sale.

Note 14 - Commitments and Contingencies

Legacy Litigation

MasTec is subject to litigation, some of which dates from the period 2001 through 2006.

Outstanding Legacy Litigation. The Company is vigorously pursuing claims in excess of \$5 million against Aon Risk Services, Inc. of Florida, an insurance broker, for breach of contract and breach of fiduciary duty for the losses arising from a denial of insurance coverage. Discovery is ongoing.

MasTec provided telecommunication infrastructure services to Adesta Communications, Inc. ("Adesta") in 2000 and 2001. Adesta filed for bankruptcy in 2001. Adesta's bankruptcy trustee sold Adesta's assets in the bankruptcy trust, and MasTec is waiting for a distribution from that trust. Based on MasTec's current understanding of the expected distribution, MasTec has reduced its receivable to approximately \$0.9 million, which is recorded in other current assets on its consolidated balance sheet at March 31, 2011.

The labor union representing the workers of Sistemas e Instalaciones de Telecomunicación S.A. ("Sintel"), a former MasTec subsidiary that was sold in 1998, filed a claim that initiated an investigative action with the Audiencia Nacional, a Spanish federal court, against Telefonica and dozens of other defendants including current and former officers and directors of MasTec (including Jorge Mas, Chairman of the Company's Board of Directors) relating to Sintel's 2000 bankruptcy. The union alleged Sintel and its creditors were damaged in the approximate amount of 300 million euros (approximately U.S. \$424 million as of March 31, 2011). In June 2009, the Audiencia Nacional issued an order that the trial phase was commencing against the MasTec defendants and other defendants. In June 2010, the investigative court issued an order for letters rogatory to the United States for assistance in obtaining evidence from and seizing the assets of the MasTec defendants and MasTec. The Company believes that the order violates U.S. and international law and is opposing the order. MasTec and the MasTec defendants believe the claims are

frivolous and are vigorously defending the matter. MasTec will vigorously defend against any potential liability. Neither MasTec nor Jorge Mas were directly involved in any of the transactions the Spanish prosecutor alleges led to Sintel's bankruptcy. MasTec's directors' and officers' insurance carrier has agreed to fund legal expenses under a reservation of rights. The amount of loss to MasTec, if any, relating to this matter cannot presently be determined.

In addition to the matters discussed above, MasTec is subject to a variety of legal cases, claims and other disputes that arise from time to time in the ordinary course of its business. MasTec cannot provide assurance that it will be successful in recovering all or any of the potential damages it has claimed or in defending claims against it.

Accrued aggregate liabilities related to the matters described above and other litigation matters amounted to \$1.6 million at March 31, 2011 and \$2.4 million at December 31, 2010.

Other Commitments and Contingencies

Leases. In the ordinary course of business, the Company enters into non-cancelable operating leases for certain of its facility, vehicle and equipment needs, including related party leases. These leases allow the Company to conserve cash by paying a monthly rental fee for use of the related facilities, vehicles and equipment rather than purchasing them. The terms of these agreements vary from lease to lease, including some with renewal options and escalation clauses. The Company may decide to cancel or terminate a lease before the end of its term, in which case the Company is typically liable for the remaining lease payments under the term of the lease. Rent expense related to operating leases was approximately \$40.9 million and \$27.6 million for the three months ended March 31, 2011 and 2010, respectively.

Letters of Credit. In the ordinary course of business, the Company is required to post letters of credit, primarily for its insurance carriers and surety bond providers. Letters of credit are also posted in support of performance under certain contracts. Such letters of credit are generally issued by a bank or similar financial institution. The letter of credit commits the issuer to pay specified amounts to the holder of the letter of credit if the holder claims that the Company has failed to perform specified actions. If this were to occur, the Company would be required to reimburse the issuer of the letter of credit. Depending on the circumstances of such a reimbursement, the Company may also have to record a charge to earnings for the reimbursement. As of March 31, 2011 and December 31, 2010, the Company had \$97.5 million and \$85.8 million, respectively, of standby letters of credit issued under its senior secured credit facility. The Company is not aware of any material claims relating to outstanding letters of credit as of March 31, 2011.

Performance and Payment Bonds. In the ordinary course of business, MasTec is required by certain customers to provide performance and payment bonds for some of the Company's contractual commitments related to projects in process. These bonds provide a guarantee to the customer that the Company will perform under the terms of a contract and that the Company will pay subcontractors and vendors. If the Company fails to perform under a contract or to pay subcontractors and vendors, the customer may demand that the surety make payments or provide services under the bond. The Company must reimburse the surety for any expenses or outlays it incurs. In August 2010, the Company increased its performance bonding capacity to accommodate growth in its businesses with bonding requirements. As of March 31, 2011, the estimated cost to complete projects secured by the Company's \$463.6 million in performance and payment bonds was \$79.4 million. As of December 31, 2010, the estimated cost to complete projects secured by the Company's \$520.0 million in performance and payment bonds was \$93.9 million.

Included in the outstanding performance and payment bonds as of both March 31, 2011 and December 31, 2010 was approximately \$3.0 million relating to the projects and assets of the state Department of Transportation business sold by the Company in 2007. In connection with the sale of these assets, MasTec remains contingently liable for certain obligations, including liabilities associated with claims on these performance and payment bonds. The buyer of the Department of Transportation related projects and assets filed for bankruptcy protection in October 2009, which increases the likelihood that MasTec will be required to assume certain obligations associated with these projects. As of March 31, 2011, the remaining cost to complete the state Department of Transportation projects was immaterial.

Self-Insurance. MasTec maintains insurance policies subject to per claim deductibles of \$1 million for its workers' compensation policy, \$2 million for its general liability policy and \$2 million for its automobile liability policy. The Company has excess umbrella coverage up to \$100 million per claim and in the aggregate. Liabilities under these insurance programs are accrued based upon management's estimates of the ultimate liability for claims reported and an estimate of claims incurred but not reported, with assistance from third-party actuaries. These insurance liabilities are difficult to assess and estimate due to unknown factors, including the severity of an injury, the determination of the Company's liability in proportion to other parties and the number of incidents not reported. Accruals are based upon known facts and historical trends and management believes such accruals to be adequate. As of March 31, 2011 and December 31, 2010, MasTec's liability for unpaid claims and associated expenses, including incurred but not reported losses related to its workers compensation, general liability and automobile liability insurance policies, was \$38.7 million and \$36.3 million, respectively, of which \$25.4 million and \$22.9 million, respectively, was reflected within non-current other liabilities.

MasTec also maintains an insurance policy with respect to employee group health claims, which is subject to annual per employee maximum losses of \$0.4 million. MasTec's liability for employee group claims as of March 31, 2011 and December 31, 2010, which is based on statistical analysis of historical claims experience and specific knowledge of actual losses that have occurred, was \$1.3 million and \$1.4 million, respectively.

The Company is required to post letters of credit and provide cash collateral to its insurance carriers, or to obtain surety bonds in certain states in which the Company is self-insured. As of March 31, 2011 and December 31, 2010, these letters of credit amounted to \$59.4 million and \$59.5 million, respectively, a portion of which is collateralized by \$18.0 million of restricted cash as of both March 31, 2011 and December 31, 2010. In addition, cash collateral deposited with insurance carriers, which is included in other long term assets in the condensed unaudited consolidated balance sheets, amounted to \$2.2 million as of both March 31, 2011 and December 31, 2010. Outstanding surety bonds related to self-insurance programs amounted to \$7.5 million as of March 31, 2011 and \$4.6 million as of December 31, 2010.

Employment Agreements. The Company has employment agreements with certain executives and other employees, which provide for compensation and certain other benefits and for severance payments under certain circumstances. Certain employment agreements also contain clauses that become effective upon a change of control of the Company. Upon the occurrence of any of the defined events in the various employment agreements, the Company will pay certain amounts to the relevant employees, which vary with the level of the employees' respective responsibility.

Collective Bargaining Agreements. Certain of MasTec's subsidiaries are party to various collective bargaining agreements with unions representing certain of their employees. The agreements require the subsidiaries party to the agreements to pay specified wages, provide certain benefits to their union employees and contribute certain amounts to multi-employer pension plans and employee benefit trusts. The multi-employer pension plan contribution rates are determined annually and assessed on a "pay-as-you-go" basis based on union employee payrolls. The required amount of future contributions cannot be determined for future periods because the number of union employees employed at any given time and the plans in which they may participate vary depending upon the location and number of ongoing projects at a given time and the need for union resources in connection with those projects. The collective bargaining agreements expire at various times and have typically been renegotiated and renewed on terms similar to the ones contained in the expiring agreements.

The Employee Retirement Income Security Act of 1974, as amended by the Multi-Employer Pension Plan Amendments Act of 1980, imposes certain liabilities upon employers who are contributors to a multi-employer plan in the event of the employer's withdrawal from, or upon termination of, such plan. In the event of a complete or partial withdrawal from any of these plans, or if the plans were to otherwise become underfunded, the subsidiaries (and potentially MasTec itself) could be assessed liabilities for the withdrawal or for additional contributions related to the underfunding of these plans. None of the subsidiaries party to the collective bargaining agreements have any current plans to withdraw from the plans. The Pension Protection Act of 2006 also added new funding rules generally applicable to plan years beginning after 2007 for multi-employer plans that are classified as "endangered," "seriously endangered," or "critical" status. For a plan in critical status, additional required contributions and benefit reductions may apply. The Company has been notified that certain plans to which a MasTec subsidiary contributes are in "critical" status and require additional contributions in the form of a surcharge on future benefit contributions required for future work performed by union employees covered by these plans. The amount of additional funds MasTec may be obligated to contribute in the future cannot be estimated, as such amounts will be based on future levels of work that require the specific use of those union employees covered by these plans.

Indemnities. The Company generally indemnifies its customers for the services it provides under its contracts, as well as other specified liabilities, which may subject the Company to indemnity claims, liabilities and related litigation. As of March 31, 2011, the Company is not aware of any material asserted or unasserted claims in connection with these indemnity obligations.

Other Guarantees. In the ordinary course of its business, from time to time, MasTec guarantees the obligations of its subsidiaries and certain investees, including obligations under certain contracts with customers, certain lease obligations and in some states, obligations in connection with obtaining contractors' licenses. MasTec also generally warrants the work it performs for a one to two year period following substantial completion of a project. MasTec has not historically accrued any reserves for potential warranty claims as they have been immaterial.

Note 15 - Concentrations of Risk

The Company has approximately 210 customers as of March 31, 2011, which include some of the largest and most prominent companies in the communications, utilities and government industries. MasTec's customers include public and private energy providers, pipeline operators, wireless service providers, satellite and broadband operators, incumbent local exchange carriers, long distance carriers and government entities. The industries served by MasTec's customers include, among others: utilities (including wind farms, solar farms and other renewable energy; natural gas and petroleum pipeline infrastructure; and industrial infrastructure), communications (including wireless, wireline and satellite communications) and government (including water, sewer and other utility and communications work on military bases).

Revenues for customers by industry for the periods indicated are as follows (in millions):

		Three Months Ended			
		March 31,			
		2011		2010	
Communications	\$:	346.0	5 6%	\$224.8	50%
Utilities		264.4	43%	214.2	48%
Government		8.1	1%	11.2	2%
	\$6	518.5	100%	\$450.2	100%

Revenue concentration information, as a percent of total consolidated revenue, is as follows:

		Three Months Ended March 31.	
	2011	2010	
Revenue from top ten customers	82%	83%	
Revenue from specific customers:			
AT&T	27%	15%	
DIRECTV®	23%	27%	

In addition, the Company derived 20% of its revenues for the three months ended March 31, 2011 from El Paso Corporation and 19% of its revenues for the three months ended March 31, 2010 from Enbridge.

Foreign Operations. As of March 31, 2011, the Company does not have significant operations or long-lived assets in countries outside of the United States. For the three months ended March 31, 2011 and 2010, \$2.0 million and \$1.1 million, respectively, of revenues were derived from foreign operations, the majority of which were earned in Latin America and the Caribbean. In addition, the Company held property and equipment of \$1.6 million and \$1.4 million, respectively, in foreign countries as of March 31, 2011 and December 31, 2010, respectively. See Note 17 – Subsequent Events regarding the April 2011 acquisition of Fabcor TargetCo Ltd., a corporation incorporated under the laws of Alberta, Canada ("Fabcor Parent"), a Canadian pipeline construction company.

Note 16 - Related Party Transactions

MasTec purchases, rents and leases equipment used in its business from a number of different vendors on a non-exclusive basis, including Neff Corp. ("Neff"). Juan Carlos Mas, who was the chairman of the Neff Board of Directors through September 30, 2010, is the brother of Jorge Mas, Chairman of MasTec's Board of Directors, and Jose Mas, MasTec's Chief Executive Officer. For the three months ended March 31, 2010, MasTec paid Neff approximately \$153,000 for equipment purchases, rentals and leases. MasTec believes the amount paid to Neff is equivalent to the payments that would have been made between unrelated parties for similar transactions acting at arm's length. As of October 1, 2010, Juan Carlos Mas was no longer serving on the Neff Board of Directors.

MasTec leases a property located in Florida from Irma S. Mas, the mother of Jorge Mas and Jose Mas. For both the three months ended March 31, 2011 and 2010, the Company paid approximately \$12,000 for lease payments in connection with this property.

MasTec has an arrangement with a customer whereby it leases employees to that customer. For the three months ended March 31, 2011 and 2010, MasTec charged approximately \$108,000 and \$121,000, respectively, to the customer. As of March 31, 2011 and December 31, 2010, \$836,000 and \$887,000 respectively, is included as accounts receivable within other current assets from this customer. The Company also has an agreement with the same customer whereby the Company provides satellite communication services. For the three months ended March 31, 2011 and 2010, revenues relating to this customer were approximately \$345,000 and \$244,000, respectively. Jorge Mas and Jose Mas are minority owners of this customer. As of March 31, 2011 and December 31, 2010, approximately \$684,000 and \$599,000, respectively, is included as trade accounts receivable from this customer.

The Company charters an aircraft from a third party that leases two of its aircraft from entities in which Jorge Mas and Jose Mas have an ownership interest. MasTec paid this unrelated chartering company approximately \$23,000 and \$29,000 for the three months ended March 31, 2011 and 2010, respectively.

On October 28, 2009, MasTec combined the three existing split dollar agreements it had entered into with Jorge Mas into a restated single split dollar agreement. Under the amended and restated split dollar agreement, MasTec is the sole owner and beneficiary of each of the policies subject to the agreement and upon the death of the insured or insureds under the applicable policy, MasTec is entitled to receive a portion of the death benefit under such policy equal to the greater of (i) the total premiums paid by MasTec on such policy, or (ii) the then cash value of such policy immediately before the death of the insured or insureds, excluding surrender charges. Upon termination of the agreement, Jorge Mas, or in the case of a second to die policy, the second to die of Jorge Mas and his wife have an option to purchase each policy subject to the agreement for a purchase price equal to the greater of the amounts referenced above. The total maximum face amount of the insurance for all policies subject to the split dollar agreement was capped at \$200 million. MasTec will make the premium payments until the agreement is terminated, which occurs upon any of the following events: (i) bankruptcy or dissolution of MasTec, or (ii) a change in control of MasTec. In connection with the split dollar agreements for Jorge Mas, MasTec paid approximately \$284,000 for the three months ended March 31, 2011. No payments were made for the three months ended March 31, 2010.

On October 28, 2009, MasTec amended and restated an existing deferred bonus agreement it had entered into with Jorge Mas. The deferred bonus under the new agreement is equal to the sum of the following amounts, determined with respect to each policy subject to the split dollar agreement: the greater of (i) the total premiums paid by MasTec under such policy, or (ii) the then cash value of such policy immediately prior to the change in control, excluding surrender charges.

On October 28, 2009, the split dollar and deferred bonus agreements between Jose Mas and MasTec were amended and restated in their entirety. The amended and restated split dollar agreement provides that one or more life insurance policies may be subject to the agreement. At this time, however, the only policy subject to the agreement is the one acquired pursuant to the original agreement, which is in an aggregate face amount of \$11 million. MasTec is the sole owner and beneficiary of any policy subject to the agreement, and upon the death of the insured, MasTec is entitled to receive a portion of the death benefit equal to the greater of (i) the total premiums paid by MasTec on such policy, or (ii) the then cash value of such policy immediately before the death of the insured, excluding surrender charges. Upon termination of the agreement, Jose Mas has an option to purchase each policy subject to the agreement for a purchase price equal to the greater of the amounts referenced above. MasTec will make the premium payments until the agreement is terminated, which occurs upon any of the following events: (i) bankruptcy or dissolution of MasTec, or (ii) a change in control of MasTec. For the three months ended March 31, 2011, MasTec did not make any payments in connection with the split dollar agreements for Jose Mas. For the three months ended March 31, 2010, MasTec paid approximately \$115,000 in connection with the split dollar agreements for Jose Mas.

The amount of the deferred bonus under the amended and restated deferred bonus agreement that is payable upon termination of the split dollar agreement is equal to the sum of the following amounts, determined with respect to each policy subject to the split dollar agreement: the greater of (i) the total premiums paid by MasTec under the terms of such policy, or (ii) the then cash value of such policy immediately prior to the change in control, excluding surrender charges.

The Company adjusts the value of these life insurance policies each period based on their current cash surrender value. The estimated fair value of these life insurance policies of \$8.5 million and \$8.3 million at March 31, 2011 and December 31, 2010, respectively, is included in other assets in the condensed unaudited consolidated balance sheets.

Note 17 – Subsequent Events

Effective April 1, 2011, MasTec purchased 100% of the capital stock of Cam Communications Inc. ("Cam Com"), a Maryland company that provides telephone, cabling, engineering, construction, equipment integration, testing, wiring and computer network services to telecommunications carriers for approximately \$4.7 million plus a five-year earn-out, equal to 20% of the excess, if any, of Cam Com's annual EBITDA over \$2.25 million plus an additional one-time payment of up to \$1.5 million if Cam Com's EBITDA exceeds \$1.5 million in the first twelve months following the acquisition.

Effective April 1, 2011, MasTec acquired all of the issued and outstanding shares of Fabcor Parent for an aggregate purchase price composed of (i) approximately \$21.1 million in cash, plus the assumption of approximately \$8.8 million of debt, which was repaid immediately, and (ii) a five-year earnout (the "Earn-out") payable annually to the seller in cash, which will be paid in Canadian dollars. The Earn-out will be equal to 30% of the excess, if any, of Fabcor's annual EBITDA over 3.6 million Canadian dollars (approximately U.S. \$3.8 million as of March 31, 2011). Fabcor Parent, together with its wholly-owned subsidiaries, Fabcor 2001 Inc., a corporation incorporated under the laws of Alberta, Canada and Fabcor Pipelines BC Inc., a corporation incorporated under the laws of British Columbia, Canada, is engaged in providing pipeline and facility construction services in Alberta and British Columbia, Canada and its services include new pipeline construction, pipeline modification and replacement, river crossing construction and replacement, integrity excavation programs, well-site construction, compressor construction, gas plant construction, compressor and gas plant modifications and plant commissioning support services.

On April 29, 2011, the Company exercised the EC Source Merger Option as described in Note 4 – Acquisitions and Other Investments in the notes to the condensed unaudited consolidated financial statements, and effective May 1, 2011 (the "Effective Date") acquired the remaining 67% interest in EC Source for an aggregate purchase price composed of 5,129,644 shares of MasTec common stock, the assumption of \$8.6 million of debt and a five-year earn-out, payable at MasTec's election in common stock, cash or a combination thereof. The dollar amount of the earn-out will be equal to 20% of the excess, if any, of EC Source's annual EBITDA over \$15 million. The MasTec shares issued on the Effective Date are subject to transfer restrictions, which will lapse 25% on the first and second anniversaries of the closing and 50% on the third anniversary of the closing. Prior to closing, MasTec issued two standby letters of credit and a performance guarantee on behalf of EC Source. The EC Source sellers indemnified MasTec in connection with the letters of credit and guarantee for the period prior to the closing of the acquisition. The Company has determined that EC Source is a variable interest entity as of March 31, 2011, however, MasTec is not the primary beneficiary and therefore has not consolidated the financial position or earnings of EC Source as of this date. EC Source's assets, liabilities and earnings will be consolidated from the Effective Date.

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

This report contains forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995. These forward-looking statements are not historical facts but are the intent, belief, or current expectations of our business and industry and the assumptions upon which these statements are based. While management believes that these forward-looking statements are reasonable as and when made, there can be no assurance that future developments affecting us will be those that we anticipate. All comments concerning our expectations for future revenues and operating results are based on our forecasts for our existing operations and do not include the potential impact of any future acquisitions. Words such as "anticipates," "expects," "intends," "will," "could," "would," "should," "may," "plans," "believes," "seeks," "estimates" and variations of these words and the negatives thereof and similar expressions are intended to identify forward-looking statements. These statements are not guarantees of future performance and are subject to risks, uncertainties, and other factors, some of which are beyond our control, are difficult to predict, and could cause actual results to differ materially from those expressed or forecasted in the forward-looking statements.

These risks and uncertainties include those described in "Management's Discussion and Analysis of Financial Condition and Results of Operations," and elsewhere in this report and in the Company's Annual Report on Form 10-K for the year ended December 31, 2010, including those described under "Risk Factors" in the Form 10-K as updated by Item 1A "Risk Factors" in this report and other of our SEC filings. Forward-looking statements that were true at the time made may ultimately prove to be incorrect or false. Readers are cautioned to not place undue reliance on forward-looking statements, which reflect our management's view only as of the date of this report. We undertake no obligation to update or revise forward-looking statements to reflect changed assumptions, the occurrence of unanticipated events or changes to future operating results.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following is a discussion of our business, financial position and results of operations as of and for the three months ended March 31, 2011 and 2010. This discussion and analysis should be read in conjunction with the unaudited consolidated financial statements and notes contained in this Form 10-Q and the audited consolidated financial statements and accompanying notes contained in our Form 10-K for the year ended December 31, 2010.

About MasTec

We are a leading national infrastructure construction company operating mainly throughout North America across a range of industries. Our activities include the engineering, building, installation, maintenance and upgrade of energy, communication and utility infrastructure, such as: electrical utility transmission and distribution, wind farms, solar farms and other renewable energy, natural gas and petroleum pipeline infrastructure, wireless, wireline and satellite communications, industrial infrastructure and water and sewer systems. Our customers are primarily in the utility, communications and government industries. Including our predecessor companies, we have been in business for more than 80 years. We offer our services primarily under the MasTec service mark and operate through a network of more than 280 locations and have approximately 9,500 employees as of March 31, 2011. We have consistently ranked among the top specialty contractors by Engineering News-Record over the past five years.

We serve a diversified domestic customer base, and our customers include some of the leading communications and utility companies in the United States. As of March 31, 2011, our top ten customers are AT&T, DIRECTV®, El Paso Corporation, Edison Mission Energy, Energy Transfer Company, Talisman Energy, CenturyLink, Progress Energy, Dominion Resources and DCP Midstream Partners, L.P. We have longstanding relationships and have developed strong alliances with many of our customers. We strive to maintain these customer relationships and our status as a preferred vendor. Often, services are provided under multi-year master service agreements and other service agreements.

Because our business is often concentrated among our major customers, our business could be negatively impacted if the amount of business we obtain from them is reduced, or if we complete the required work on projects and cannot replace the work with similar projects. Revenue concentration information, as a percent of total consolidated revenue, is as follows:

	Three Month	Three Months Ended March 31,	
	March		
	<u>2011</u>	2010	
Revenue from top ten customers	82%	83%	
Revenue from specific customers:			
AT&T	27%	15%	
DIRECTV®	23%	27%	

Our relationship with AT&T is primarily based upon master service agreements, other service agreements and construction/installation contracts for AT&T's wireless and wireline infrastructure businesses. Our relationship with DIRECTV® is based upon two agreements to provide installation and maintenance services for DIRECTV® and, in support of the installation business, to provide marketing and sales services on behalf of DIRECTV®. Refer to "Recent Developments" for additional information regarding our purchase option agreement with Red Ventures LLC, which could reduce future revenues from DIRECTV®.

In addition, we derived 20% of our revenues for the three months ended March 31, 2011 from El Paso Corporation and 19% of our revenues for the three months ended March 31, 2010 from Enbridge. Our relationship with El Paso Corporation is based upon a construction contract for a natural gas pipeline. Our relationship with Enbridge is based upon construction/installation contracts for oil pipelines.

Recent Developments

Debt Exchange. During the first quarter of 2011, the Company exchanged \$105.3 million of its original 4.0% senior convertible notes and \$97 million of its original 4.25% senior convertible notes (together, the "Original Notes") for identical principal amounts of new 4.0% and 4.25% senior convertible notes (the "New Notes") for an exchange fee of approximately 50 basis points, or approximately \$1 million. The terms of the New Notes are substantially identical to those of the Original Notes, except that the New Notes have an optional physical, cash or combination settlement feature and contain certain conditional conversion features. Following the exchange, \$9.7 million and \$3.0 million of the original 4.0% and 4.25% notes, respectively, remained outstanding.

Because the New Notes have an optional cash settlement feature, and we intend to settle the principal amount in cash, the conversion shares underlying the principal amount of the New Notes are not included in our diluted share count. If, however, the average stock price per share exceeds the \$15.76 conversion price for the new 4.0% notes or the \$15.48 conversion price for the new 4.25% notes, then the resulting amount, in shares, of any premium will be included in our diluted share count ("premium shares"). For the quarter ended March 31, 2011, our average stock price exceeded the conversion prices of the New Notes, which resulted in the inclusion of 1.2 million premium shares in our diluted share count for the quarter ended March 31, 2011, as compared with 13.0 million shares associated with the corresponding principal amounts under the Original Notes for the quarter ended March 31, 2010. The reduction in diluted share count was partially offset by a \$1.1 million reduction in the after tax interest addback associated with the Original Notes under the "if-converted" method of accounting and \$0.6 million of after tax accretion expense associated with the New Notes. See Note 3 – Earnings Per Share and Note 9 – Debt in the notes to the condensed unaudited consolidated financial statements for additional details. The number of premium shares that will be included in our diluted share count in the future will vary with fluctuations in our share price. Higher actual share prices will result in additional premium shares included in our diluted share count. Because the number of shares required to be included in our diluted share count will vary with changes in our actual share price, we cannot predict the dilutive impact of any such premium shares on our earnings per share in future periods.

Notwithstanding our present intention to settle conversions of our New Notes in cash, we cannot assure you that we will be able to do so due to restrictions in our credit facility, which prior to its expiration in May 2013, without the consent of the lenders thereunder, prohibits us from borrowing under the credit facility to settle any conversions of the New Notes and from paying more than \$10 million in the aggregate from cash on hand to settle conversions of the New Notes. If we were required to settle conversions of our new senior convertible notes in accordance with our stated intent to settle principal amounts due in cash, and we were unable to do so with existing cash balances or through our credit facility, we could be required to obtain additional funding. We cannot be certain that such funding would be available on terms acceptable to us, or at all.

Amendment to Purchase Option Agreement. In February 2011, we entered into an amended and restated purchase option agreement with Red Ventures and certain of Red Ventures' affiliates, which grants Red Ventures a purchase option to acquire our wholly owned subsidiary, DirectStar, and its subsidiaries. As amended, the purchase option is exercisable from March 1, 2012 through November 30, 2012 and allows Red Ventures to pay up to 35% of the option purchase price in the form of a secured note with a one year term. In connection with the amendment to the purchase option, we agreed to increase certain commissions that DirectStar presently pays to Red Ventures for managing the DirectStar business. See Note 4 - Acquisitions and Other Investments in the notes to the condensed unaudited consolidated financial statements for additional details. DirectStar, in support of the DIRECTV * installation business, provides marketing and sales services on behalf of DIRECTV *. Should Red Ventures execute its purchase option agreement, our revenues from DIRECTV * would be reduced. Without DirectStar's revenues from services provided to DIRECTV *, our DIRECTV * revenues would have been reduced from 23% to 18% of total consolidated revenue for the three months ended March 31, 2011.

Acquisitions. In April 2011, we acquired Fabcor Parent, a Canadian construction pipeline company, for \$21.1 million in cash plus approximately \$8.8 million of assumed debt, which was repaid immediately. In addition, the purchase price includes a five-year earn-out payable annually to the seller in cash. We also acquired Cam Com, a company specializing in equipment construction and network services for telecommunications carriers, in April 2011 for approximately \$4.7 million in cash plus a five-year earn-out. Additionally, in April 2011, we exercised the EC Source Merger Option as described in Note 4 – Acquisitions and Other Investments in the notes to the condensed unaudited consolidated financial statements, and effective May 1, 2011, we acquired the remaining 67% interest in EC Source for an aggregate purchase price composed of 5,129,644 shares of MasTec common stock, the assumption of \$8.6 million in debt and a five-year earn-out, payable at our election in common stock, cash or a combination thereof. See Note 17 – Subsequent Events in the notes to the condensed unaudited consolidated financial statements for additional details.

Overview of Results

The first quarter of 2011 was a record first quarter for MasTec, with revenues of \$618.5 million. Revenue growth was broad-based, with strong results associated with many of the industries we serve, including wireless, pipeline, install-to-the-home and energy transmission. Overall, revenues grew 37% from the prior year period, all of which was organic growth. Our performance has been positively impacted by several factors, including strong end-user demand for new technologies in the wireless and install-to-the-home industries, continued execution of a large pipeline project and increased levels of activity in other areas of our business. While many areas of our business experienced growth, certain components were negatively impacted by competitive pressures and timing of project execution, which partially offset our revenue growth.

Net income increased by approximately \$13.7 million, or 185%, versus the same period in the prior year. Among the factors contributing to this growth were: revenue growth as previously discussed; improvement in our gross margin percentage; and improved leveraging of depreciation, amortization and general and administrative expenses as a percentage of revenue. Higher net income and a decrease in average diluted shares outstanding drove an increase in diluted earnings per share of \$0.16 per share, or 160%, versus the same period in 2010.

As discussed above, in the first quarter of 2011 we exchanged 94% of our previously outstanding senior convertible notes for new senior convertible notes. This exchange resulted in fewer diluted shares outstanding as of March 31, 2011 as compared with the prior year period. See Note 9 – Debt and Note 3 – Earnings Per Share in the notes to the condensed unaudited consolidated financial statements for additional details. The decrease in average diluted shares outstanding was partially offset by the issuance in December 2010 of approximately 1.9 million shares in connection with an amendment to a purchase agreement for one of our historical acquisitions. Refer to Note 4 – Acquisitions and Other Investments in the notes to the consolidated financial statements of our 2010 Form 10-K for additional details.

Revenue

We provide engineering, building, installation, maintenance and upgrade services to our customers. The primary industries served by our customers are communications, utilities and government. Customer revenues by industry for the periods indicated were as follows (in millions):

		For the Three Months Ended March 31,			
	201	1	2010		
Communications	\$ 346.0	56%	\$224.8	50%	
Utilities	264.4	43%	214.2	48%	
Government	8.1	1%	11.2	2%	
	\$618.5	100%	\$450.2	100%	

We enter into various types of contracts, including unit price, time and materials, cost-plus and fixed price contracts. Under unit price contracts, we agree to perform work for a price per unit of work performed. Fixed price contracts provide for a fixed fee for an entire project, the final terms and prices of which can be subject to negotiation with the customer, as is customary in the industry. Work performed on a time and materials basis is billed based upon negotiated hourly rates plus material costs. Cost-plus contracts are based on actual costs incurred for materials, equipment and labor, plus an agreed upon markup. Although the terms of our contracts vary considerably, most are on either a unit price or a fixed price basis. We complete a substantial majority of our fixed price projects within one year, while we frequently provide maintenance, install and repair work under unit price or fixed price master service or other service agreements that are renewable periodically.

We recognize revenue on long term construction contracts on a percentage of completion basis. Some of our long-term construction projects are pursuant to master service or other service agreements. Under fixed price contracts, revenues are recognized as work on the contract progresses using the cost-to cost method. Under this method, revenue is recognized based on the percentage of total costs incurred to date in proportion to total estimated costs to complete the contract. Fixed price contracts generally include retainage provisions under which a percentage of the contract price is withheld until the project has been completed and the work has been accepted by our customer. For projects that are of a short-term nature, primarily under master service or other service agreements, revenue is recognized when units are completed or when the related services are performed.

Over half of our revenues are derived from projects performed under master service and other service agreements, which are generally multi-year agreements. Certain of our master service agreements are exclusive up to a specified dollar amount per work order for each defined geographic area, but do not obligate our customers to undertake any larger infrastructure projects or other work with us. Work performed under master service and other service agreements is typically generated through work orders, each of which is performed for a fixed fee. Services provided under these agreements range from engineering, project management and installation work to maintenance and upgrade services. Master service and other service agreements are frequently awarded on a competitive bidding basis, although customers are sometimes willing to negotiate contract extensions beyond their original terms without rebidding. Our master service agreements and other service agreements have various terms, depending upon the nature of the services provided, and typically provide for termination on short or no advance notice.

The remainder of our work is generated pursuant to contracts for specific projects or jobs that may require the construction and installation of an entire infrastructure system or specified units within an infrastructure system. Customers are billed with varying frequency, generally monthly or upon attaining specific milestones. Such contracts generally include retainage provisions under which 2% to 15% of the contract price is withheld from us until the work has been completed and accepted by the customer.

Revenues by type of contract for the periods indicated were as follows (in millions):

	F	For the Three Months Ended March 31,		
	2011		2010	
Master service and other service agreements	\$367.6	59%	\$ 230.2	51%
Installation/construction project agreements	250.9	41%	220.0	49%
	\$618.5	100%	\$450.2	100%

As shown in the table above, over one-third of our first quarter 2011 revenues were from non-recurring, project specific work. Seasonality tends to have a greater impact on our non-recurring project revenues. The proportion of our revenues from non-recurring project work in any given quarter can fluctuate based upon our project mix. If we are not able to replace work from completed projects with new project work, we may not be able to maintain our current revenue levels, or our current level of capacity and resource utilization. We actively review our backlog of project work and take appropriate action to minimize such exposure.

Backlog

Estimated backlog represents the amount of revenue we expect to realize over the next 18 months from future work on uncompleted contracts, including new contractual agreements on which work has not begun. Our backlog estimates include amounts under master service and other service agreements in addition to construction projects. We determine the amount of backlog for work under master service and other service agreements based on historical trends, anticipated seasonal impacts and estimates of customer demand based on communications with our customers. The following presents 18-month backlog for our business as of March 31, 2011 and December 31, 2010 (in millions):

	As of	As of
	March 31, 2011 (1)(2)	December 31, 2010 (2)
Estimated 18-month backlog	\$ 2,576.5	\$ 2,354.5

- (1) Backlog associated with businesses acquired subsequent to March 31, 2011 is not reflected in our backlog estimate. See Note 17 – Subsequent Events in the notes to the condensed unaudited consolidated financial statements for details of businesses acquired subsequent to March 31, 2011.
- (2) Our 18-month backlog estimates as of March 31, 2011 and December 31, 2010 do not include approximately \$300 million of revenues from a project we have been awarded that is subject to financing contingencies.

While our backlog estimates include amounts under master service and other service agreements, our customers are not contractually committed to purchase a minimum amount of services under these agreements, most of which can be cancelled on short or no advance notice. There can be no assurance as to our customers' requirements or that our estimates are accurate. In addition, timing of revenues for construction and installation projects included in our backlog can be subject to change as a result of customer delays, regulatory requirements and other project related factors. These changes could cause estimated revenues to be realized in periods later than originally expected, or not at all. As a result, our backlog as of any particular date is an uncertain indicator of future revenues and earnings.

Critical Accounting Policies and Estimates

This discussion and analysis of our financial condition and results of operations is based upon our condensed unaudited consolidated financial statements, which have been prepared in accordance with U.S. GAAP. The preparation of these financial statements requires us to make estimates and judgments that affect the amounts reported in our financial statements and the accompanying notes. On an on-going basis, we evaluate our estimates, including those related to revenue recognition, provisions for contract losses, allowances for doubtful accounts, securities available for sale, goodwill and intangible assets, acquisition-related contingent consideration, reserves and accruals, impairment of assets, income taxes, insurance reserves, litigation and contingencies. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances, the results of which form the basis of making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. As management estimates, by their nature, involve judgment regarding future uncertainties, actual results may differ from these estimates if conditions change or if certain key assumptions used in making these estimates ultimately prove to be materially incorrect. Refer to Note 2 - Basis of Presentation and Significant Accounting Policies in the notes to our condensed unaudited consolidated financial statements of this Quarterly Report on Form 10-Q and to our most recent Annual Report on Form 10-K for further information regarding our critical accounting policies and estimates.

Results of Operations

Economic, Industry and Market Factors

Despite the results we achieved in the first quarter of 2011, we recognize that we continue to operate in a challenging business environment, as do our customers. We closely monitor the effect that changes in economic and market conditions may have on our customers. General economic conditions, as well as the highly competitive nature of our industry, have resulted in pricing pressure for the services we provide. Work is often awarded through a bidding process, where price is often a principal factor in the selection process. In the face of increased pricing pressure, we strive to maintain our profit margins through productivity improvements and cost reduction programs. We also operate in an industry affected by market and regulatory impacts beyond our control. Changes in renewable energy portfolio standards, new or changing regulatory requirements affecting the industries we serve, and/or changes in technology can impact demand for our services. In addition, fluctuations in market prices for oil, gas and other fuel sources can impact demand for our pipeline and renewable energy construction services. We actively monitor economic, industry and market factors affecting our business, however, we cannot predict the impact such factors may have on our future results of operations, liquidity and cash flows.

Impact of Seasonality and Cyclical Nature of Business

Our revenues and results of operations can be subject to seasonal and other variations. These variations are influenced by weather, customer spending patterns, bidding seasons, project schedules and timing and holidays. Typically, our revenues are lowest in the first quarter of the year because cold, snowy or wet conditions cause delays. Revenues in the second quarter are typically higher than in the first quarter, as some projects begin, but continued cold and wet weather can often impact second quarter productivity. The third and fourth quarters are typically the best of the year, as a greater number of projects are underway and weather is normally more accommodating to work on projects. In the fourth quarter, many projects tend to be completed by customers seeking to spend their capital budget before the end of the year, which generally has a positive impact on our revenues. However, the holiday season and inclement weather can cause delays, which could reduce revenues and increase costs on affected projects. Any quarter may be positively or negatively affected by out of the ordinary weather patterns, such as excessive rainfall or warm winter weather, making it difficult to predict quarterly revenue variations.

Additionally, our industry can be highly cyclical. Fluctuations in end-user demand within the industries we serve, or in the supply of services within those industries, can impact demand for our services. As a result, our business may be adversely affected by declines or delays in new projects. Project schedules, in particular in connection with large construction and installation projects, can create fluctuations in revenues, which may adversely affect us in a given period. The financial condition of our customers and their access to capital, variations in project gross margins, regional, national and global economic and market conditions as well as acquisitions or strategic investments can also materially affect quarterly results. Accordingly, our operating results in any particular period may not be indicative of the results that can be expected for any other period.

Gross Margin

Gross margin is our gross profit expressed as a percentage of revenues. Costs of revenues, which is subtracted from revenues to obtain gross profit, consists primarily of salaries, employee wages and benefits, subcontracted services, equipment rentals and repairs, fuel and other equipment expenses, material costs, parts and supplies, insurance and facilities expenses. Factors impacting our gross margin include:

Revenue Mix. The mix of revenues derived from the projects we perform impacts margins, as certain projects provide higher margin opportunities. Installation work is often obtained on a fixed price basis, while maintenance work is often performed under pre-established or time and materials pricing arrangements. Margins for installation work may vary from project to project, and can be higher than maintenance and upgrade work due to the fact that fixed price contracts often have a higher level of risk than other types of project work. Changes in project mix between installation work and maintenance or upgrade services can impact our margins in a given period. Additionally, the mix of project revenues by industry served can also have an impact on margins, as margin opportunities can vary by industry and over time.

Seasonality, Weather and Geographic Mix. As discussed above, seasonal patterns can have a significant impact on margins. Generally, business is slower at the beginning of the year. Adverse or favorable weather conditions can impact margins in a given period. For example, extended periods of rain or snowfall can negatively impact revenues and margins as a result of reduced productivity from projects being delayed or temporarily placed on hold. Conversely, in periods when weather remains dry and temperatures are accommodating, more work can be done, sometimes with less cost, which can favorably impact margins. In addition, the mix of business conducted in different parts of the country can affect margins due to the geographic characteristics associated with the physical location where the work is being performed, such as mountainous or rocky terrain versus open terrain. Site conditions, including unforeseen underground conditions, can also impact margins.

Performance Risk. Margins may fluctuate due to the volume of work performed, project pricing and job productivity. Job productivity can be impacted by weather, geography, customer decisions and crew productivity. Crew productivity can be influenced by job terrain, for example, whether project work is in an open or encumbered right of way. Additionally, weather conditions and other factors can also have a direct impact on job and crew productivity, which affects project margins.

Subcontracted Resources. Our use of subcontracted resources in a given period varies, based upon activity levels and the amount and location of existing in-house resources and capacity. Work that is subcontracted can yield lower margins than self-perform work. As a result, changes in the mix of subcontracted versus self-perform work can impact our margins.

Material versus Labor Costs. In many cases, our customers are responsible for supplying their own materials on projects; however, under certain contracts, we may agree to provide all or part of the required materials. Margins are typically lower on projects where we furnish a significant amount of materials, due to the fact that mark-ups on materials are generally lower than on labor costs. Therefore, increases in the percentage of work with significant materials requirements could decrease our overall margins.

Insurance Costs. Margins can also be impacted by insurance costs as additional claims arise and as circumstances and conditions of existing claims change. We maintain insurance policies subject to per claim deductibles of \$1 million for our worker's compensation policy and \$2 million for our general liability and automobile liability policies. We also have employee healthcare benefit plans for our employees not subject to collective bargaining agreements, which are subject to a deductible of \$0.4 million per claimant per year.

Selling, General and Administrative Expense

General and administrative expenses consist primarily of compensation and benefit expenses, travel expenses and related costs for our finance, benefits and risk, legal, facilities, information services and executive personnel. General and administrative expenses also include outside professional and accounting fees, transaction expenses, expenses associated with computer equipment and software used in the administration of the business and various forms of insurance.

Interest Expense, Net

Interest expense, net, consists of contractual interest expense on outstanding debt obligations, amortization of deferred financing costs, debt discount and investor fee accretion associated with our new senior convertible notes, line of credit and letter of credit fees, offset by interest earned on cash, cash equivalents and fixed income investments.

Other Income, Net

Other income, net, consists primarily of gains or losses from sales of assets and investments, income or losses from equity method investments as well as any other-than-temporary impairment losses recognized in connection with our available for sale auction rate securities.

Litigation and Contingencies

We are currently involved in various claims and legal proceedings. Quarterly, we review the status of each significant matter and assess our potential financial exposure. We accrue a liability for an estimated loss if the potential loss from any claim or legal proceeding is considered probable and the amount can be reasonably estimated. Significant judgment is required in both the determination of probability and the determination as to whether the amount of an exposure is reasonably estimable. Because of uncertainties related to these matters, accruals are based only on the information available at the time. As additional information becomes available, we reassess the potential liability related to our pending claims and litigation and may revise our estimates. Such revisions in estimates of the potential liabilities could have a material impact on our results of operations and financial position.

See Note 14 - Commitments and Contingencies in the notes to our condensed unaudited consolidated financial statements for additional information regarding legal proceedings and contingencies.

Comparison of Quarterly Results

The following table reflects our consolidated results of operations in dollar and percentage of revenue terms for the periods indicated (dollar amounts in thousands):

		For the Three Months Ended March 31,		
	2011		2010	
Revenue	\$618,492	100.0%	\$ 450,231	100.0%
Costs of revenue	528,553	85.5	388,855	86.4
Gross profit	89,939	14.5	61,376	13.6
Depreciation and amortization	15,109	2.4	14,175	3.2
General and administrative expenses	32,499	5.2	27,673	6.1
Interest expense, net	7,912	1.3	7,376	1.6
Other income, net	(165)	(0.0)	(297)	(0.0)
Income before provision for income taxes	34,584	5.6	12,449	2.7
Provision for income taxes	(13,488)	(2.2)	(5,054)	(1.1)
Net income	21,096	3.4	7,395	1.6
Net loss attributable to non-controlling interests	(10)	(0.0)	(16)	(0.0)
Net income attributable to MasTec	\$ 21,106	3.4%	\$ 7,411	1.6%

Three Months Ended March 31, 2011 Compared to Three Months Ended March 31, 2010

Revenue. Our revenue was \$618.5 million for the three months ended March 31, 2011, as compared with \$450.2 million for the same period in 2010, representing an increase of \$168.3 million or 37.4%, all of which was organic growth. Revenues were favorably impacted by our wireless, pipeline, install-to-the-home and energy transmission customers. Key customers driving growth in the first quarter of 2011 included AT&T, El Paso Corporation and DIRECTV®. In the first quarter of 2011, the wireless industry continued to invest in new infrastructure for its 4G technology. This industry trend, in addition to expansion of the geographic areas in which we do project work, yielded higher revenues from wireless projects in the first quarter of 2011 as compared with 2010. Revenue growth from wireless projects was offset slightly by certain pricing discounts that became effective in the first quarter of 2011. We also experienced an increase in revenues from pipeline projects in the first quarter of 2011. The increase was driven by continued execution of a large natural gas pipeline project with El Paso Corporation, as well as from increased activity levels on other natural gas pipeline projects. Install-to-the-home project activity was strong in the first quarter of 2011. While first quarter revenue growth was strong in many components of our business, other areas experienced lower activity versus the prior year, due in part to competitive pressures and timing of project execution, which partially offset our revenue growth.

Costs of revenue. Our costs of revenue were \$528.6 million, or 85.5% of revenue, for the three months ended March 31, 2011, as compared with \$388.9 million, or 86.4% of revenue, for the corresponding period in 2010, a \$139.7 million increase, or 35.9%. The dollar increase is attributable to higher costs associated with increased revenues, as described above. As a percentage of revenue, costs of revenue decreased 90 basis points. This improvement was primarily attributable to lower material costs as a percent of revenue due to our mix of project revenues. The decrease in material costs was partially offset by increases in other types of expenses, including fuel. Higher fuel costs were driven by higher fuel prices as well as increased levels of fuel consumption to support growth in project work. Other expense items, such as wages, including subcontractor expense, and equipment maintenance, increased slightly as a percent of revenue due to changes in our project mix. Additionally, as discussed under "Recent Developments," beginning in the first quarter of 2011, certain commissions that we pay in connection with our DIRECTV * installation projects have increased, which partially offset our overall gross margin improvement.

Depreciation and amortization. Depreciation and amortization was \$15.1 million for the three months ended March 31, 2011, as compared with \$14.2 million for the same period in 2010, representing an increase of \$0.9 million, or 6.6%. The increase was driven by depreciation expense attributable to certain leased equipment classified as capital leases, which had previously been classified as operating leases. Modifications to the lease terms effective in 2011 caused the reclassification of these leases. The resulting increase in depreciation expense is offset by a decrease in rent expense, which is recorded within costs of revenue. See Note 9 – Debt in the notes to the unaudited consolidated financial statements for additional details. The increase in depreciation expense was partially offset by a decrease in amortization expense from historical acquisitions.

General and administrative expenses. General and administrative expenses were \$32.5 million, or 5.2% of revenue, for the three months ended March 31, 2011, as compared with \$27.7 million, or 6.1% of revenue, for the same period in 2010, representing an increase of \$4.8 million, or 17.4%. The dollar increase resulted largely from higher labor and other costs in support of the growth in the business. As a percentage of revenue, general and administrative expenses decreased approximately 90 basis points. This improvement is primarily attributable to improved leveraging of administrative labor as a result of our revenue growth.

Interest expense, net. Interest expense, net of interest income, was \$7.9 million, or 1.3% of revenue, for the three months ended March 31, 2011, as compared with approximately \$7.3 million, or 1.6% of revenue, for the same period in 2010, representing an increase of approximately \$0.6 million. This increase is primarily attributable to accretion expense from the debt discount and investor fees associated with our new 4.0% and 4.25% convertible notes, which was partially offset by a decrease in interest expense on equipment notes. In the fourth quarter of 2010, we refinanced \$13.4 million of 7.05% equipment notes with a new equipment note bearing interest at approximately 3.53%.

Other income, net. Other income, net, was \$0.2 million for three months ended March 31, 2011, as compared with \$0.3 million for the three months ended March 31, 2010. The reduction in other income, net, is primarily attributable to lower gains on sales of assets versus the same period in the prior year, partially offset by income on equity method investments.

Income taxes. Income taxes were \$13.5 million for the three months ended March 31, 2011, as compared with \$5.1 million for the three months ended March 31, 2010, representing an increase of \$8.4 million. This increase is primarily due to higher taxable income, partially offset by the impact of a lower tax rate. Our effective tax rate was 39.0% for the three months ended March 31, 2011 as compared with an effective tax rate of 40.6% for the three months ended March 31, 2010. The lower current year tax rate is attributable to higher utilization of certain tax deductions for which we first became eligible during 2010. Prior to 2011, these deductions were only available to us on a limited basis as we were benefiting from the use of net operating loss carryforwards.

Earnings Before Interest, Taxes, Depreciation and Amortization ("EBITDA")

EBITDA is a non-GAAP financial measure that reflects earnings excluding the impact of interest expense, provision for income taxes, depreciation and amortization. We use EBITDA, as we believe analysts, lenders, investors and others do, to evaluate our performance, both internally and versus our peers, because it excludes certain items that can vary widely across different industries or among companies within the same industry. For example, interest expense can be dependent upon a company's capital structure, debt levels and credit ratings. Accordingly, the impact of interest expense on earnings can vary significantly among companies. During the first quarter of 2011, our interest expense increased due to the inclusion of accretion expense associated with our new senior convertible notes, which contain an optional physical, cash or combination conversion feature. Issuers with convertible debt instruments that can be settled in cash or other assets on conversion are required to separately account for the liability and equity components of the instrument in a manner that reflects what the entity's borrowing rate would have been on identical, but nonconvertible debt as of the instrument's issuance date. As a result, we separated the principal balance of the new senior convertible notes between the fair value of the debt component and the fair value of the common stock conversion feature. We discounted the values of the notes at an estimated rate of 6.73%. The resulting debt discount will be accreted to interest expense over the remaining terms of the new senior convertible notes using the effective interest method, which will increase interest expense above the New Notes' 4.0% and 4.25% cash coupon interest rates.

Tax positions of companies can also vary because of their differing abilities to take advantage of tax benefits and because of the tax policies of the jurisdictions in which they operate. As a result, effective tax rates and provisions for income taxes can vary considerably among companies and over time. For example, in 2011, we expect our effective tax rate to decrease to 39.0% from 40.6% in 2010. The lower current year tax rate is attributable to an increase in our ability to utilize certain tax deductions.

EBITDA also excludes depreciation and amortization because companies utilize productive assets of different ages and use different methods of both acquiring and depreciating productive assets. These differences can result in considerable variability in the relative costs of productive assets and depreciation and amortization expense among companies. EBITDA should not be considered in isolation from, and is not intended to represent an alternative measure of, operating results or net income as determined in accordance with U.S. GAAP. EBITDA is also not defined below as it is in our senior secured credit facility or in the indentures governing our notes; therefore, EBITDA as presented in this discussion should not be used for purposes of determining our compliance with related covenants.

Management also considers EBITDA to be an indicator of liquidity because we use it to measure our ability to service debt, fund capital expenditures and expand our business. The following table reflects a reconciliation of our EBITDA in dollar and percentage of revenue terms for the periods indicated (dollar amounts in millions). The table below may contain slight summation differences due to rounding.

		March 31,		
	2011		201	0
EBITDA Reconciliation:				
Net income	\$ 21.1	3.4%	\$ 7.4	1.6%
Interest expense	7.9	1.3%	7.3	1.6%
Provision for income taxes	13.5	2.2%	5.1	1.1%
Depreciation and amortization	15.1	2.4%	14.2	3.2%
Earnings before interest, taxes, depreciation and amortization ("EBITDA")	\$57.6	9.3%	\$ 34.0	7.6%

Three Months Ended

The table below summarizes our effective tax rates for the periods indicated:

	I nree Mon	i nree Months Ended		
	Marc	ch 31,		
	2011	2010		
Effective tax rate	39.0%	40.6%		

Financial Condition, Liquidity and Capital Resources

Our primary sources of liquidity are cash flows from continuing operations, availability under our credit facility and our cash balances. Our primary liquidity needs are for working capital, income taxes, capital expenditures, insurance collateral in the form of cash and letters of credit, earn-out obligations and debt service. We also evaluate opportunities for strategic acquisitions and/or investments from time to time that may require cash, and may consider opportunities to either repurchase outstanding debt or repurchase outstanding shares of our common stock in the future.

Capital Expenditures. We estimate that we will spend approximately \$40 million in 2011 on capital expenditures. The increase versus prior year capital expenditures of approximately \$30 million is due to higher expected business activity levels in 2011 as compared with 2010. The nature of our business is equipment intensive and actual capital expenditures can increase or decrease from estimates depending upon business activity levels. We will continue to evaluate lease versus buy decisions to meet our equipment needs and based on this evaluation, our capital expenditures may increase or decrease in the future. In January 2011, we modified the terms of certain of our equipment operating leases. The modifications changed the classification of the corresponding leases from operating to capital leases. As a result, \$24.0 million of capital lease assets and corresponding capital lease obligations were recorded as of January 1, 2011. We also expect to continue to sell older equipment as we upgrade to new equipment.

Earn-out Payments. We have made certain acquisitions and have agreed to pay earn-out payments to certain of the sellers, generally based on the future performance of the acquired businesses. Certain of these earn-out payments may be made either in cash or, under certain circumstances, MasTec common stock, or a combination thereof, at our option. For the three months ended March 31, 2011 and 2010, we made cash payments of \$17.5 million and \$9.3 million, respectively, related to such earn-out obligations.

Income Taxes. Our income tax expense increased from \$5.1 million for the three month period ended March 31, 2010 to \$13.5 million for the same period in 2011. Our cash tax payments will increase in 2011 as we fully utilized our available net operating loss carryforwards in the fourth quarter of 2010.

Working Capital. We need working capital to support seasonal variations in our business, primarily due to the impact of weather conditions on external construction and maintenance work and the spending patterns of our customers, both of which influence the timing of associated spending to support related customer demand. Our business is typically slowest in the first quarter of each calendar year and we generally experience seasonal working capital needs from approximately April through October to support growth in unbilled revenue and accounts receivable, and to a lesser extent, inventory. Our billing terms are generally net 30 days, and some of our contracts allow our customers to retain a portion (from 2% to 15%) of the contract amount until the job is completed. For certain customers, we maintain inventory to meet the material requirements of the contracts. Occasionally, certain of our customers pay us in advance for a portion of the materials we purchase for their projects, or allow us to pre-bill them for materials purchases up to specified amounts. Vendor terms are generally 30 days. Certain agreements with subcontractors contain a "pay-when-paid" provision, whereby our payments to subcontractors are made only after we are paid by our customers.

Debt Service. As of March 31, 2011, our common stock trading price exceeded the \$15.76 and \$15.48 conversion prices of our outstanding 4.0% and 4.25% senior convertible notes, respectively. Currently, we intend to settle the principal amounts of our new senior convertible notes upon any conversion thereof in cash. As of March 31, 2011, we had outstanding \$202.3 million aggregate principal amount of our new senior convertible notes. Notwithstanding our present intention to settle conversions of our new senior convertible notes in cash, we cannot assure you that we will be able to do so due to restrictions in our credit facility, which prior to its expiration in May 2013, without the consent of the lenders thereunder, prohibits us from borrowing under the credit facility to settle any conversions of the new senior convertible notes and from paying more than \$10 million in the aggregate from cash on hand to settle conversions of the new senior convertible notes. If we were required to settle conversions of our new senior convertible notes in accordance with our stated intent to settle principal amounts due in cash, and we were unable to do so with existing cash balances or through our credit facility, we could be required to obtain additional funding. We cannot be certain that such funding would be available on terms acceptable to us, or at all.

Summary of Financial Condition, Liquidity and Capital Resources

General U.S. and global economic conditions have not had a significant impact on our overall financial position, results of operations or cash flows for the three months ended March 31, 2011. Given the generally good credit quality of our customer base, we do not expect collections issues to materially impact our liquidity in the next twelve months. In addition, as a result of our current capital structure, including our credit facility, we do not anticipate that current overall market and economic conditions will materially impact our liquidity. We anticipate that funds generated from our operations, borrowings under our credit facility and our cash balances will be sufficient to meet our working capital requirements, anticipated capital expenditures, insurance collateral requirements, earn-out obligations, letters of credit and debt service obligations for at least the next twelve months.

Sources and Uses of Cash

As of March 31, 2011, we had \$263.4 million in working capital, defined as current assets less current liabilities, compared to \$235.1 million as of December 31, 2010. Cash and cash equivalents, including approximately \$18.3 million and \$18.0 million of restricted cash at March 31, 2011 and December 31, 2010, respectively, increased from \$177.6 million at December 31, 2010 to \$198.0 million at March 31, 2011. Restricted cash related to collateral for certain letters of credit is invested in certificates of deposit with a maturity of 90 days.

Sources and uses of cash are summarized below (in millions):

		For the Three Months Ended March 31,		
	2011	2010		
Net cash provided by operating activities	\$ 50.3	\$ 34.7		
Net cash used in investing activities	(31.1)	(14.0)		
Net cash provided by (used in) financing activities	1.2	(6.4)		

Operating Activities. Cash flow from operations is primarily influenced by demand for our services, operating margins and the type of services we provide, but can also be influenced by working capital needs such as the timing of collection of receivables and the settlement of payables and other obligations. Working capital needs are generally higher from April through October due to the seasonality of our business. Conversely, a portion of working capital assets are typically converted to cash in the first quarter. Net cash provided by operating activities increased by \$15.6 million to \$50.3 million for the three months ended March 31, 2011. This increase was primarily driven by higher net income. To a lesser extent, changes in working capital assets and liabilities, as well as deferred taxes, also contributed to the growth in operating cash flows. Net changes in accounts receivable and billings in excess of costs and earnings balances were partially offset by increases in accounts payable and accrued expenses. Accounts payable and accrued expenses increased as a result of timing of payroll and accounts payable disbursements. Accounts receivable and billings in excess of costs and earnings, the balances of which are primarily driven by timing of project activity, contractual billing terms and pace of collections, decreased.

Investing Activities. Net cash used in investing activities increased by \$17.1 million to \$31.1 million for the three months ended March 31, 2011. The increase was primarily driven by an increase in cash used for payments of earn-out obligations and an increase in capital expenditures to support growth in our business. As discussed in Note 17 – Subsequent Events in the notes to the condensed unaudited consolidated financial statements, in April 2011, we acquired Cam Com and Fabcor Parent. Purchase price consideration for these acquisitions included \$25.8 million in cash.

Financing Activities. We had \$1.2 million of cash provided by financing activities for the three months ended March 31, 2011, as compared with \$6.4 million of cash used in financing activities for the three months ended March 31, 2010. The \$7.6 million decrease in cash used in financing activities is primarily attributable to an increase in cash received from stock option exercises and \$2.9 million of related tax benefits. During the first quarter of 2011, approximately 667,000 stock options were exercised for cash proceeds of approximately \$6.6 million, as compared with cash proceeds for stock options of \$0.7 million in the same period in 2010. The increase in cash proceeds and tax benefits from stock option exercises was partially offset by the payment of \$1.0 million in the first quarter of 2011 for investor fees related to our debt exchange. In connection with the acquisition of Fabcor Parent in April 2011, we assumed \$8.8 million of debt, which was repaid immediately. Additionally, in connection with the acquisition of EC Source in May 2011, we assumed \$8.6 million of debt. See Note 17 – Subsequent Events in the notes to the condensed unaudited consolidated financial statements for additional information.

Credit Facility

We have a senior secured credit facility, which we refer to as our Credit Facility. Our Credit Facility has a maximum amount of available borrowing capacity of \$210 million, subject to certain restrictions, and expires in May 2013. The maximum available borrowing capacity may be increased to \$260 million, provided that we are in compliance with all provisions and covenants of the Credit Facility and that such additional borrowing would not violate the indenture that governs our senior notes. We may consider increasing our borrowing capacity to \$260 million as provided under the Credit Facility, and/or further expanding our borrowing capacity to accommodate future business growth. Because our existing lenders are not obligated to provide additional capacity, however, our ability to secure additional financing cannot be assured. At March 31, 2011, we did not have an outstanding balance on the Credit Facility for borrowed money.

The amount we can borrow under our Credit Facility at any given time is based upon a formula that takes into account, among other things, eligible billed and unbilled accounts receivable, equipment, real estate and eligible cash collateral, which can result in borrowing availability of less than the full amount of the Credit Facility. At March 31, 2011 and December 31, 2010, net availability under the Credit Facility totaled \$89.3 million and \$124.2 million, respectively, net of outstanding standby letters of credit aggregating \$97.5 million and \$85.8 million at March 31, 2011 and December 31, 2010, respectively. These letters of credit mature at various dates and most have automatic renewal provisions subject to prior notice of cancellation. The Credit Facility is collateralized by a first priority security interest in substantially all of our assets and the assets of our wholly-owned subsidiaries and a pledge of the stock of certain of our operating subsidiaries.

Interest on outstanding balances on the Credit Facility accrues at variable rates based, at our option, on the agent bank's base rate (as defined in the Credit Facility) plus a margin of between 1.25% and 1.75%, or at LIBOR plus a margin of between 2.25% and 3.00%, depending on certain financial thresholds. At March 31, 2011, the margin over LIBOR was 2.25% and the margin over the base rate was 1.25%. In addition, the Credit Facility includes an unused facility fee ranging from 0.5% to 0.75% based on usage. At March 31, 2011, the unused facility fee was 0.75%.

Our Credit Facility contains customary events of default (including cross-default) provisions and covenants related to our operations that prohibit, among other things, making investments and acquisitions in excess of specified amounts, incurring additional indebtedness in excess of specified amounts, creating liens against our assets, prepaying subordinated indebtedness and engaging in certain mergers or combinations without the prior written consent of the lenders. The Credit Facility also limits our ability to make certain distributions or pay dividends. In addition, we are required to maintain a minimum fixed charge coverage ratio of 1.2 to 1.0, which is calculated as available cash flow divided by debt carrying costs, as defined in the Credit Facility. Any deterioration in the quality of billed and unbilled receivables, reduction in the value of our equipment or an increase in our lease expense related to real estate would reduce availability under the Credit Facility. At March 31, 2011, we were in compliance with all provisions and covenants of the Credit Facility.

In November 2010, we entered into an amendment to our Credit Facility in connection with a debt exchange offer relating to \$115 million of our then outstanding 4.0% senior convertible notes and \$100 million of our then outstanding 4.25% senior convertible notes. Under the amendment, the lenders consented to the exchange offer and to the issuance of new 4.0% senior convertible and new 4.25% senior convertible notes, subject to, among other things, the condition that, prior to the expiration of the Credit Facility in May 2013, we may not use any borrowings under the Credit Facility, or more than \$10 million in the aggregate from cash on hand to defease, redeem or settle conversions of the new notes without prior consent of the lenders under the Credit Facility.

Based upon current availability under our Credit Facility, liquidity and cash flow projections, we believe we will be in compliance with the Credit Facility's terms and conditions and the minimum availability requirements for the next twelve months. We are dependent upon borrowings and letters of credit under this Credit Facility to fund operations. Should we be unable to comply with the terms and conditions of the Credit Facility, we would be required to obtain modifications to the Credit Facility or another source of financing to continue to operate. We may also not be able to achieve our projections; however, we do not expect this to adversely affect our ability to remain in compliance with the Credit Facility's minimum net availability requirements and minimum fixed charge coverage ratio for at least the next twelve months.

Senior Convertible Notes

New Senior Convertible Notes. During the first quarter of 2011, we exchanged \$105.3 million of our original 4.0% senior convertible notes and \$97 million of our original 4.25% senior convertible notes (together, the "Original Notes") for identical principal amounts of new 4.0% and 4.25% senior convertible notes (the "New Notes") for an exchange fee of approximately 50 basis points, or approximately \$1 million. Transaction costs of approximately \$0.1 million and \$0.5 million were recognized for the quarters ended March 31, 2011 and December 31, 2010, respectively, in connection with the exchange within selling, general and administrative expenses. The terms of the New Notes are substantially identical to those of the Original Notes, except that the New Notes have an optional physical, cash or combination settlement feature and contain certain conditional conversion features. Following the exchange offer, \$9.7 million and \$3.0 million of the original 4.0% and 4.25% notes, respectively, remain outstanding.

Because the New Notes have an optional cash settlement feature, and we intend to settle the principal amount of any conversions thereof in cash, the conversion shares underlying the principal amount of the New Notes are not included in our diluted share count. If, however, the average stock price per share exceeds the \$15.76 conversion price for the new 4.0% notes or the \$15.48 conversion price for the new 4.25% notes, then the resulting amount, in shares, of any premium will be included in our diluted share count ("premium shares"). Notwithstanding our present intention to settle conversions of our New Notes in cash, we cannot assure you that we will be able to do so due to restrictions in our credit facility, which prior to its expiration in May 2013, without the consent of the lenders thereunder, prohibits us from borrowing under the credit facility to settle any conversions of the New Notes and from paying more than \$10 million in the aggregate from cash on hand to settle conversions of the New Notes. If we were required to settle conversions of our new senior convertible notes in accordance with our stated intent to settle principal amounts due in cash, and we were unable to do so with existing cash balances or through our credit facility, we could be required to obtain additional funding. We cannot be certain that such funding would be available on terms acceptable to us, or at all.

Old Senior Convertible Notes. In November 2009, we issued \$100 million of 4.25% senior convertible notes due December 15, 2014 in a private placement. Of this amount, \$97 million was canceled and exchanged in the first quarter of 2011 for a like principal amount of new senior convertible notes in connection with our debt exchange as discussed above. The senior convertible notes bear interest at a rate of

4.25% per year, payable semi-annually in arrears, on June 15 and December 15 of each year, commencing on June 15, 2010. On or prior to December 12, 2014, holders may convert their notes into shares of our common stock at an initial conversion rate of 64.6162 shares of our common stock per \$1,000 principal amount of notes, which represents an initial conversion price of approximately \$15.48 per share, subject to customary anti-dilution adjustment terms for these types of notes. Approximately \$3.7 million in financing costs were incurred in connection with the issuance of these notes. These deferred financing costs are included in other assets in the consolidated balance sheet and are being amortized over the term of the notes.

In June 2009, we issued \$115 million of 4.0% senior convertible notes due June 15, 2014 in a registered offering. Of this amount, \$105.3 million was canceled and exchanged in the first quarter of 2011 for a like principal amount of new senior convertible notes in connection with our debt exchange as discussed above. The senior convertible notes bear interest at a rate of 4.0% per year, payable semi-annually in arrears, on June 15 and December 15 of each year, commencing on December 15, 2009. On or prior to June 13, 2014, holders may convert their notes into shares of our common stock at an initial conversion rate of 63.4417 shares of our common stock per \$1,000 principal amount of notes, which represents an initial conversion price of approximately \$15.76 per share, subject to customary anti-dilution adjustment terms for these types of notes. Approximately \$5.4 million in financing costs were incurred in connection with the issuance of these notes, which are included in other assets in the consolidated balance sheet and are being amortized over the term of the notes.

The 4.25% and 4.0% senior convertible notes are guaranteed by substantially all of our operating subsidiaries. There are no financial covenants on these notes, however, there are certain nonfinancial provisions and covenants associated with these notes. At March 31, 2011, we were in compliance with all such provisions and covenants.

Senior Notes

As of March 31, 2011, we had \$150 million of outstanding 7.625% senior notes due February 2017, with interest due semi-annually. The notes contain default (including cross-default) provisions and covenants restricting many of the same transactions as under our Credit Facility. The indenture that governs our 7.625% senior notes allows us to incur additional indebtedness to the extent that our fixed charge coverage ratio, as therein defined, is at least 2:1. The fixed charge coverage ratio is calculated as consolidated EBITDA for the most recent four fiscal quarters for which internal financial statements are available, divided by fixed charges for such four quarter period, as such terms are defined in the indenture. If the fixed charge coverage ratio is less than 2:1, we are still permitted to incur the following additional indebtedness, among others: credit facilities under a defined threshold, renewals to existing debt permitted under the indenture, capital lease obligations up to 5% of our consolidated net assets, plus an additional \$50 million of indebtedness at any time the senior notes remain outstanding. The senior notes are guaranteed by substantially all of our operating subsidiaries. At March 31, 2011, we were in compliance with all the provisions and covenants of the 7.625% senior notes.

Acquisition Debt

Equipment Term Loan. In connection with the acquisition of Pumpco, we entered into an equipment term loan in the aggregate amount of \$22.5 million at 7.05% interest, payable in 60 monthly installments, maturing in 2013. In October 2010, we repaid the \$13.4 million outstanding principal balance of the 7.05% equipment term loan with the proceeds of a new loan in the amount of \$13.5 million, bearing interest at 3.5267%. The 3.5267% equipment term loan is payable in 36 monthly installments, maturing in 2013 and is secured by most of Pumpco's existing equipment and guaranteed by MasTec.

Other Acquisition Related Notes Payable and Capital Lease Obligations. We assumed approximately \$34 million in notes payable for equipment and capital lease obligations in connection with the acquisition of Precision. In connection with the acquisition of Wanzek, we assumed approximately \$15 million of Wanzek's debt. We also assumed approximately \$9.5 million in notes payable for equipment and capital lease obligations in connection with the acquisition of Pumpco. As of March 31, 2011, \$16.0 million and \$5.1 million, respectively, remain outstanding on the Precision and Wanzek acquisition related debt described above. Pumpco's acquisition related debt of \$9.5 million has been substantially repaid as of March 31, 2011.

Except for one note with an immaterial principal balance at March 31, 2011, there are no financial covenants associated with the acquisition debt described above.

Subsequent Events. As discussed in Note 17 – Subsequent Events in the notes to the condensed unaudited consolidated financial statements, in April 2011, we acquired Fabcor Parent, a Canadian pipeline construction company. In connection with this acquisition, we assumed \$8.8 million of debt, which was repaid immediately. We also acquired EC Source, an extra high voltage specialty transmission contractor, in May 2011. In connection with the EC Source acquisition, we assumed \$8.6 million of debt.

Debt Guarantees

The subsidiary guarantees related to our senior convertible notes and senior notes are full and unconditional and joint and several, and there are generally no contractual restrictions on our ability to obtain funds from our subsidiaries. Also, MasTec, Inc. is a holding company with no independent assets or operations, and our subsidiaries that do not guarantee the senior convertible notes and senior notes are minor, individually, and in the aggregate, as such term is defined under the rules and regulations of the SEC. Accordingly, condensed consolidating financial information for MasTec, Inc. and its subsidiaries is not presented.

Auction Rate Securities

Our securities available for sale consist of auction-rate securities, which represent interests in pools of student loans guaranteed by the U.S. government under the Federal Family Education Loan Program and a structured finance security. As of March 31, 2011, we held \$22.5 million in par value of auction rate securities, with an estimated fair value and carrying value of \$19.0 million. Cumulative unrealized losses, net of unrealized gains, totaled \$0.8 million as of March 31, 2011, which are recorded in other comprehensive income, net of applicable income taxes.

Structured Finance Auction Rate Securities. As discussed in Note 7 – Securities Available for Sale in the notes to the condensed unaudited consolidated financial statements, following the sale of two of our three structured finance auction rate securities in the fourth quarter of 2010, we determined that we no longer met the criteria for intent to hold with respect to our remaining structured finance security. Accordingly, future declines in this security's carrying value will be recognized in earnings, whereas increases in its carrying value will be recorded in other comprehensive income, net of applicable income taxes. Cumulative gross unrealized gains in other comprehensive income associated with this security totaled \$0.4 million, or \$0.2 million net of applicable income taxes, as of March 31, 2011.

Student Loan Auction Rate Securities. As of March 31, 2011 our student loan auction rate securities have a par value and cost basis of \$17.5 million. We continue to have intent to hold, and more likely than not, will not be required to sell our student loan auction rate securities until maturity, or until their par value can be recovered. Cumulative gross unrealized losses on these securities, which are included as a component of other comprehensive income, total \$1.2 million, or \$0.7 million net of applicable income taxes, as of March 31, 2011.

In April 2011, the issuer of one of our student loan auction rate securities redeemed its security at the security's par value of \$4.6 million. As of March 31, 2011, the estimated fair value of this security was reclassified to current assets. In connection with the redemption, \$0.5 million of cumulative unrealized losses, along with the corresponding tax impact of \$0.2 million, were eliminated from other comprehensive income in April 2011.

See Note 7 - Securities Available for Sale in the notes to our condensed unaudited consolidated financial statements.

Off-Balance Sheet Arrangements

As is common in our industry, we have entered into certain off-balance sheet arrangements in the ordinary course of business. Our significant off-balance sheet transactions include liabilities associated with non-cancelable operating leases, letter of credit obligations, performance and payment bonds entered into in the normal course of business, self-insurance liabilities, liabilities associated with multi-employer pension plans and liabilities associated with certain indemnification and guarantee arrangements. We do not have any material off-balance sheet financing arrangements with variable interest entities.

Leases. In the ordinary course of business, we enter into non-cancelable operating leases for certain of our facility, vehicle and equipment needs, including related party leases. These leases allow us to conserve cash by paying a monthly rental fee for use of the related facilities, vehicles and equipment rather than purchasing them. The terms of these agreements vary from lease to lease, including some with renewal options and escalation clauses. We may decide to cancel or terminate a lease before the end of its term, in which case we are typically liable for the remaining lease payments under the term of the lease. Rent expense related to operating leases was approximately \$40.9 million for the three months ended March 31, 2011.

Letters of Credit. In the ordinary course of business, we are required to post letters of credit, primarily for our insurance carriers and surety bond providers. Such letters of credit are generally issued by a bank or similar financial institution. Letters of credit are also posted in support of performance under certain contracts. The letter of credit commits the issuer to pay specified amounts to the holder of the letter of credit if the holder claims that we have failed to perform specified actions. If this were to occur, we would be required to reimburse the issuer of the letter of credit. Depending on the circumstances of such a reimbursement, we may also have to record a charge to earnings for the reimbursement. As of March 31, 2011, we had \$97.5 million of standby letters of credit issued under our Credit Facility. We are not aware of any material claims relating to outstanding letters of credit as of March 31, 2011 and do not believe it is likely that any material claims will be made under any of our outstanding letters of credit in the foreseeable future. We may be required to post additional letters of credit or other collateral in favor of our insurance carriers, surety bond providers or customers in the future. Posting letters of credit reduces the borrowing availability under our Credit Facility.

Performance and Payment Bonds. In the ordinary course of business, we are required by certain customers to provide performance and payment bonds for some of our contractual commitments related to projects in process. These bonds provide a guarantee to the customer that we will perform under the terms of a contract and that we will pay subcontractors and vendors. If we fail to perform under a contract or to pay subcontractors and vendors, the customer may demand that the surety make payments or provide services under the bond. We must reimburse the

surety for any expenses or outlays it incurs. In August 2010, we increased our performance bonding capacity to accommodate growth in our businesses with bonding requirements. If our business continues to grow, our bonding requirements may increase in the future. If we are unable to successfully renew or obtain sufficient bonding capacity in the future, it could limit our ability to bid on certain contracts or perform work for certain customers. At March 31, 2011, the estimated cost to complete projects secured by our \$463.6 million in performance and payment bonds was \$79.4 million. We believe it is unlikely that we will have to fund significant claims under performance and payment bonds in the foreseeable future.

Included in our outstanding performance and payment bonds at March 31, 2011 was approximately \$3.0 million relating to the projects and assets of the state Department of Transportation business sold in 2007. In connection with the sale of these assets, we remain contingently liable for certain obligations, including liabilities associated with claims on these performance and payment bonds. The buyer of the Department of Transportation related projects and assets filed for bankruptcy protection in October 2009, which increases the likelihood that we will be required to assume certain obligations associated with these projects. As of March 31, 2011, the remaining cost to complete for the state Department of Transportation projects was immaterial.

Self-Insurance. We maintain insurance policies subject to per claim deductibles of \$1 million for our workers' compensation policy, \$2 million for our general liability policy and \$2 million for our automobile liability policy. We have excess umbrella coverage up to \$100 million per claim and in the aggregate. Liabilities under these insurance programs are accrued based upon our estimate of the ultimate liability for claims reported and an estimate of claims incurred but not reported, with assistance from third-party actuaries. These insurance liabilities are difficult to assess and estimate due to unknown factors, including the severity of an injury, the determination of our liability in proportion to other parties and the number of incidents not reported. Accruals are based upon known facts and historical trends and we believe such accruals to be adequate. As of March 31, 2011, our liability for unpaid claims and associated expenses, including incurred but not reported losses related to our workers compensation, general liability and automobile liability insurance policies, was \$38.7 million, of which \$25.4 million was reflected within non-current other liabilities.

We also maintain an insurance policy with respect to employee group health claims, which is subject to annual per employee maximum losses of \$0.4 million. Our liability for employee group claims as of March 31, 2011, which is based on statistical analysis of historical claims experience and specific knowledge of actual losses that have occurred, was \$1.3 million.

We are required to post letters of credit and provide cash collateral to our insurance carriers, or to provide surety bonds in certain states in which we are self-insured. As of March 31, 2011, these letters of credit amounted to \$59.4 million, a portion of which is collateralized by \$18.0 million of restricted cash at March 31, 2011. In addition, cash collateral deposited with insurance carriers, which is included in other long term assets in the condensed unaudited consolidated balance sheets, amounted to \$2.2 million as of March 31, 2011. Outstanding surety bonds related to self-insurance programs amounted to \$7.5 million as of March 31, 2011.

Employment Agreements. We have employment agreements with certain executives and other employees, which provide for compensation and certain other benefits and for severance payments under certain circumstances. Certain employment agreements also contain clauses that become effective upon a change of control of the Company. Upon the occurrence of any of the defined events in the various employment agreements, we will pay certain amounts to the relevant employees, which vary with the level of the employees' respective responsibility.

Collective Bargaining Agreements. Certain of our subsidiaries are party to various collective bargaining agreements with unions representing certain of their employees. The agreements require the subsidiaries party to the agreements to pay specified wages, provide certain benefits to their union employees and contribute certain amounts to multi-employer pension plans and employee benefit trusts. The multi-employer pension plan contribution rates are determined annually and assessed on a "pay-as-you-go" basis based on union employee payrolls. The required amount of future contributions cannot be determined for future periods because the number of union employees employed at any given time and the plans in which they may participate vary depending upon the location and number of ongoing projects at a given time and the need for union resources in connection with those projects. The collective bargaining agreements expire at various times and have typically been renegotiated and renewed on terms similar to the ones contained in the expiring agreements.

The Employee Retirement Income Security Act of 1974, as amended by the Multi-Employer Pension Plan Amendments Act of 1980, imposes certain liabilities upon employers who are contributors to a multi-employer plan in the event of the employer's withdrawal from, or upon termination of, such plan. In the event of a complete or partial withdrawal from any of these plans, or if the plans were to otherwise become underfunded, the subsidiaries (and potentially MasTec itself) could be assessed liabilities for the withdrawal or for additional contributions related to the underfunding of these plans. None of the subsidiaries party to the collective bargaining agreements have any current plans to withdraw from the plans. The Pension Protection Act of 2006 also added new funding rules generally applicable to plan years beginning after 2007 for multi-employer plans that are classified as "endangered," "seriously endangered," or "critical" status. For a plan in critical status, additional required contributions and benefit reductions may apply. We have been notified that certain plans to which a MasTec subsidiary contributes are in "critical" status and require additional contributions in the form of a surcharge on future benefit contributions required for future work performed by union employees covered by these plans. The amount of additional funds we may be obligated to contribute in the future cannot be estimated, as such amounts will be based on future levels of work that require the specific use of those union employees covered by these plans.

Indemnities. We generally indemnify our customers for the services we provide under our contracts, as well as other specified liabilities, which may subject us to indemnity claims, liabilities and related litigation. As of March 31, 2011, we were not aware of any material asserted or unasserted claims in connection with these indemnity obligations.

Other Guarantees. In the ordinary course of business, from time to time, we guarantee the obligations of our subsidiaries and certain investees, including obligations under certain contracts with customers, certain lease obligations and in some states, obligations in connection with obtaining contractors' licenses. We also generally warrant the work we perform for a one to two year period following substantial completion of a project. We have not historically accrued any reserves for potential warranty claims as they have been immaterial.

Impact of Inflation

The primary inflationary factors affecting our operations are labor and fuel costs, and to a lesser extent, material costs. Although we did experience increases in these costs in the first three months of 2011 as compared with 2010, the increases in labor and material costs were primarily due to revenue growth and changes in business mix. Fuel costs have also increased versus the same period in 2010 due to higher prices as well as to revenue growth and business mix. The price of fuel is subject to unexpected fluctuations from events outside of our control, including geopolitical events and fluctuations in global supply and demand. Significant fuel price increases could adversely impact our operating results in the future. We closely monitor inflationary factors and any impact they may have on our operating results or financial condition.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Interest Rate Risk

None of our outstanding debt at March 31, 2011 was subject to variable interest rates as we had no amounts outstanding on our Credit Facility as of March 31, 2011. Interest under the Credit Facility accrues interest at variable rates based, at our option, on the agent bank's base rate (as defined in the Credit Facility) plus a margin of between 1.25% and 1.75%, or at the LIBOR rate plus a margin of between 2.25% and 3.00%, depending on certain financial thresholds. At March 31, 2011, the margin over LIBOR was 2.25% and the margin over the base rate was 1.25%.

Our fixed interest rate debt primarily includes \$150 million in 7.625% senior notes, \$115 million in 4.0% senior convertible notes, \$100 million in 4.25% senior convertible notes and a \$13.5 million 3.5267% equipment term loan.

Foreign Currency Risk

As of March 31, 2011, we had limited operations in certain foreign countries, primarily Latin America and the Caribbean. Due to the limited nature of our foreign operations, fluctuations in foreign currency exchange rates have not had a material impact on our financial statements. In April 2011, we acquired Fabcor Parent, a Canadian pipeline construction company. Due to this recent acquisition and/or if we otherwise expand our operations outside of the United States, our exposure to fluctuations in foreign currency exchange rates could increase in the future. See Note 17 – Subsequent Events in the notes to the condensed unaudited consolidated financial statements regarding the acquisition of Fabcor Parent.

Auction Rate Securities

Our securities available for sale consist of auction rate securities. Our auction rate securities represent interests in pools of student loans guaranteed by the U.S. government under the Federal Family Education Loan Program and a structured finance security. The structured finance security has an attached credit default swap under which the principal value would be partially or fully forfeited at net default rates on the underlying corporate debt obligations ranging from 8% to 9% as of March 31, 2011. The actual net default rate as of March 31, 2011 is 4.89%. Both the structured finance security and the credit default swap are collateralized by investment grade credit-linked notes made up of floating rate international bank notes or credit card receivable notes. As of March 31, 2011, we hold \$22.5 million in par value of auction rate securities, with an estimated fair value and carrying value of \$19.0 million. Cumulative unrealized losses, net of gains, total \$0.8 million as of March 31, 2011, which are recorded in other comprehensive income, net of applicable income taxes.

In April 2011, the issuer of one of our student loan auction rate securities redeemed its security at the security's par value of \$4.6 million. As of March, 31, 2011, the estimated fair value of this security was reclassified to current assets. In connection with the redemption, \$0.5 million of cumulative unrealized losses, along with the corresponding tax impact of \$0.2 million, were eliminated from other comprehensive income in April 2011.

See Note 7 - Securities Available for Sale in the notes to condensed unaudited consolidated financial statements.

ITEM 4. CONTROLS AND PROCEDURES

As of the end of the period covered by this report, we carried out an evaluation, under the supervision and with the participation of our management, including our Chief Executive Officer and our Chief Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures, as defined in Rule 13a-15(e) under the Securities Exchange Act of 1934, as amended ("the Exchange Act"). Based upon that evaluation, we concluded that as of March 31, 2011, our disclosure controls and procedures are effective to ensure that information required to be disclosed in reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the Securities and Exchange Commission and that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, to allow for timely decisions regarding required disclosures.

Changes in Internal Control over Financial Reporting

There were no changes in our internal control over financial reporting identified in connection with the evaluation required by paragraph (d) of Exchange Act Rules 13a-15 or 15d-15 that occurred during the period covered by this report that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

Refer to Note 14 – Commitments and Contingencies in the notes to our condensed unaudited consolidated financial statements included in this Quarterly Report on Form 10-Q for a discussion of any recent material developments related to our legal proceedings since the filing of our most recent Annual Report on Form 10-K.

ITEM 1A. RISK FACTORS

Except as set forth below, there have been no material changes to any of the risk factors disclosed in our recently filed Annual Report on Form 10-K.

We derive a significant portion of our revenue from a few customers, and the loss of one of these customers or a reduction in their demand for our services could impair our financial performance.

For the three months ended March 31, 2011, we derived approximately 27%, 23% and 20% of our revenue from AT&T, DIRECTV *, and El Paso Corporation, respectively. For the three months ended March 31, 2010, we derived approximately 27%, 19% and 15% of our revenue from DIRECTV *, Enbridge and AT&T, respectively. In addition, our ten largest customers accounted for approximately 82% and 83% of our revenue in the three months ended March 31, 2011 and 2010, respectively. Because our business is concentrated among relatively few major customers, and certain of our services are provided on a non-recurring, project by project basis, we could experience a reduction in our results of operations, cash flows and liquidity if the amount of business we obtain from these customers is reduced, or if we complete the required work on our projects and cannot replace them with similar projects. We have recently experienced an increase in the proportion of our revenues from non-recurring project specific work, which may further increase this risk if we are not able to replace completed project work with new work. In addition, many of the contracts with our largest customers may be canceled on short or no advance notice. Any of these factors could negatively impact our results of operations, cash flows and liquidity.

Additionally, pursuant to our February 2011 amended and restated purchase option agreement with Red Ventures LLC ("Red Ventures"), Red Ventures has an option to purchase DirectStar TV LLC ("DirectStar") and its subsidiaries (which, in support of the DIRECTV ® installation business, provides marketing and sales services on behalf of DIRECTV®) from MasTec at any time from March 1, 2012 to November 30, 2012, for an amount equal to the sum of: (i) the shareholders' equity of DirectStar as of May 31, 2010, (ii) five percent (5%) of adjusted net income (generally, the net income (loss) before provision for income taxes) of DirectStar from January 1, 2010 until the last day of the month immediately prior to the date of the sale and (iii) \$25,600,000. Should Red Ventures exercise its purchase option, our revenues and profits from DIRECTV® would be reduced.

We have a significant amount of debt. Our substantial indebtedness could adversely affect our business, financial condition and results of operations and our ability to meet our payment obligations.

We have a significant amount of debt and substantial debt service requirements. As of March 31, 2011, we had approximately \$413.6 million of outstanding debt.

This level of debt could have significant consequences on our future operations, including:

- making it more difficult for us to meet our payment and other obligations;
- our failing to comply with the financial and other restrictive covenants contained in our debt agreements, which could trigger an event of default that results in all of our debt becoming immediately due and payable;
- reducing the availability of our cash flow to fund working capital, capital expenditures, acquisitions or strategic investments and other general corporate requirements, and limiting our ability to obtain additional financing for these purposes;
- subjecting us to the risk of increased sensitivity to interest rate increases on our indebtedness with variable interest rates, including borrowings under our credit facility;
- preventing us from paying dividends;
- limiting our flexibility in planning for, or reacting to, and increasing our vulnerability to changes in our business, the industry in which we
 operate and the general economy; and
- placing us at a competitive disadvantage compared to our competitors that have less debt or are less leveraged.

Any of the above-listed factors could have an adverse effect on our business, financial condition and results of operations and our ability to meet our payment obligations.

Our ability to meet our payment and other obligations under our debt instruments depends on our ability to generate significant cash flow in the future. This, to some extent, is subject to general economic, financial, competitive, legislative and regulatory factors as well as other factors that are beyond our control. We cannot assure you that our business will generate cash flow from operations, or that future borrowings will be available to us under our credit facility or otherwise in an amount sufficient to enable us to meet our payment obligations and to fund other liquidity needs. If we are not able to generate sufficient cash flow to service our debt obligations, we may need to refinance or restructure our debt, sell assets, reduce or delay capital investments, or seek to raise additional capital, and some of these activities may be on terms that are unfavorable or highly dilutive. Our ability to refinance our indebtedness will depend on the capital markets and our financial condition at such time. If we are unable to implement one or more of these alternatives, we may not be able to meet our payment obligations.

As of March 31, 2011, the closing price of our common stock of \$20.80, as reported by the New York Stock Exchange, exceeded the \$15.76 and \$15.48 conversion prices of our outstanding 4.0% and 4.25% senior convertible notes, respectively. Currently, we intend to settle the principal amounts of our new senior convertible notes upon any conversion thereof in cash. As of March 31, 2011, there was outstanding \$202.3 million aggregate principal amount of our new senior convertible notes. Notwithstanding our present intention to settle conversions of our new senior convertible notes in cash, we cannot assure you that we will be able to do so due to restrictions in our credit facility, which prior to its expiration in May 2013, without the consent of the lenders thereunder, prohibits us from borrowing under the credit facility to settle any conversions of the new senior convertible notes and from paying more than \$10 million in the aggregate from cash on hand to settle conversions of the new senior convertible notes. If we were required to settle conversions of our new senior convertible notes in accordance with our stated intent to settle principal amounts due in cash, and we were unable to do so with existing cash balances or through our credit facility, we could be required to obtain additional funding. We cannot be certain that such funding would be available on terms acceptable to us, or at all.

There may be future sales or other dilution of our equity, which may adversely affect the market price of our common stock. In connection with certain completed acquisitions and financing transactions, we have issued shares of our common stock or securities that are convertible into shares of our common stock, and in addition, we have the option to issue shares of our common stock instead of cash as consideration for future earn-out obligations. We may agree to issue additional securities in connection with other future acquisition or financing transactions; which, if issued, would dilute your share ownership and could lead to volatility in our common stock price.

We grow our business organically as well as through acquisitions. One method of acquiring companies or otherwise funding our corporate activities is through the issuance of additional equity securities. In connection with certain completed acquisitions, we have the option to issue shares of our common stock instead of cash for certain earn-out obligations. Such issuances could have the effect of diluting our earnings per share as well as our existing shareholders' individual ownership percentages and could lead to volatility in our common stock price. Our Amended and Restated Articles of Incorporation provide that we may issue up to a total 145,000,000 shares of common stock, of which 79,219,910 shares were outstanding as of March 31, 2011. Additionally, in November 2010, we acquired 33% of the outstanding equity interests in EC Source Services LLC, ("EC Source"), together with an option to acquire the balance of EC Source's outstanding equity. On April 29, 2011, we exercised our option and on May 2, 2011, we consummated the acquisition of EC Source. In partial consideration for this acquisition, we issued 5,129,644 shares of our common stock, which are subject to certain transfer restrictions.

We are not restricted from issuing additional common stock. The issuance of additional shares of our common stock upon conversion of any of our \$215 million aggregate principal amount of convertible notes, in connection with future acquisitions or other issuances of our common stock or convertible securities, including outstanding options and warrants, or otherwise, will dilute the ownership interest of our common shareholders.

Sales of a substantial number of shares of our common stock or other equity-related securities in the public market could depress the market price of our common stock and impair our ability to raise capital through the sale of additional equity or equity-linked securities. We cannot predict the effect that future sales of our common stock or other equity-related securities would have on the market price of our common stock.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

The information contained in Item 5 of this Quarterly Report on Form 10-Q is incorporated by reference in this Item 2.

The following table provides information about repurchases of our common stock during the three month period ended March 31, 2011:

<u>Period</u>	Total Number Average of Shares Purchased (1)	Average Price Paid per Share	Total Number of Shares Purchased as Part of Publicly Announced Program	Approximate Dollar Value of Shares that May Yet be Purchased under the Program
January 1, 2011 through January 31, 2011	_	_	_	_
February 1, 2011 through February 28, 2011	_		_	_
March 1, 2011 through March 31, 2011	5,785	\$20.45	_	_

 Represents shares of common stock withheld for income tax purposes in connection with shares of restricted stock issued to certain employees and directors.

ITEM 5. OTHER INFORMATION

As previously reported, on November 16, 2010, we entered into a membership interest purchase agreement pursuant to which, for \$10 million, we purchased a 33% interest in EC Source, which is involved in the engineering, design, construction and maintenance of electric transmission and substation projects. Additionally, we acquired the EC Source Merger Option granting us the right to acquire the balance of the equity in EC Source pursuant to the terms of an Agreement and Plan of Merger entered into on November 16, 2010, by and among MasTec, EC Source, certain wholly owned subsidiaries of MasTec and the members of EC Source (the "Merger Agreement").

On April 29, 2011, we exercised the EC Source Merger Option, and, on May 2, 2011 (the "Closing Date"), we consummated the acquisition of EC Source. Pursuant to the Merger Agreement, on the Closing Date, we issued to the other EC Source equity holders (the "Sellers") an aggregate of 5,129,644 shares (the "Shares") of our common stock, par value \$0.10 ("Common Stock"). Additionally, the Sellers are entitled to a five-year earn-out (the "Earn-out"), payable at our election in Common Stock, cash or a combination thereof. The dollar amount of the Earn-out will be equal to 20% of the excess, if any, of EC Source's annual EBITDA over \$15 million. The Shares are subject to transfer restrictions which will lapse 25% on the first and second anniversaries of the Closing Date and 50% on the third anniversary of the Closing Date.

We issued the Shares in reliance upon the exemption from registration provided by Section 4(2) of the Securities Act of 1933, as amended (the "Securities Act"). Each Seller represented to us that such Seller was an "accredited investor," as defined in Rule 501(a) promulgated under the Securities Act, and that such Seller was acquiring the Shares for investment and not with a view to the distribution thereof.

The foregoing description of the Merger Agreement is only a summary and is qualified in its entirety by reference to the complete text of the Merger Agreement, which is filed as Exhibit 10.44 to this Quarterly Report on Form 10-Q and incorporated by reference in this Item 5.

ITEM 6. EXHIBITS

Exhibit No.	<u>Description</u>
10.44*	Agreement and Plan of Merger, dated as of November 16, 2010, by and among MasTec, Inc., EC Source Services, LLC and the other parties thereto.
23.1*	Consent of Independent Valuation Firm.
31.1*	Certification of Chief Executive Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2*	Certification of Chief Financial Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1*	Certification of Chief Executive Officer, pursuant to 18 U.S.C. Section 1350, as Adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2*	Certification of Chief Financial Officer, pursuant to 18 U.S.C. Section 1350, as Adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

^{*} Filed herewith.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

MASTEC, INC.

Date: May 4, 2011

/s/ Jose R. Mas

Jose R. Mas Chief Executive Officer (Principal Executive Officer)

/s/ C. Robert Campbell

C. Robert Campbell Chief Financial Officer (Principal Financial and Accounting Officer)

Exhibit Index

Exhibit No.	<u>Description</u>
10.44	Agreement and Plan of Merger, dated as of November 16, 2010, by and among MasTec, Inc., EC Source Services, LLC and the other parties thereto.
23.1	Consent of Independent Valuation Firm.
31.1	Certification of Chief Executive Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of Chief Financial Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certification of Chief Executive Officer, pursuant to 18 U.S.C. Section 1350, as Adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2	Certification of Chief Financial Officer, pursuant to 18 U.S.C. Section 1350, as Adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

AGREEMENT AND PLAN OF MERGER

Dated November 16, 2010

by and among

MasTec, Inc.,

ECS Acquisition Merger Subsidiary I, Inc.,

ECS Acquisition Merger Subsidiary II, LLC,

EC Source Services, LLC,

Force Capital Partners, LLC,

FCP Investments, LLC,

Casey Maslonka,

Martin Maslonka,

Jon Maslonka,

Sidney Strauss

and

Justin Campbell

TABLE OF CONTENTS

1.	1. THE MERGER 2					
	1.1	The Merger	2			
	1.2	Effective Time	2			
	1.3	Closing	2			
	1.4	Subsequent Merger	3			
	1.5	Governing Documents	4			
	1.6	Manager; Board of Directors	4			
	1.7	Management	4			
	1.8	Effect on Units	4			
	1.9	Closing Date Merger Consideration	4			
		Estimate of Net Working Capital, Tangible Net Worth; PP&E	8			
		Post-Closing Purchase Price Adjustment	9			
		Transactions at Closing; Exchange of Certificates; Exchange Agent	10			
		Withholding	11			
		Further Action	11			
	1.15	Options	11			
2.	REPF	RESENTATIONS AND WARRANTIES OF ECS AND THE MEMBERS	12			
	2.1	Organization; Predecessors; Subsidiaries	12			
	2.2	Capitalization of the Companies	12			
	2.3	Organization, Power and Authorization	13			
	2.4	Authorization of Governmental Authorities	13			
	2.5	Noncontravention	14			
		Financial Statements	14			
	2.7	Absence of Undisclosed Liabilities	15			
	2.8	Absence of Certain Developments	15			
	2.9	Assets	16			
		Accounts Receivable	17			
		Backlog	17			
		Real Property	17			
		Tangible Personal Property	18			
		Intellectual Property	18			
		Legal Compliance; Illegal Payments; Permits	20			
		Inventories Final and Deposit Plans	21			
		Employee Benefit Plans Environmental Matters	21 23			
		Contracts; Work-in-Process	25			
		Affiliate Transactions	27			
		Customers and Suppliers	27			
		Customer Warranties	27			
		Product Liability	27			
		Capital Expenditures and Investments	27			
		Employees	28			
		Litigation; Governmental Orders	29			
		Insurance	29			
	2.28	Banking Facilities	29			
	2.29	Powers of Attorney	29			
	2.30	No Brokers	29			
	2.31	Surety Bonds	29			
		Reorganizations for Federal Income Tax Purposes	30			
		No Material Adverse Effect	30			
	2.34	Securities Law Matters	30			
3.	REPF	RESENTATIONS AND WARRANTIES OF PARENT	31			
	3.1	Organization	31			
	3.2	Power and Authorization	31			

	3.3	Authorization of Governmental Authorities	31
	3.4	Noncontravention Noncontravention	32
	3.5	No Brokers	32
	3.6 3.7	Securities Parent Financial Statements	32 32
4.	PRE-	CLOSING COVENANTS	32
	4.1	Access and Investigation	32
	4.2	Operation of the Companies and their Businesses	32
	4.3	Commercially Reasonable Efforts; Notification; Hart-Scott-Rodino Act Filings	36
	4.4	Acquisition Proposals	36
	4.5	Bank Accounts; Powers of Attorney	37
	4.6	Notice Complete in the control of th	37
	4.7	Consultation	37
	4.8 4.9	Interim Financial Statements Environmental Review	37 37
		Employee Benefit Plans	38
		Employment Agreements	38
		Health and Life Insurance Plans	38
_			
٥.	POST	C-CLOSING COVENANTS	39
		Confidentiality	39
	5.2	Restrictive Covenants	39
	5.3	Cooperation Regarding Financial Statements	40
		Litigation Support	40
		Restrictions on Transfer of Parent Shares	41
	5.6	Member Suretyship Obligations	41
6.	CON	DITIONS PRECEDENT TO PARENT'S OBLIGATION TO CLOSE	41
		Accuracy of Representations and Warranties	41
		ECS's and Members' Performance	41
	6.3	No Actions	42
	6.4	No Claim Regarding Units or Merger Consideration	42
		No Prohibition	42
	6.6	No Material Adverse Effect	42
		HSR Act Third Ports Concerts	42
		Third-Party Consents Merger Option	42 42
		Employment Agreements	42
		Non-Competition Agreements	42
		Environmental Review	42
		Member Affiliate Leases	42
7		DITIONS PRECEDENT TO ECS'S AND MEMBERS' OBLIGATIONS TO CLOSE	42
٠.			
	7.1	Accuracy of Representations and Warranties	42
	7.2	Parent's Performance No Prohibition	43
	7.3 7.4	HSR Act	43
_			43
8.		MINATION	43
	8.1	Termination Events	43
	8.2	Effect of Termination	43

9.	INDEN	MNIFICATION	43
	9.1	Indemnification by the Members	43
	9.2	Indemnity by Parent	44
	9.3	Time for Claims	44
	9.4	Indemnification Limits	45
	9.5	Third Party Claims	45
	9.6	Limitations on Liability	46
	9.7	Exclusive Remedy	46
	9.8 9.9	Members' Representative	47 47
	9.10	Knowledge and Investigation Remedies Cumulative	47
	9.11	Right of Setoff	47
	9.12	ECS's Indemnification Obligations	47
	9.13	Time of Payment of Claims	47
	9.14	Closing Balance Sheet	47
10.	TAX N	MATTERS	47
	10.1	Representations and Obligations Regarding Taxes	47
	10.2	Indemnification for Taxes	49
	10.3	Covenants With Respect To Taxes	49
11.	MISCI	ELLANEOUS	50
	11.1	Notices	50
	11.2	Members' Representative	52
	11.3	Publicity	53
	11.4	Succession and Assignment; No Third-Party Beneficiary	53
	11.5	Amendments and Waivers	53
	11.6	Further Assurances	53
	11.7	Entire Agreement Disalogue Schodulary Listed Dogwests etc.	53
	11.8 11.9	Disclosure Schedules; Listed Documents, etc Counterparts; Execution	54 54
		Survival	54
		Severability	54
		Headings	54
		Construction	54
		Governing Law	54
		Jurisdiction; Venue; Service of Process	54
		Waiver of Jury Trial	5 5
		Expenses	5 5
		Specific Performance	5 5
		Joint and Several	5 5
		Recitals	5 5
		Representation by Lewis and Roca LLP	5 5
12.		NITIONS; CERTAIN RULES OF CONSTRUCTION	5 6
	12.1	Definitions	56
	12.2	Glossary of Other Defined Terms	62
	12.3	Rules of Construction	65
	iibits:		
	ibit A	Articles of Merger	
	ibit B	General Release	
	iibit C iibit D	Certificate of ECS Certificate of Parent	
ĽXÍ.	ոսու	Continuate of a archi	

Exhibit E Martin Maslonka Employment Agreement

Exhibit F Letter of Transmittal
Exhibit G InfraSource and Option Holder Indemnification
Exhibit H Software Booked As PP&E
Exhibit I Martin Maslonka Marriage Certificate
Exhibit J Form of Employment Agreement Amendments

SCHEDULES

1.9.1(b)	Per Unit Closing Date Merger Consideration
2.1.1	Jurisdictions in which the Companies are qualified to transact business
2.1.2	Predecessor Status, etc.
2.2.1	Capitalization
2.2.2	Indirect Members Capitalization
2.2.5	Encumbrances, etc.
2.2.6	Debt
2.4	Subsequent Merger Filings
2.5(a)	Noncontravention under Company Agreements
2.5(b)	Noncontravention under Merger Documents and Option Agreement
2.6.1	Financial Statements
2.6.2(a)	Noncompliance with GAAP
2.6.2(b)	Internal Accounting Controls
2.7(a) and 2.7(b)	Undisclosed Liabilities
2.8(a) and 2.8(b)	Certain Developments
2.9.1	Ownership of Assets
2.9.2	Sufficiency of Assets
2.10	Accounts Receivable
2.11	Backlog
2.12	Real Property
2.13	Tangible Personal Property
2.14	Intellectual Property
2.15.1	Compliance
2.15.3	Permits
2.15.4	Additional Compliance Representations
2.16	Inventories
2.17(a)	Employee Benefit Plans
	± •
2.17(b)	Employee Benefit Plans - Documents Provided
2.17(d)	Employee Pension Benefit Plan
2.17(e)	Employee Benefit Plans - Actions, Suits or Claims
2.17(f)	Employee Benefit Plans - Compensation
2.17(g)	Employee Benefit Claims - Changes
2.17(i)	Certain Matters with Respect to Employee Benefit Plans
2.18	Environmental Matters
2.19	Contracts
2.19.3	Breach, etc.
2.19.4	Bids
2.19.5	Work-in-Process
2.20	Affiliate Transactions
2.21(a) and 2.21(b)	Customers and Suppliers
2.22	Customer Warranty Claims
2.23	Product Liability
2.24	Capital Expenditures and Investments
2.25	Employees
2.26	Litigation; Governmental Orders
2.27	Insurance
2.28	Banking Facilities
	<u>e</u>
2.29	Powers of Attorney
2.30(a) and 2.30(b)	No Brokers
2.31.1	Surety Bonds
2.31.2	Compliance with Surety Bonds
2.33	Material Adverse Effects
3.4	Noncontravention
10	Taxes
10.1.1	Tax Returns
10.1.2	Tax Clearances
10.1.3	Tax Audits and Disputes
10.1.4	Ventures Taxed as Partnerships
	÷

AGREEMENT AND PLAN OF MERGER

This Agreement and Plan of Merger, dated November 16, 2010, as amended or otherwise modified (the "<u>Agreement</u>"), by and among MasTec, Inc., a Florida corporation (the "<u>Parent</u>"), ECS Acquisition Merger Subsidiary I, Inc., a Florida corporation and wholly-owned subsidiary of Parent ("<u>Merger Subsidiary</u>"), ECS Acquisition Merger Subsidiary II, LLC, a Florida limited liability company and wholly-owned subsidiary of Parent ("<u>Sister Subsidiary</u>"), EC Source Services, LLC, a Nevada limited liability company ("<u>ECS</u>"), Force Capital Partners, LLC, a Nevada limited liability company ("<u>FCP Investments</u>", and together with Force Capital, the "<u>Entity Members</u>"), Jon Maslonka, a resident of the State of Wyoming ("<u>Jon</u>"), Sidney Strauss, a resident of the State of Arizona ("<u>Strauss</u>"), Justin Campbell, a resident of the State of Wyoming ("<u>Campbell</u>," Campbell together with Jon and Strauss, the "<u>Non-Entity Members</u>" and together with the Entity Members, the "<u>ECS Members</u>", and Casey Maslonka, a resident of the State of Arizona and Martin Maslonka, a resident of the State of Nevada (together, the "<u>Indirect Members</u>" and together with the ECS Members, the "<u>Members</u>").

RECITALS

WHEREAS, simultaneous with the execution and delivery of this Agreement, Parent, ECS and the Members are entering into a Membership Interest Purchase Agreement (the "Membership Interest Purchase Agreement"), pursuant to which Parent is purchasing Preferred Units of ECS representing a thirty-three percent (33%) interest in ECS on a fully diluted basis and all of the Preferred Units of ECS (the "Preferred Units");

WHEREAS, the ECS Members own all of the outstanding Common Units of ECS (the "Common Units"), which together with the Preferred Units are all of the outstanding LLC Units of ECS (the "Units");

WHEREAS, simultaneous with the execution and delivery of this Agreement and as a condition to Parent entering into the Membership Interest Purchase Agreement, Parent, ECS and the Members are entering into a Merger Option Agreement (the "Option Agreement"), pursuant to which Parent shall have the right, but not the obligation, to exercise an option (the "Merger Option") and cause the provisions of this Agreement which are contingent upon Parent's exercise of the Merger Option to become operative;

WHEREAS, until Parent exercises the Merger Option, none of Parent, Merger Subsidiary or Sister Subsidiary shall have any obligation or Liability pursuant to this Agreement;

WHEREAS, from the date hereof, as set forth herein ECS and the Members have certain obligations and Liabilities pursuant to this Agreement in order to facilitate Parent's ability to exercise the Merger Option;

WHEREAS, the Indirect Members are all of the members of Force Capital;

WHEREAS, ECS is the sole shareholder of Energy Environmental Group, Inc., a Nevada corporation ("<u>EEG</u>"), T&D Power, Inc., a Nevada corporation ("<u>T&D</u>"), and AT Power, Inc., a Nevada corporation ("<u>ATP</u>"), and the sole member of EC Source Aviation, LLC, a Nevada limited liability company ("<u>ECS Aviation</u>", and collectively with EEG, T&D, and ATP, the "<u>Subsidiaries</u>", and together with ECS, the "<u>Companies</u>", and each individually, a "<u>Company</u>");

WHEREAS, as a result of their direct or indirect ownership interests in ECS, each of the Members has benefitted from Parent's purchase of the Preferred Units and Merger Option pursuant to and for the consideration set forth in the Membership Interest Purchase Agreement and will benefit from the Contemplated Transactions, and in consideration therefor, is entering into this Agreement and shall carry out such Member's obligations hereunder;

WHEREAS, Parent, Merger Subsidiary, Sister Subsidiary, ECS, and the Members are hereby agreeing that should Parent exercise the Merger Option, pursuant to this Agreement they will enter into a business combination transaction pursuant to which Merger Subsidiary will merge with and into ECS, with ECS continuing as the surviving entity (the "Merger"), and immediately thereafter, ECS will merge with and into Sister Subsidiary with Sister Subsidiary continuing as the surviving entity (the "Subsequent Merger"), all upon the terms and subject to the conditions set forth in this Agreement and in accordance with the Florida Business Corporation Act (the "FBCA"), the Nevada Revised Statutes (the "NRS"), and the Florida Limited Liability Company Act (the "FLLCA");

WHEREAS, Force Capital, as Manager of ECS has approved this Agreement and the Contemplated Transactions, including the Merger and the Subsequent Merger, in each case, subject to the terms and conditions hereof;

WHEREAS, Force Capital, as Manager of ECS has recommended to the ECS Members, the approval and adoption of this Agreement, the Merger and consummation of the Contemplated Transactions, subject to the terms and conditions hereof;

WHEREAS, the ECS Members have unanimously approved this Agreement and the Contemplated Transactions;

WHEREAS, on or prior to the exercise of the Merger Option, (i) the respective Boards of Directors of each of Parent and Merger Subsidiary and Parent as the Managing Member of Sister Subsidiary will have approved this Agreement and the Contemplated Transactions, including the Merger and the Subsequent Merger, (ii) Parent as Merger Subsidiary's sole shareholder, will have approved this Agreement and the Contemplated Transactions, and (iii) the Board of Directors of Parent will have approved the issuance of the Parent Shares in connection with the Merger, in each case of (i)-(iii), subject to the terms and conditions hereof:

WHEREAS, for U.S. federal income tax purposes, it is intended that the Merger and the Subsequent Merger, taken together, qualify as a "reorganization" under the provisions of Section 368 of the Code, and the Treasury Regulations promulgated and the rulings issued thereunder;

WHEREAS, simultaneous with the execution and delivery of this Agreement and as a condition to Parent entering into the Membership Interest Purchase Agreement, the Merger Option and this Agreement, Martin Maslonka is duly executing and delivering to Parent the Martin Maslonka Employment Agreement attached hereto as Exhibit E with an effective date as of the Closing Date; and

WHEREAS, Parent, Merger Subsidiary, Sister Subsidiary, ECS, and the Members desire to make certain representations, warranties, covenants and agreements in connection with the Merger and to prescribe certain conditions to the Merger.

AGREEMENT

NOW THEREFORE, in consideration of the premises and mutual promises herein made, and in consideration of the representations, warranties, covenants and agreements herein contained, Parent, Merger Subsidiary, Sister Subsidiary, ECS, and the Members hereby agree as follows:

1. THE MERGER

- 1.1 The Merger. Upon the terms and subject to the conditions set forth in this Agreement, at Closing the parties hereto shall cause the Merger to be consummated by ECS and Merger Subsidiary executing and delivering Articles of Merger (the "Articles of Merger"), substantially in the form attached hereto as Exhibit A, which, together with this Agreement or an appropriate Plan of Merger, shall be filed at Closing with the Secretary of State of the State of Florida in accordance with the FBCA, and the Secretary of State of the State of Nevada, along with all other required documents, in accordance with the NRS. Subject to the terms and conditions of this Agreement, at the Effective Time, Merger Subsidiary shall be merged with and into ECS in accordance with, and with the effects provided in, the applicable provisions of the FBCA and the NRS. ECS shall be the Surviving Entity (sometimes hereinafter referred to as the "Surviving Entity") resulting from the Merger and, as a result, shall become a wholly owned subsidiary of Parent, shall continue to be governed by the laws of the State of Nevada and shall succeed to and assume all of the rights and obligations of Merger Subsidiary, and the separate corporate existence of Merger Subsidiary shall cease.
- 1.2 Effective Time. The Merger shall become effective upon (i) the Articles of Merger having been accepted for filing by the Secretary of State of the State of Nevada or (ii) at such later time as is set forth in the Articles of Merger (the "Effective Time"). The Merger shall have the effects set forth in Section 92A.250 of the NRS and Section 607.1106 of the FBCA.

1.3 Closing.

1.3.1 Time and Place of Closing. The Closing of the Contemplated Transactions (the "Closing") shall take place at the earlier of (i) a time and on a date mutually agreed upon by the parties hereto and (ii) 1:00 p.m., Coral Gables, Florida time, on the third Business Day following the date on which all of the conditions to the obligations of the parties set forth in Sections 6 and 7 (other than the conditions to be satisfied at Closing) have been satisfied or waived as provided therein (the "Closing Date"). The Closing shall take place at the offices of Greenberg Traurig, P.A, located at 333 Avenue of the Americas, Miami, Florida 33131, unless another time, date or place is agreed to in writing by the parties hereto. Subject to the provisions of Section 8, failure to consummate the Merger on the date and time and at the place determined pursuant to this Section 1.3.1 shall not result in the termination of this Agreement or relieve any party of any obligation under this Agreement. Notwithstanding the time and date upon which the Closing actually occurs, for purposes of determining a change in control in accordance with GAAP, if the Closing occurs on any day of a calendar month other than the first, then the Closing shall be deemed to have occurred and become effective as of 12:01 a.m. Coral Gables, FL time on the first day of such calendar month. Notwithstanding anything to the contrary set forth herein and for purposes of clarity, the Closing Balance Sheet shall be prepared and determined as of the date upon which Closing actually occurs, without regard to the previous sentence. All deliveries, payments and other transactions and documents relating to the Closing shall be interdependent and none shall be effective unless and until all are effective (except for any of same as to which the party entitled to the benefit thereof has expressly waived satisfaction or performance thereof as a condition precedent to the Closing).

1.3.2 Closing Deliveries.

- (a) At Closing ECS and the Members will deliver, or cause to be delivered, to Parent:
 - (i) the Certificates;
 - (ii) Articles of Merger duly executed by ECS;
- (iii) written consents duly executed by all of the ECS Members, Force Capital as the manager of ECS, and the manager and members of the Entity Members authorizing their execution and delivery of this Agreement and approval of the Contemplated Transactions;
- (iv) an external hard drive (which shall be permanent and accessible, without the need for any password, with readily and commercially available software) containing, in electronic format, all documents posted to the datasite maintained by Lewis and Roca LLP on behalf of ECS as of Closing (the "Data Room");
- (v) a general release (a "General Release") substantially in the form of Exhibit B duly executed by each Member in favor of the Companies and Parent;
- (vi) certificates of good standing of each Company and Entity Member in its jurisdiction of organization and with respect to each Company each of the foreign jurisdictions in which it is qualified;
 - (vii) certified copies of the charter documents of each Company and Entity Member;
- (viii) certificates as to the incumbency of ECS's and the Entity Members' officers executing this Agreement and any of the Company Agreements;
 - (ix) the Letters of Transmittal;
- (x) the Martin Maslonka Marriage Certificate substantially in the form of Exhibit I duly executed by Martin Masloka and his wife and
- (xi) resignations of those directors and officers of the Companies (solely with respect to such offices and positions, and not with respect to employment) as requested by Parent.
- (b) At Closing Parent will deliver, or cause to be delivered:
 - (i) the Per Unit Closing Date Merger Consideration to the Exchange Agent and the Option Consideration to the Option Holders;
 - (ii) Articles of Merger duly executed by Merger Subsidiary;
 - (iii) Reasonably satisfactory evidence of Parent's due authorization of the Contemplated Transactions;
- (iv) written consent of the sole shareholder and board of directors of Merger Subsidiary and the sole member of Sister Subsidiary authorizing the execution and delivery of this Agreement and the consummation of the Contemplated Transactions; and
- (v) a certificate of active status of each of Parent, Merger Subsidiary and Sister Subsidiary from the Secretary of State of the State of Florida.

1.4 Subsequent Merger.

1.4.1 Immediately following the Effective Time and in accordance with the FLLCA and the NRS, Parent will cause the Surviving Entity to merge with and into Sister Subsidiary and the separate limited liability company existence of the Surviving Entity shall thereupon cease. Sister Subsidiary shall be the surviving limited liability company in the Subsequent Merger, and shall succeed to and assume all the rights and obligations of the Surviving Entity in accordance with Section 92A.250 of the NRS and Section 608.4383 of the FLLCA.

- 1.4.2 At the effective time of the Subsequent Merger and without any further action on the part of the Surviving Entity, Parent, or Sister Subsidiary, all of the membership interest of the Surviving Entity immediately prior to the effective time of the Subsequent Merger shall automatically be retired and cease to be outstanding and Parent shall continue to hold 100% of the membership interest of Sister Subsidiary.
 - 1.4.3 After the Subsequent Merger, references herein to the Surviving Entity shall refer to Sister Subsidiary.
- 1.4.4 This Agreement is intended to constitute a "plan of reorganization" with respect to the Merger and Subsequent Merger, taken together, for United States Federal Income Tax purposes pursuant to which, for such purposes, the Merger and the Subsequent Merger, taken together, are to be treated as a "reorganization" under section 368(a) of the Code (to which each of Parent, Sister Subsidiary and ECS are to be parties under section 368(b) of the Code) in which ECS is to be treated as merging directly with and into Sister Subsidiary with the Units converted in such Merger as provided for in Section 1.8.

1.5 Governing Documents.

- 1.5.1 At the Effective Time, the Articles of Organization of ECS, as in effect immediately prior to the Effective Time of the Merger, shall be the Articles of Organization of the Surviving Entity. At the effective time of the Subsequent Merger, the Articles of Organization of Sister Subsidiary, as in effect immediately prior to such effective time, shall be the Articles of Organization of Sister Subsidiary until thereafter changed or amended as provided therein or by applicable Law (the "Articles of Organization"), except that Article I thereof shall be amended, by the filing of Articles of Merger or other appropriate documents, to read in its entirety as follows: "the name of the Limited Liability Company is EC Source Services, LLC."
- 1.5.2 At the Effective Time, the operating agreement of ECS, as in effect immediately prior to the Effective Time of the Merger, shall be the operating agreement of the Surviving Entity. At the effective time of the Subsequent Merger, the operating agreement of Sister Subsidiary, as in effect immediately prior to such effective time, shall be the operating agreement of Sister subsidiary until thereafter changed or amended as provided therein or by applicable Law (the "Operating Agreement").
- 1.6 <u>Manager</u>; <u>Board of Directors</u>. The Board of Directors of Merger Subsidiary immediately prior to the Effective Time shall be the Board of Directors of the Surviving Entity at the Effective Time. The Board of Directors of the Surviving Entity immediately prior to the effective time of the Subsequent Merger shall be the Board of Directors of Sister Subsidiary at such effective time, to serve until the earlier of their resignation or removal or until their respective successors are duly elected and qualified, as the case may be.
- 1.7 <u>Management</u>. The officers of Merger Subsidiary immediately prior to the Effective Time shall be the officers of the Surviving Entity at the Effective Time. The officers of the Surviving Entity immediately prior to the effective time of the Subsequent Merger shall be the officers of Sister Subsidiary at such effective time, to serve until the earlier of their resignation or removal or until their respective successors are duly elected and qualified, as the case may be.
- 1.8 Effect on Units. As of the Effective Time of the Merger, by virtue of the Merger and without any further action on the part of Parent, Merger Subsidiary, ECS or any holder of an equity interest of Parent, Merger Subsidiary or ECS:
- 1.8.1 The outstanding shares of common stock of Merger Subsidiary issued and outstanding immediately prior to the Effective Time shall become 100% of the membership interest of the Surviving Entity.
- 1.8.2 Each Common Unit issued and outstanding immediately prior to the Effective Time shall be converted into the right to receive, upon surrender of the certificates formally representing all such membership interest units (the "Certificates"), the Per Unit Merger Consideration. As of the Effective Time, all of the Common Units shall no longer be outstanding and shall automatically be cancelled and shall cease to exist, and the ECS Members shall cease to have any rights with respect thereto, except the right to receive the Per Unit Merger Consideration to be paid and delivered in consideration therefor upon surrender of each such Certificate in accordance with Section 1.12, without interest.
- 1.8.3 Each Preferred Unit shall automatically be retired and cease to be outstanding, and no consideration will be delivered or deliverable in exchange therefor.

1.9 Closing Date Merger Consideration.

- 1.9.1 Per Unit Closing Date Merger Consideration.
 - (a) The "Per Unit Closing Date Merger Consideration" shall be equal to that number of Parent Shares equal to the quotient of:
 - (i) Subject to adjustment as set forth in Sections 1.10.3 and 1.11.4, the sum of (such sum, the "Total Parent Shares")

 (A) 5,153,846 (such number being subject to adjustment as appropriate for any stock splits, stock dividends, share combinations or the like) plus (B) the quotient of (1) the aggregate exercise price of the ECS Options outstanding immediately prior to the Effective Time of the Merger, divided by (2) \$13; divided by

- (ii) the number of Unit Equivalents outstanding immediately prior to the Effective Time of the Merger.
- (b) No later than the second Business Day prior to the Closing Date, ECS shall deliver to Parent Schedule 1.9.1(b) setting forth the name of each ECS Member, the name of each holder of ECS Options (each, an "Option Holder" and collectively, the "Option Holders"), the number of Common Units and/or ECS Options held by each and (i) the Per Unit Closing Date Merger Consideration to be paid by Parent to such ECS Member at the Closing in accordance with Section 1.8.2 and this Section 1.9.1 or (ii) the consideration to be paid by Parent to such Option Holder at the Closing in accordance with Section 1.15 (the "Option Consideration"). Schedule 1.9.1(b) shall also include wire instructions or other payment instructions for each ECS Member for any Earn-Out Payment to be paid in cash, in Parent's sole discretion, as set forth in Section 1.9.2.
 - (c) The Per Unit Closing Date Merger Consideration shall be issued in accordance with Section 1.12.

1.9.2 Earn-Out Payments.

- (a) In addition to the Per Unit Closing Date Merger Consideration, Parent shall pay to each ECS Member such ECS Member's Common Unit Percentage of each Earn-Out Payment (the "Per Unit Earn-Out Merger Consideration") as determined pursuant to this Section 1.9.2 at the times, in the manner and to the extent an Earn-Out Payment is earned. An "Earn-Out Payment" shall be earned for an Earn-Out Period if the Net EBITDA of the Companies on a consolidated basis for such Earn-Out Period is greater than the EBITDA Threshold for such Earn-Out Period and the amount of such Earn-Out Payment shall be equal to (i) 20% multiplied by (ii) the excess of (x) the Net EBITDA of the Companies on a consolidated basis for such Earn-Out Period minus (y) the EBITDA Threshold for such Earn-Out Period.
 - (b) For purposes of this <u>Section 1.9.2</u>, the following terms are defined as follows:
 - (i) "First Earn-Out Period" shall mean the period commencing with the first day of the first calendar month immediately after the Closing Date and ending on the last day of such fiscal year.
 - (ii) "Second Earn-Out Period" shall mean the period commencing with the first day of the first full fiscal year commencing immediately after the Closing Date and ending on the last day of such fiscal year.
 - (iii) "Third Earn-Out Period" shall mean the period commencing with the first day of the second full fiscal year commencing immediately after the Closing Date and ending on the last day of such fiscal year.
 - (iv) "Fourth Earn-Out Period" shall mean the period commencing with the first day of the third full fiscal year commencing immediately after the Closing Date and ending on the last day of such fiscal year.
 - (v) "<u>Fifth Earn-Out Period</u>" shall mean the period commencing with the first day of the fourth full fiscal year commencing immediately after the Closing Date and ending on the last day of such fiscal year.
 - (vi) "Sixth Earn-Out Period" shall mean the period commencing with the first day of the fifth full fiscal year commencing immediately after the Closing Date and ending on the day immediately preceding the fifth anniversary of the first day of the First Earn-Out Period.
 - (vii) Each of the "First Earn-Out Period," "Second Earn-Out Period," "Third Earn-Out Period," "Fourth Earn-Out Period," "Fifth Earn-Out Period" and "Sixth Earn-Out Period" is an "Earn-Out Period."

- (viii) "EBITDA Threshold" means (a) \$15,000,000 for each of the Second Earn-Out Period, Third Earn-Out Period, Fourth Earn-Out Period and Fifth Earn-Out Period and, (b) for each of the First Earn-Out Period and the Sixth Earn-Out Period, an amount equal to the product of \$1,250,000 multiplied by the number of calendar months included in the First Earn-Out Period and Sixth Earn-Out Period, respectively.
- (ix) "Net EBITDA" for any Earn-Out Period means EBITDA of the Companies on a consolidated basis for such Earn-Out Period, minus the amount by which Net EBITDA of the Companies on a consolidated basis was negative for the prior Earn-Out Period, if applicable. For example, if the Companies on a consolidated basis have Net EBITDA in the First Earn-Out Period of negative \$1,000,000 and EBITDA in the Second Earn-Out Period of \$15,500,000, then Net EBITDA for the Second Earn-Out Period would be \$14,500,000.
- (x) "EBITDA" means for any Earn-Out Period the sum of (a) the net income or loss generated by the Companies' operations, on a consolidated basis, as they exist at the Effective Time and as they may grow or contract after the Closing Date in the normal course of business through internal growth and not through acquisitions or other extraordinary or non-recurring transactions (the "Existing Operations"), after deduction of all costs, expenses, interest, taxes, depreciation, amortization, and other proper charges from the Existing Operations (to include costs and expenses for services or benefits provided to the Companies by Parent and its Affiliates but subject to the limitations set forth in this paragraph) (including (i) the cost of any bonuses or incentive payments earned with respect to the applicable period (excluding any Earn-Out Payment), (ii) the cost of any equity granted to employees, or other derivative securities granted during the applicable period, valued in accordance with the Black-Scholes model and amortized over the vesting period of any equity granted, (iii) the cost of capitalized leases, amortized over the lease term and (iv) all interest and amortization expense (but excluding principal repayments) with respect to the Companies' Debt, including all Debt reflected on the Closing Balance Sheet), in each case of (i) through (iv) as attributable to the Existing Operations (to include costs and expenses for services or benefits provided to the Companies by Parent and its Affiliates but subject to the limitations set forth in this paragraph) plus (b) all (i) interest expense (ii) federal, state, or local taxes to the extent based upon net income and not revenue, (iii) depreciation expense, and (iv) amortization expense, in each case of (i) through (iv) as attributable to the Existing Operations, but only to the extent deducted in determining the amount in clause (a). In determining the costs to be deducted in calculating net income or loss under clause (a), any costs for services or benefits provided to the Companies by Parent or any Affiliate of Parent shall not exceed Parent's or such Affiliate's actual cost of furnishing such services or benefits.
- (c) Unless there is a disagreement as described in subsection (e) of this Section 1.9.2, Parent will pay to the ECS Members the Earn-Out Payment with respect to a specific Earn-Out Period, if any Earn-Out Payment is owed, no later than the 121st day after the end of such Earn-Out Period, at Parent's option (i) in cash in immediately available funds or (ii) in Parent Shares, valued at the average daily closing price for Parent Shares as reported by the New York Stock Exchange (or as quoted on such other principal U.S. exchange or over-the-counter market on which Parent Shares are then traded) for those trading days occurring during the 60 calendar days immediately preceding the date such Parent Shares are issued (such aggregate price for any Earn-Out Period, the "Earn-Out Period Share Value") or (iii) in any combination of cash in immediately available funds or Parent Shares. If Parent issues Parent Shares for all, or any portion, of any Earn-Out Payment, such shares, at the time of their issuance, will be validly issued, fully paid and nonassessable.
- (d) Except as set forth in Section 1.9.2(h), the continued employment of the ECS Members by the Companies, Parent or their Affiliates is not a condition to Parent's obligation to pay any Earn-Out Payment, if earned; provided that nothing in this Agreement constitutes an agreement or understanding to employ for a term of years or otherwise guarantee any ECS Member's employment with the Companies, Parent or their Affiliates following the Closing.
- (e) On or prior to the 90th day after the end of the applicable Earn-Out Period, Parent will calculate the Earn-Out Payment, if any, for such period and provide to the Members' Representative Parent's calculation of the Earn-Out Payment. For thirty (30) days following Parent's delivery of the Earn-Out Payment calculation to the Members' Representative (the "Earn-Out Review Period"), the Members' Representative may review the Earn-Out Payment calculation and other documentation. During the Earn-Out Review Period, Parent shall provide the Members' Representative with access to all books and records reasonably requested to review Parent's calculation of the Earn-Out Payment. If the Members' Representative objects to the amount of the Earn-Out Payment for any reason, the Members' Representative shall, on or before the last day of the Earn-Out Review Period, so inform Parent, in writing (an "Earn-Out Objection"), setting forth a specific description of the basis of the Members' Representative's determination and the adjustments to the Earn-Out Payment that the Members' Representative believes should be

made. To the extent any disagreement therewith is not described in an Earn-Out Objection received by Parent on or before the last day of the Earn-Out Review Period, then the Earn-Out Payment calculation prepared by Parent and delivered to the Members' Representative shall be deemed agreed, final and binding on the parties. If the Members' Representative timely delivers an Earn-Out Objection to Parent, and Parent and the Members' Representative are unable to resolve all of their disagreements with respect to the Earn-Out Payment within thirty (30) days after the end of the Earn-Out Review Period, then Parent and the Members' Representative shall jointly retain the CPA Firm, which, acting as an expert and not as an arbitrator, shall determine, on the basis set forth in and in accordance with this Section 1.9.2(e), and only with respect to those specific adjustments in the Earn-Out Objection on which Parent and the Members' Representative have not agreed, whether and to what extent, if any, the Earn-Out Payment amount requires adjustment. Parent and the Members' Representative shall instruct the CPA Firm to deliver its written determination to Parent and the Members' Representative no later than 30 days after submitting the matter to it for resolution. At the time of retention of the CPA Firm, Parent shall specify in writing to the CPA Firm and the Members' Representative the amount of Parent's computation of the Earn-Out Payment (the "Parent's Earn-Out Position"), and the Members' Representative shall specify in writing to the CPA Firm and to Parent the amount of the Members' Representative's computation of the Earn-Out Payment (the "Members' Representative's Earn-Out Position"). The CPA Firm's determination shall be conclusive and binding upon the parties. In resolving any disputed item, the CPA Firm may not assign a value to any disputed item that is greater than the greatest value claimed by Parent or the Members' Representative at the time the CPA Firm is retained or less than the smallest value claimed for the item by Parent or the Members' Representative at such time. The scope of the disputes to be resolved by the CPA Firm is limited to whether the calculation of the Earn-Out Payment was done in a manner consistent with GAAP and in accordance with this Agreement, and the CPA Firm is not to make any other determination unless jointly requested in writing by the Members' Representative and Parent. The fees and disbursements of the CPA Firm and the reasonable attorneys' fees and expenses of the parties relating to the disputes submitted to the CPA Firm (collectively, the "Earn-Out Dispute Expenses") shall be borne (i) jointly and severally by the Members in that proportion equal to a fraction (expressed as a percentage) the numerator of which is equal to Members' Representative's Earn-Out Position minus the Earn-Out Payment determined by the CPA Firm, and the denominator of which is equal to Members' Representative's Earn-Out Position minus Parent's Earn-Out Position and (ii) by Parent in that proportion equal to a fraction (expressed as a percentage) equal to one (1) minus the fraction described in clause (i). For example, if the Members' Representative's Earn-Out Position is that the Earn-Out Payment should be \$5,000,000 and Parent's Earn-Out Position is that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines the Earn-Out Payment should be \$4,000,000, the CPA Firm determines the Out Payment should be \$4,600,000 and the Earn-Out Dispute Expenses are \$50,000, then (i) the Members shall jointly and severally pay \$20,000 (40%) of the Earn-Out Dispute Expenses and (ii) Parent shall pay \$30,000 (60%) of the Earn-Out Dispute Expenses. Parent and the Members shall cooperate with the CPA Firm during its resolution of the disagreement and make readily available to the CPA Firm all relevant books and records and any work papers (including those of the parties' respective accountants, to the extent permitted by such accountants) relating to the determination of the Earn-Out Payment and all other items reasonably requested by the CPA Firm in connection therewith.

- (f) Following the Closing Date, Parent will continue to operate the Companies during the period for which the ECS Members are entitled to Earn-Out Payments substantially as previously operated, subject to the business requirements of Parent and its Affiliates taken as a whole. Parent will be permitted, following the Closing Date, to make changes in its sole discretion to the operations, corporate organization, personnel, accounting practices, and other aspects of the Companies and the Businesses so long as such changes are made in good faith, in the best interests of the Companies, Parent, and their respective Affiliates or to conform to standard practices applicable generally to other Parent Affiliates, and not with the specific intent of reducing amounts that otherwise would be payable to the ECS Members.
- (g) Limited Transferability of Earn-Out. Without Parent's prior written consent, which may be withheld in Parent's sole discretion, each ECS Member's rights to receive Earn-Out Payments, if any, are personal to such ECS Member and shall not, in whole or in part, be sold, transferred, assigned or otherwise conveyed or pledged or encumbered or have Encumbrances created with respect to them; provided, however, that subject to Parent's prior written consent, which consent may not be unreasonably withheld, a Non-Entity Member may transfer such right to a Member of the Immediate Family of such Non-Entity Member as a result of his death or in connection with bona fide estate planning purposes, in each case subject to the transferee's agreeing in writing to the terms and conditions of this Section 1.9.2 and with a copy of such written agreement being promptly provided to Parent.
- (h) Forfeiture of Earn-Out Payments. If Parent reasonably believes any ECS Member or Indirect Member (a " <u>Defaulting Member</u>") is in breach of <u>Section 5.1 or 5.2</u> (each a "<u>Non-Competition Obligation</u>") or <u>Section 5.5</u> (a "<u>Lockup Obligation</u>"), Parent shall provide such Defaulting Member with written notice of such alleged breach (a "<u>Default Notice</u>"), which Default Notice shall include a description of the acts that Parent has reasonably determined constitute a breach of the Non-Competition Obligations or a Lockup Obligation. No later than thirty (30) days after the Defaulting Member's receipt of a Default Notice, such Defaulting Member shall provide Parent with written notice (" <u>Objection Notice</u>") if such Defaulting Member objects to Parent's determination. To the extent a disagreement with a

Default Notice is not described in a timely Objection Notice, no such disagreement may be raised by the Defaulting Member and with respect to any such disagreement, Parent's determination shall be conclusive, final and binding on the parties. In addition to any other remedies available to it, if a Defaulting Member admits in writing to a breach of a Non-Competition Obligation or a Lockup Obligation, fails to timely deliver an Objection Notice as set forth above or is found by a court with jurisdiction to have breached a Non-Competition Obligation or a Lockup Obligation, Parent may terminate all of the Defaulting Member's rights under this Agreement to any Earn-Out Payments, or in the case of an Indirect Member who is a Defaulting Member, all of the Entity Members' rights under this Agreement to any Earn-Out Payments, and which, in any such case, if already paid shall be refunded to Parent by the Defaulting Member. If the Defaulting Member provides Parent with an Objection Notice, Parent and the Defaulting Member shall negotiate in good faith to resolve such differences. If the parties are unable to resolve their disagreement within 60 days after Parent's receipt of the Objection Notice, either party may submit such dispute to a court with jurisdiction as set forth in Section 11.15. Any such termination with respect to the Defaulting Member shall not affect the rights of any other ECS Member under this Agreement to any Earn-Out Payments, except as set forth above with respect to the Indirect Members and Entity Members. Nothing in this Section 1.9.2(h) shall preclude Parent from seeking immediate injunctive relief for any breach or threatened breach of a Non-Competition Obligation or Lockup Obligation; nor shall it limit any other remedies available to Parent under this Agreement or applicable Law.

1.10 Estimate of Net Working Capital, Tangible Net Worth; PP&E.

- 1.10.1 At least five (5) Business Days prior to the Closing Date, ECS shall prepare in accordance with GAAP and deliver to Parent ECS's good faith reasonable estimate of the Companies' consolidated balance sheet as of the Closing Date (the "Estimated Closing Balance Sheet") and a certificate (substantially in the form of Exhibit C) executed by both the chief executive officer and the chief financial officer of each Company (the "Closing Certificate") setting forth ECS's estimate of each of the following as set forth on the Estimated Closing Balance Sheet:
 - (a) Net Working Capital (the "Estimated Net Working Capital");
 - (b) Tangible Net Worth (the "Estimated Tangible Net Worth"); and
 - (c) the net book value of the PP&E (the "Estimated PP&E").
- 1.10.2 Following receipt of the Closing Certificate, ECS shall permit Parent and its Representatives at all reasonable times and upon reasonable notice to review ECS's and its accountants' working papers relating to the Estimated Closing Balance Sheet and Closing Certificate as well as the Companies' accounting books and records relating to the determination of the Estimated Closing Balance Sheet and Closing Certificate, and ECS shall make reasonably available its Representatives responsible for the preparation of the Estimated Closing Balance Sheet and the Closing Certificate in order to respond to the inquiries of Parent.
 - 1.10.3 The Total Parent Shares shall be reduced as follows:
 - (a) if the Estimated PP&E is less than eight million dollars (\$8,000,000) (the "PP&E Target Amount"), the Total Parent Shares shall be reduced by the quotient of (1) the difference between the PP&E Target Amount minus the Estimated PP&E divided by (2) \$13;
 - (b) if the Estimated Net Working Capital is less than one million one hundred thousand dollars (\$1,100,000) (the "Net Working Capital Target Amount"), the Total Parent Shares shall be reduced by the quotient of (1) the difference between the Net Working Capital Target Amount minus the Estimated Net Working Capital, divided by (2) \$13; and
 - (c) if the Estimated Tangible Net Worth is less than eight million dollars (\$8,000,000) (the "Tangible Net Worth Target Amounf"), the Total Parent Shares shall be reduced by the quotient of (1) the difference between the Tangible Net Worth Target Amount minus the Estimated Tangible Net Worth, divided by (2) \$13; provided however, that (i) if the Estimated PP&E is less than the PP&E Target Amount, the Tangible Net Worth Target Amount shall be reduced by the amount of such deficit and (ii) if Estimated Net Working Capital is less than the Net Working Capital Target Amount, the Tangible Net Worth Target Amount shall be reduced by the amount of such deficit.
- 1.10.4 In the event that all of (and only if all of): (i) the Estimated PP&E exceeds the PP&E Target Amount (such excess, if any exists, the "PP&E Excess"), (ii) the Estimated Net Working Capital exceeds the sum of: (x) the Net Working Capital Target Amount plus (y) the Unpaid Preferred Dividend (such excess, if any exists, the "Net Working Capital Excess"), and (iii) the Estimated Tangible Net Worth exceeds the sum of: (x) Tangible Net Worth Target Amount plus (y) the Unpaid Preferred Dividend (such excess, if any exists, the "Tangible Net Worth Excess"), Parent shall pay to each ECS Member immediately prior to the Closing an amount of cash in

immediately available funds equal to such ECS Member's Common Unit Percentage of the Excess Cash. For purposes of this Agreement, "Excess Cash." means the product of: (i) sixty-seven percent (67%) multiplied by (ii) the lesser of (x) the Net Working Capital Excess and (y) the Tangible Net Worth Excess.

1.11 Post-Closing Purchase Price Adjustment.

- 1.11.1 As soon as practicable but in no event more than one hundred twenty (120) days following the Closing Date, Parent shall prepare, or cause to be prepared, and deliver to the Members' Representative: (a) a consolidated balance sheet of the Companies as of the Closing Date prepared in accordance with GAAP (the "Closing Balance Sheet"); and (b) a certificate substantially in the form of Exhibit D executed by the Chief Financial Officer of Parent (the "Final Closing Statement") setting forth each of the following as set forth on the Closing Balance Sheet: (i) the actual Net Working Capital ("Actual Net Working Capital"); (ii) the actual Tangible Net Worth ("Actual Tangible Net Worth"); and (iii) the actual net book value of PP&E (the "Actual PP&E").
- 1.11.2 Members' Representative and its accountants shall complete their review of the Closing Balance Sheet and Final Closing Statement within 30 days after delivery thereof by Parent. During such review period, Parent shall provide Members' Representative with access to all books and records reasonably requested by Members' Representative to review the Closing Balance Sheet and the Final Closing Statement. If the Members' Representative objects to the Closing Balance Sheet or Final Closing Statement for any reason, the Members' Representative shall, on or before the last day of such 30-day period, so inform Parent in writing (a "ECS Members' Objection"), setting forth a specific description of the basis of the Members' Representative's determination and the adjustments to the Closing Balance Sheet or Final Closing Statement that the Members' Representative believes should be made. To the extent any disagreement therewith is not described in a ECS Members' Objection received by Parent on or before the last day of such 30-day period, then all items described on the Closing Balance Sheet and Final Closing Statement delivered by Parent to the Members' Representative shall be deemed agreed, final and binding on the parties.
- 1.11.3 If the Members' Representative timely delivers a ECS Members' Objection to Parent and the Members' Representative and Parent are unable to resolve all of their disagreements with respect to the proposed adjustments set forth in ECS Members' Objection within thirty (30) days following Parent's receipt of the ECS Members' Objection, then they shall jointly retain the CPA Firm, which, acting as an expert and not as an arbitrator, shall determine, on the basis set forth in and in accordance with this Section 1.11, and only with respect to those items in ECS Members' Objection on which the Members' Representative and Parent have not agreed, whether and to what extent, if any, the number of Total Parent Shares or Excess Cash, if applicable, requires adjustment. Parent and the Members' Representative shall instruct the CPA Firm to deliver its written determination to Parent and the Members' Representative no later than 30 days after submitting the matter to it for resolution. At the time of retention of the CPA Firm, Parent shall specify in writing to the CPA Firm and the Members' Representative Parent's computation of the Total Parent Shares and Excess Cash, if applicable (the sum of: (i) such Total Parent Shares plus (ii) the quotient of (x) such Excess Cash divided (y) \$13, the "Parent's Position"), and the Members' Representative shall specify in writing to the CPA Firm and to Parent the Members' Representative's computation of the Total Parent Shares and Excess Cash, if applicable (the sum of: (i) such Total Parent Shares plus (ii) the quotient of (x) such Excess Cash divided by (y) \$13, the "Members' Representative's Position"). The CPA Firm's determination shall be conclusive and binding upon the Members' Representative, Parent and the Members. In resolving any disputed item, the CPA Firm may not assign a value to any disputed item that is greater than the greatest value claimed by the Members' Representative in calculating the Members' Representative's Position or Parent in calculating the Parent's Position or less than the smallest value claimed for such item by the Members' Representative in calculating the Members' Representative's Position or Parent in calculating Parent's Position at such time. The scope of the disputes to be resolved by the CPA Firm is limited to whether the preparation of the Closing Balance Sheet and the calculation of Actual Net Working Capital, Actual Tangible Net Worth and Actual PP&E, were done in a manner consistent with GAAP and otherwise in accordance with this Agreement, and the CPA Firm is not to make any other determination unless jointly requested in writing by the Members' Representative and Parent. The fees and disbursements of the CPA Firm and the reasonable attorneys' fees and expenses of the parties relating to the disputes submitted to the CPA Firm (collectively, the "Merger Consideration Dispute Expenses") shall be borne (i) jointly and severally by the Members in that proportion equal to a fraction (expressed as a percentage rounded to one decimal place) the numerator of which is equal to Members' Representative's Position minus the CPA Firm's Position, and the denominator of which is equal to Members' Representative's Position minus Parent's Position and (ii) by Parent in that proportion equal to a fraction (expressed as a percentage) equal to one (1) minus the fraction described in clause (i). For the purposes of this Agreement the "CPA Firm's Position" means the sum of: (i) the Total Parent Shares determined by the CPA Firm <u>plus</u> (ii) the quotient of (x) the Excess Cash determined by the CPA Firm divided by (y) \$13. For example, if Parent's Position is that Total Shares should be 5,000,000 Parent Shares and Excess Cash \$0, the Members' Representative's Position is that Total Shares should be 5,200,000 Parent Shares and Excess Cash \$1 million, and the CPA Firm determines that the Total Shares should be 5,100,000 Parent Shares and Excess Cash \$0 and the Merger Consideration Dispute Expenses are \$50,000, then: (i) Parent shall pay \$18,050 (36.1%)f the Merger Consideration Dispute Expenses and (ii) the Members shall jointly and severally pay \$31,950 (63.9%) of the Merger Consideration Dispute Expenses. The Members' Representative and Parent shall cooperate with the CPA Firm during its resolution of the disagreement and make readily available to the CPA Firm all relevant books and records and any work papers (including those of the parties' respective accountants, to the extent permitted by such accountants) relating to the Closing Balance Sheet, Final Closing Statement, and the Members' Representative's Objection and all other items reasonably requested by the CPA Firm in connection therewith.

- 1.11.4 The Closing Balance Sheet, Net Working Capital, Tangible Net Worth, and PP&E, as agreed to (or deemed to have been agreed to) between Parent and Members' Representative or as determined by the CPA Firm, as applicable, shall be conclusive and binding on all of the parties hereto and shall be deemed the "<u>Actual Closing Balance Sheet</u>", <u>Actual Net Working Capital</u>, <u>Actual Tangible Net Worth</u>, and <u>Actual PP&E</u>, respectively, for all purposes herein
 - (a) Upon completion of the calculation of the Actual Closing Balance Sheet, Actual Net Working Capital, Actual Tangible Net Worth and Actual PP&E, in accordance with this Section 1.11, the Total Parent Shares shall be recalculated pursuant to Section 1.10.3 and the Excess Cash shall be recalculated pursuant to Section 1.10.4, replacing all references to "Estimated PP&E" with "Actual PP&E", "Estimated Net Working Capital" with "Actual Net Working Capital," "Estimated Tangible Worth" with "Actual Tangible Net Worth," in each case, in each place where such reference appears, and the following adjustments (the "Purchase Price Adjustment") made:
 - (i) If the Total Parent Shares as adjusted pursuant to <u>Section 1.10.3</u> using the Actual Net Working Capital, Actual Tangible Net Worth, and Actual PP&E, shown on the Actual Closing Balance Sheet is greater than the Total Parent Shares as adjusted pursuant to <u>Section 1.10.3</u> using the Estimated Net Working Capital, Estimated Tangible Net Worth, and Estimated PP&E shown on the Estimated Closing Balance Sheet, then Parent shall issue to each ECS Member the number of Parent Shares equal to such ECS Member's Common Unit Percentage of such excess.
 - (ii) If the Total Parent Shares as adjusted pursuant to <u>Section 1.10.3</u> using the Actual Net Working Capital, Actual Tangible Net Worth and Actual PP&E shown on the Actual Closing Balance Sheet is less than the Total Parent Shares as adjusted pursuant to <u>Section 1.10.3</u> using Estimated Net Working Capital, Estimated Tangible Net Worth and Estimated PP&E shown on the Estimated Closing Balance Sheet, then the Members shall jointly and severally pay to Parent an amount in cash in immediately available funds equal to such difference multiplied by \$13 or return to Parent for cancellation that number of Parent Shares equal to such difference.
 - (iii) If the Excess Cash calculated pursuant to Section 1.10.4 using the Actual Net Working Capital, Actual Tangible Net Worth, and Actual PP&E, shown on the Actual Closing Balance Sheet is greater than the Excess Cash calculated pursuant to Section 1.10.4 using the Estimated Net Working Capital, Estimated Tangible Net Worth, and Estimated PP&E shown on the Estimated Closing Balance Sheet, then Parent shall pay to each ECS Member an amount of cash in immediately available funds equal to such ECS Member's Common Unit Percentage of such excess.
 - (iv) If the Excess Cash calculated pursuant to Section 1.10.4 using the Actual Net Working Capital, Actual Tangible Net Worth, and Actual PP&E, shown on the Actual Closing Balance Sheet is less than the Excess Cash calculated pursuant to Section 1.10.4 using the Estimated Net Working Capital, Estimated Tangible Net Worth, and Estimated PP&E shown on the Estimated Closing Balance Sheet, then the Members shall jointly and severally pay to Parent an amount in cash in immediately available funds equal to such difference.
 - 1.12 Transactions at Closing; Exchange of Certificates; Exchange Agent.
- 1.12.1 Exchange Agent. Prior to the Effective Time, Parent shall designate its then transfer agent for Parent Shares to act as agent (the "Exchange Agent") for the ECS Members to receive the Per Unit Closing Date Merger Consideration to which the ECS Members are entitled at Closing. Prior to the Effective Time, Parent shall deposit with the Exchange Agent the Per Unit Closing Date Merger Consideration to which the ECS Members are entitled at Closing.
 - 1.12.2 Procedures for the Issuance of Parent Shares and Option Consideration.
 - (a) Prior to receiving any portion of the Per Unit Closing Date Merger Consideration or Option Consideration, each holder of record of Unit Equivalents, shall have delivered to Parent and, other than the Option Holders, the Exchange Agent (i) a properly completed and duly executed Letter of Transmittal substantially in form of Exhibit F (each individually, a "Letter of Transmittal" and collectively, the "Letters of Transmittal") and (ii) in the case of Common Units, the Certificates held of record by such holder evidencing such holder's Common Units issued and outstanding immediately prior to the Effective Time (or if any Certificate shall have been lost, stolen or destroyed, an affidavit of that fact and, if required by Parent, the execution of an unsecured indemnity agreement reasonably acceptable to Parent by such holder against any claim that may be made against it on account of the alleged loss, theft or destruction of such Certificate). Upon surrender of a Letter of Transmittal and, in the case of the ECS Members, a Certificate to the Exchange Agent (or alternatively, delivery of an affidavit and indemnity agreement reasonably acceptable to Parent), a Unit Equivalent Holder shall be entitled to receive in exchange the consideration set forth in

Section 1.9.1 or 1.15, as applicable. If any portion of the aggregate consideration to be paid as set forth herein is to be paid to a Person other than the Person in whose name the Certificate so surrendered is registered in the case of a holder of Common Units or other than an Option Holder in the case of a holder of ECS Options, it shall be a condition of exchange that (x) such Certificate (in the case of Common Units) or Notice of Assignment (in the case of ECS Options) shall be properly endorsed or otherwise in proper form for transfer and that the Person requesting such exchange shall pay any transfer or other Taxes required by reason of the exchange to a Person other than the registered holder of such Certificate or establish to the reasonable satisfaction of the Exchange Agent and Parent that such Tax has been paid or is not applicable and (y) such Person execute and deliver to Parent a certificate containing an appropriate representation regarding ownership and title to the Unit Equivalents represented by the applicable Certificate(s) or ECS Option and a corresponding unsecured indemnity agreement reasonably acceptable to Parent against any claim that may be made against it on account of Parent's payment of consideration to a Person other than the Person in whose name the surrendered Certificate is registered or a Person other than the Option Holder. As of the Effective Time, each Certificate and ECS Option shall be deemed to represent only the right to receive, upon surrender of such Certificate or delivery of the Letter of Transmittal only, in the case of an ECS Option, in accordance with this Section 1.12.2, the consideration into which the Unit Equivalents represented thereby shall have been converted pursuant to Section 1.9.1 or 1.15, as applicable.

- (b) If, after the Effective Time, Certificates are presented to the Exchange Agent, Parent or the Companies for any reason, they shall be canceled and exchanged for the aggregate consideration set forth above in <u>Section 1.9.1</u> represented by such Certificates.
- (c) Subject to applicable Law, any portion of the Per Unit Closing Date Merger Consideration made available to the Exchange Agent pursuant to Section 1.12.2 that remains unclaimed by the Unit Equivalent Holders one year after the Effective Time shall be returned to Parent and any Unit Equivalent Holder who has not exchanged such Unit Equivalents for the Per Unit Closing Date Merger Consideration in accordance with this Section 1.12.2 prior to that time shall thereafter look only to Parent for payment of the Per Unit Closing Date Merger Consideration.
- (d) None of Parent, the Companies or their respective Affiliates shall be liable to any former Unit Equivalent Holder for any Per Unit Closing Date Merger Consideration or Option Consideration properly delivered to a public official pursuant to, and in accordance with, any applicable abandoned property, escheat or similar Law. If any Certificate has not been surrendered prior to the date that the consideration for such Unit Equivalents would escheat, any Parent Shares represented thereby shall, to the extent permitted by applicable Law, automatically become the property of Parent, free and clear of all claims or interest of any Person previously entitled thereto.
- (e) Fractional Parent Shares. Notwithstanding any other provision of this Agreement to the contrary, no fractional Parent Shares will be issued and any ECS Member entitled to receive a fractional Parent Share but for this Section 1.12.2(e) shall be entitled to receive a cash payment from Parent in lieu thereof, equal to such fraction multiplied by \$13.
- 1.13 <u>Withholding</u>. The Exchange Agent, Parent, ECS or the Surviving Entity (as appropriate) shall be entitled to deduct and withhold from consideration otherwise payable pursuant to this Agreement to Unit Equivalent Holders such amounts as are required to be deducted and withheld with respect to the making of such payment under the Code, or any provision of state, local or foreign Tax Law. To the extent that amounts are so withheld, (i) such withheld amounts shall be treated for all purposes of this Agreement as having been paid to Unit Equivalent Holders and (ii) the Exchange Agent, Parent, ECS or the Surviving Entity (as appropriate) shall provide to the Members' Representative written notice of the amounts so deducted or withheld.
- 1.14 Further Action. If at any time after the Effective Time of the Merger, the Surviving Entity shall consider or be advised that any further assignments or assurances in law or otherwise are necessary or desirable to vest, perfect or confirm, of record or otherwise, in the Surviving Entity, all rights, title and interests in all real estate and other property and all privileges, powers and franchises of ECS and Merger Subsidiary, then the Surviving Entity and its proper officers and directors, in the name and on behalf of ECS and Merger Subsidiary, shall execute and deliver all such proper deeds, assignments and assurances in law and do all things commercially reasonable and proper to vest, perfect or confirm title to such property or rights in the Surviving Entity and otherwise to carry out the purpose of this Agreement, and the proper officers and directors of the Surviving Entity are fully authorized in the name of ECS and Merger Subsidiary or otherwise to take any and all such action. Any and all reasonable costs (including attorneys' fees and expenses) incurred by the Surviving Entity resulting from performing actions authorized herein shall be paid jointly and severally by the Members and Parent shall have the right to set-off such costs against amount otherwise payable to the Unit Equivalent Holders.
- 1.15 Options. ECS shall take all requisite action such that at or immediately prior to the Effective Time, each ECS Option shall be cancelled and without exercise or need for any other action on behalf of any Option Holder other than the delivery of a properly completed and duly executed Letter of Transmittal, Parent shall pay or, at its expense, cause ECS to pay through ECS's payroll system to each Option Holder an amount of cash in immediately available funds equal to the Option Value, where the "Option Value" is equal to the

product of (1) the excess, if any, of (A) the product of (x) the Per Unit Closing Date Merger Consideration multiplied by (y) \$13, minus (B) the applicable exercise price of such ECS Option, multiplied by (2) the number of Common Units issuable upon the exercise of such ECS Option in full immediately prior to the Effective Time.

2. REPRESENTATIONS AND WARRANTIES OF ECS AND THE MEMBERS

In order to induce Parent to enter into and perform this Agreement and to consummate the Contemplated Transactions, ECS and the Members hereby jointly and severally represent and warrant to Parent that the following statements contained in this Section 2 are true and correct as of the date of this Agreement:

2.1 Organization; Predecessors; Subsidiaries.

- 2.1.1 Organization. ECS and ECS Aviation are each a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Nevada with full power and authority to carry on its business as it is now being conducted and to own, operate and lease its properties and assets. Each Subsidiary (other than ECS Aviation) is a corporation duly organized, validly existing and in good standing under the Laws of the State of Nevada with full power and authority to carry on its business as it is now being conducted and to own, operate and lease its properties and assets. Each Company is duly qualified or licensed as a foreign limited liability company or corporation, as the case may be, to do business and is in good standing in every jurisdiction in which the conduct of its business, and the ownership or lease of its properties, require it to be so qualified or licensed; all of such jurisdictions are listed on Schedule 2.1.1. Except as set forth on Schedule 2.1.1, no Company has any subsidiaries, holds any Investments or has any obligation to make any Investments in any Person. The Entity Members are limited liability companies duly organized, validly existing and in good standing under the Laws of the State of Nevada with full power and authority to carry on their business as it is now being conducted and to own, operate and lease their properties and assets. Contained in the Data Room are accurate and complete copies of (x) the Organizational Documents of the Companies and (y) the minute books of the Companies which contain records of all meetings held of, and other actions taken by, their members, managing members, board of directors and any committees appointed by its members, managing members, board of directors or shareholders. ECS's Fifth Amended and Restated Operating Agreement, dated the date hereof, is the sole and exclusive "operating agreement" of ECS for purposes of the NRS, and there are no other agreements, written or oral, that are part of the "operating agreement" for t
- 2.1.2 <u>Predecessors</u>. <u>Schedule 2.1.2</u> sets forth a list of (a) any Person that has ever merged with or into any of the Companies, (b) any Person a majority of whose capital stock (or similar outstanding ownership interests) has ever been acquired by any of the Companies, (c) any Person all or substantially all of whose assets have ever been acquired by any of the Companies and (d) any prior names of each Company or any Person described in clauses (a) through (c) (each such Person, a "<u>Predecessor</u>").

2.2 Capitalization of the Companies.

- 2.2.1 The ECS Members own all of the Common Units which, together with the Preferred Units, are all of the outstanding Units of ECS and except for the ECS Options, which are all set forth in Schedule 2.2.1, there are no other outstanding Equity Securities of ECS. Schedule 2.2.1 sets forth the number of Common Units owned by each ECS Member. Schedule 2.2.1 also sets forth for each ECS Option: (i) the per-unit exercise price payable therefor, (ii) the number of Common Units for which such ECS Option is exercisable, (iii) whether the Option Holder is an employee of any Company, and (iv) whether or not any such ECS Option is an "incentive stock option" as such term is defined in the Code. None of the ECS Options is vested or exercisable, and other than the payment of the Option Consideration in accordance with the terms of this Agreement, no Option Holder shall be entitled to any other remuneration or rights as a result of the Contemplated Transactions. Except as set forth in the Merger Option and for the ECS Options, there are no outstanding options, warrants, instruments, rights (including conversion or preemptive rights and rights of first refusal), proxy or member agreements, or other agreements or instruments of any kind, including convertible debt instruments, for the purchase or acquisition from any Company or Member of any of the Equity Securities of any Company. Except as set forth in the New Operating Agreement, no Company or Member is party or subject to any agreement or understanding that affects or relates to the voting or giving of written consents with respect to any Equity Security of any Company or by a member or manager of any Company. ECS owns of all of the outstanding Equity Securities of the Subsidiaries.
- 2.2.2 The Indirect Members own all of the outstanding Equity Securities of Force Capital. Martin Maslonka and Force Capital own all of the outstanding Equity Securities of FCP Investments. The Indirect Members ownership percentages of Force Capital and Martin Maslonka's and Force Capital's ownership percentages of FCP Investments are as set forth on Schedule 2.2.2. Except as set forth in Schedule 2.2.2, there are no outstanding options, warrants, instruments, rights (including conversion or preemptive rights and rights of first refusal), proxy or member agreements, or other agreements or instruments of any kind, including convertible debt instruments, for the purchase or acquisition from any Entity Member or any Indirect Member of any of the Equity Securities of any Entity Member and no Entity Member or Indirect Member is a party or subject to any agreement or understanding that affects or relates to the voting or giving of written consents with respect to any Equity Security of any Entity Member or by a member or manager of any Entity Member.

- 2.2.3 Each issued and outstanding Equity Security of the Companies and the Entity Members (i) has been duly authorized and validly issued and is fully paid and non-assessable, and (ii) was issued in compliance with all Legal Requirements concerning the issuance of securities. No Equity Security of any Company or Entity Member is subject to, nor was it issued in violation of, any purchase option, call option, right of first refusal or offer, preemptive right, subscription right or any similar right. No Company or Entity Member has violated the 1933 Act, any state "blue sky" or securities laws, any other similar Legal Requirement or any preemptive or other similar rights of any Person in connection with the issuance or redemption of any its Equity Securities. Except for the Equity Securities set forth on Schedule 2.2.2, there are no bonds, debentures, notes, or other Debt or Equity Securities of any Entity Member with voting rights on any matters on which the holders of Equity Securities of the Company may vote.
- 2.2.4 Except as set forth in the New Operating Agreement, no Company or, except as set forth in the Organizational Documents of Force Capital, no Entity Member has granted or agreed to grant any registration rights, including piggyback rights, or any right of first offer to any Person.
- 2.2.5 Encumbrances, etc. Except as set forth in Schedule 2.2.5, the New Operating Agreement or in the award agreements pursuant to which the ECS Options were awarded (the "ECS Award Agreements"), there is no outstanding Contractual Obligation to which any Company or Member is a party or by which it is bound obligating such Company or Member to issue, deliver or sell, or cause to be issued, delivered or sold Equity Securities of any Company or Entity Member. Except as set forth in Schedule 2.2.5 or the New Operating Agreement, there are no outstanding obligations of any Company or Member (contingent or otherwise) to repurchase, redeem or otherwise acquire any Equity Securities of any Company or Entity Member. There are no unit-appreciation rights, ownership interest-based performance units, "phantom" equity rights or other Contractual Obligations (contingent or otherwise) pursuant to which any Person is or may be entitled to receive any payment or other value based on the revenues, earnings or financial performance or other attribute of any Company or Member, the Business of any Company or Member or the assets of any Company or Member or calculated in accordance therewith or to cause any Company or Member to file a registration statement under the 1933 Act, or which otherwise relate to the registration of any Equity Securities of any Company or Entity Member. Except as set forth in Schedule 2.2.5, the New Operating Agreement or in the ECS Award Agreements, there are no voting trusts, proxies or other Contractual Obligations to which any Company or Member is a party or by which any of them is bound with respect to the issuance, holding, acquisition, voting or disposition of any Equity Securities of any Company or Entity Member. Except as set forth in Schedule 2.2.5 or the New Operating Agreement, there are no existing Contractual Obligations between any Company or Member on the one hand, and any other Person, on the other hand, regarding the Preferred Units or any other Equity Securities of any Company or Entity Member. The Common Units and ECS Options are held of record and beneficially owned by the Unit Equivalent Holders free and clear of all Encumbrances, other than those Encumbrances arising under this Agreement or the New Operating Agreement and restrictions on the transfer of securities arising under federal and state securities laws.
- 2.2.6 <u>Debt; Guarantees</u>. The Companies have no Debt except as set forth on <u>Schedule 2.2.6</u>. For each item of Debt, <u>Schedule 2.2.6</u> correctly sets forth the debtor, the principal amount of the Debt, the creditor, the maturity date and the collateral, if any, securing the Debt. Except as set forth in <u>Schedule 2.2.6</u>, no Company has any Liability in respect of a Guarantee of any Liability of any other Person.
- 2.3 Organization, Power and Authorization. ECS and Entity Members have all requisite power and authority, and all other Members have all capacity, to execute and deliver this Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement or to be executed by ECS or Members in connection with the consummation of the Contemplated Transactions, including the Membership Interest Purchase Agreement, the Option Agreement, the Articles of Organization Amendment (as defined in the Membership Interest Purchase Agreement) and the New Operating Agreement (the "Company Agreements") to the extent ECS or the Members are parties thereto, and to consummate the Contemplated Transactions. The execution and delivery by ECS and the Entity Members of this Agreement and the Company Agreements and managers, and no other authorization on the part of ECS or the Entity Members or any other Person is necessary to authorize the execution and delivery of this Agreement and the Company Agreements by ECS or the Entity Members or the consummation by ECS or the Entity Members of the Contemplated Transactions. This Agreement and each Company Agreement has been, or when executed and delivered will be, duly executed and delivered by ECS and Members, and this Agreement and the Company Agreements constitute, or when executed and delivered will constitute, legal, valid and binding obligations of ECS and Members, Enforceable against ECS and Members in accordance with its and their terms.
- 2.4 <u>Authorization of Governmental Authorities</u>. No action by (including any authorization, consent or approval), or in respect of, or filing with, any Governmental Authority is required for, or in connection with, the valid and lawful execution, delivery and performance by ECS and the Members of this Agreement and the Company Agreements or consummation of the Contemplated Transactions by ECS and the Members except for: (a) compliance with the HSR Act, (b) the filing of the Articles of Merger as provided in <u>Section 1.1</u>, (c) the filing in connection with the Subsequent Merger of the filings noted in <u>Schedule 2.4</u>, (d) the other Governmental Authorizations disclosed in <u>Schedule 2.4</u>, and (e) filings disclosed on <u>Schedule 2.4</u> which will be required to (i) reflect the admission of Parent as a member of ECS, (ii) change ECS to a manager-managed limited liability company and identify the members of the board of directors of ECS as its managers, or (iii) obtain new Permits.

- 2.5 <u>Noncontravention</u>. Except as disclosed on <u>Schedule 2.5(a)</u>, none of the execution, and delivery by ECS or the Members of any Company Agreement and performance by ECS and the Members of any Company Agreement (other than this Agreement and the Option Agreement) nor, except as disclosed on <u>Schedule 2.5(b)</u>, the performance by ECS and the Members of this Agreement and the Option Agreement will or the consummation of the transactions contemplated by this Agreement and the Option Agreement will:
 - (a) violate any provision of any Legal Requirement applicable to the Companies or the Members;
 - (b) conflict with, result in a breach or violation of, default under, or give rise to a right for any third-party to accelerate, terminate or obtain any prepayment penalty under (in any such case, with or without notice, lapse of time or both) any Contractual Obligation of any Company or Member;
 - (c) result in the creation or imposition of an Encumbrance upon, or the forfeiture of, any Asset;
 - (d) result in a breach or violation of, or default under, the Organizational Documents of any Company or Member; or
 - (e) require any action by (including any authorization, consent or approval) or in respect of (including notice to), any Person under any Contractual Obligation of any Company or Member.

2.6 Financial Statements.

2.6.1 Financial Statements. Attached to Schedule 2.6.1 is a copy of (a) the consolidated unaudited balance sheet of the Companies, other than ECS Aviation (the "Most Recent Balance Sheet,") as of September 30, 2010 (the "Most Recent Balance Sheet Date") and the related consolidated unaudited statements of income and cash flows of the Companies (other than ECS Aviation) for the nine months then ended (together with the Most Recent Balance Sheet, the "Interim Financials") and (b) the audited balance sheets of the Companies (other than ECS Aviation) as of December 31, 2009, 2008 and 2007 and the related audited statements of income and cash flows of the Companies (other than ECS Aviation) for the fiscal years then ended, accompanied in each case by any notes thereto and the report of the independent registered certified public accounting firm (collectively, the "ECS Audited Financials", and together with the Interim Financials and the financial statements to be delivered pursuant to Section 4.8, the "Financials"). Attached to Schedule 2.6.1 is a copy of (x) the unaudited balance sheet of ECS Aviation as of the Most Recent Balance Sheet Date (the "Most Recent ECS Aviation Balance Sheet") and the related unaudited statement of profit and loss of ECS Aviation for the nine months then ended, and (y) the unaudited blance sheets of ECS Aviation as of December 31, 2009 and 2008 and the related unaudited statements of profit and loss of ECS Aviation for the fiscal years then ended (collectively, the "ECS Aviation Financials"). Except as permitted, and in compliance with all the terms and conditions required, by the New Operating Agreement, since the Most Recent Balance Sheet Date, (i) no Company has distributed, sold or otherwise disposed of any property or other Assets other than in the Ordinary Course of Business, and (ii) no Company has made or granted any cash distributions or dividends. Except as set forth on Schedule 2.6.1 or in the notes to the applicable Financials, there are no extraordinary or non-recurring items of income or expense of the Companies that individually, or in the aggregate, equal or exceed \$300,000 during the periods covered by such Financials. Notwithstanding anything to the contrary contained herein, all representations and warranties in this Agreement with respect to the financial statements of ECS Aviation shall be based on the ECS Aviation Financials unless and until Financials are delivered pursuant to Section 4.8, at which time all representations and warranties in this Agreement with respect to the financial statements of ECS Aviation (other than those that are as of the date hereof) shall be based on such new financial statements and the term Financials, Most Recent Financials, Most Recent Financials Balance Sheet, and similar terms shall thereafter be deemed to include the financial statements of ECS Aviation provided pursuant to Section 4.8.

2.6.2 Compliance with GAAP, etc.

(a) Except as set forth in Schedule 2.6.2(a), the Financials (i) are complete and correct and were prepared in accordance with the books and records of the Companies and (ii) have been prepared in accordance with GAAP (subject, in the case of unaudited Financials, to normal year-end audit adjustments, the effect of which will not, individually or in the aggregate, be materially adverse and, in the case of all unaudited Financials, the absence of footnotes that, if presented, would not differ materially from those included in the ECS Audited Financials). The ECS Aviation Financials are complete and correct and were prepared in accordance with the books and records of ECS Aviation. The Financials fairly present in all material respects in accordance with GAAP the financial position of the Companies as of the respective dates thereof and the results of the operations of the Companies for the respective periods covered thereby. The ECS Aviation Financials fairly present in all material respects the financial position of ECS Aviation as of the respective dates thereof and the results of the operations of ECS Aviation for the respective periods covered thereby. The utilization by the Companies (other than ECS Aviation) of the percentage of completion methodology for the recognition of revenues in the Financials is in conformity with GAAP. The gross profit and gross profit percentage of jobs in process in the Financials have been properly recognized in accordance with GAAP. The Financials contain adequate reserves for the realization of all Assets and for all reasonably anticipated Liabilities in accordance with GAAP.

- (b) The books and records of the Companies (other than ECS Aviation) have been maintained in accordance with sound business practices and all applicable Legal Requirements and reflect in all material respects all financial transactions of the Companies that are required to be reflected in accordance with GAAP. The books and records of ECS Aviation have been maintained in accordance with sound business practices and all applicable Legal Requirements and reflect in all material respects all financial transactions of ECS Aviation. The Companies maintain books and records accurately reflecting in all material respects their Assets and Liabilities and, except as set forth in Schedule 2.6.2(b) maintain proper and adequate internal accounting controls that provide assurance that (i) transactions are executed with management's authorization; (ii) transactions are recorded as necessary to permit preparation of the financial statements of the Companies and to maintain accountability for the Companies' Assets; (iii) access to the Companies' Assets is permitted only in accordance with management's authorization; (iv) the reporting of the Companies' Assets is compared with existing Assets at regular intervals; and (v) accounts, notes and other receivables and inventory are recorded accurately, and proper and adequate procedures are implemented to effect the collection thereof on a current and timely basis.
- 2.7 Absence of Undisclosed Liabilities. As of the date hereof, except as set forth on Schedule 2.7(a), the Companies have no Liabilities except for (a) Liabilities set forth on the face of the Most Recent Balance Sheet or the Most Recent ECS Aviation Balance Sheet, as applicable, or as described in the notes to the ECS Audited Financials, (b) Liabilities incurred in the Ordinary Course of Business since the Most Recent Balance Sheet Date (none of which results from, arises out of, or relates to any breach or violation of, or default under, a Contractual Obligation or Legal Requirement and none of which constitutes a Material Adverse Effect) and (c) Liabilities under any of the Company Agreements; and no facts or circumstances exist that could result in any such Liability. As of the Closing Date, except as set forth on Schedule 2.7(b), the Companies have no Liabilities except for (a) Liabilities set forth on the face of the balance sheet of the most recent Financials delivered pursuant to Section 4.8 or, if none exist, set forth on the face of the Most Recent Balance Sheet or the Most Recent ECS Aviation Balance Sheet, as applicable (such balance sheet, the "Most Recent Financials Balance Sheet" and the date as of which such Financials are presented, the "Most Recent Financials Balance Sheet Date") or described in the notes to the most recent Audited Financials or if none exist, the ECS Audited Financials (such Financials, the "Most Recent Audited Financials" and, collectively with (i) the most recent Unaudited Quarterly Financials, to the extent any such Financials cover periods occurring after the Most Recent Audited Financials and (ii) the most recent Unaudited Monthly Financials, to the extent any such Financials cover periods occurring after the later of the period covered by the Most Recent Quarterly Financials and the period covered by the Most Recent Audited Financials, the "Most Recent Financials"), (b) Liabilities incurred in the Ordinary Course of Business since the Most Recent Financials Balance Sheet Date (none of which results from, arises out of, or relates to any breach or violation of, or default under, a Contractual Obligation or Legal Requirement and none of which constitutes a Material Adverse Effect) and (c) Liabilities under any of the Company Agreements; and no facts or circumstances exist that could result in any such Liability.
- 2.8 <u>Absence of Certain Developments</u>. As of the date hereof, since the Most Recent Balance Sheet Date, except as set forth on <u>Schedule 2.8(a)</u>, and as of the Closing Date, since the Most Recent Financials Balance Sheet Date, except as set forth on <u>Schedule 2.8(b)</u>; the Businesses have been conducted only in the Ordinary Course of Business and:
 - (a) no Company has (i) amended its Organizational Documents, (ii) admitted any Person as a member or shareholder, as applicable, or (iii) issued, sold, granted or otherwise disposed of any Equity Security;
 - (b) no Company has become liable in respect of any Guarantee nor has it incurred, assumed or otherwise become liable in respect of any Debt or made any loans, advances or capital contributions to or Investments in any Person (except for travel advances in the Ordinary Course of Business):
 - (c) no Company has sold, leased, licensed, transferred or otherwise disposed of any of its Assets, except Inventory in the Ordinary Course of Business;
 - (d) no Company has permitted any of its Assets to become subject to an Encumbrance other than a Permitted Encumbrance;
 - (e) no Company has made or committed to make any capital expenditure except in accordance with the Companies' annual capital expenditures budget, copies of which are contained in the Data Room;
 - (f) no Company has (i) made any declaration, setting aside or payment of any distribution with respect to, or any repurchase, redemption or other acquisition of, any Equity Security or (ii) entered into, or performed, any transaction with, or for the benefit of any Member or any Affiliate of any Member;
 - (g) there has been no material loss, destruction, damage or eminent domain taking (in each case, whether or not insured) affecting, the Companies, any Business or any material Asset;

- (h) no Company has increased the Compensation payable or paid, whether conditionally or otherwise, to (i) any employee, consultant, independent contractor or agent other than in the Ordinary Course of Business, (ii) any manager or officer of the Companies or (iii) any Affiliate of the Companies:
- (i) no Company has entered into any Contractual Obligation providing for the employment or consultancy of any Person on a full-time, part-time, consulting or other basis other than in the Ordinary Course of Business or otherwise providing Compensation or other benefits to any Person other than in the Ordinary Course of Business;
- (j) no Company has made any change in its methods of accounting or accounting practices (including with respect to reserves) or its pricing policies, payment or credit practices or failed to pay any creditor any amount owed to such creditor when due or granted any extensions of credit;
- (k) no Company has made, changed or revoked any material Tax election, elected or changed any method of accounting for Tax purposes, settled any Action in respect of Taxes or entered into any Contractual Obligation in respect of Taxes with any Governmental Authority;
 - (1) no Company has terminated or closed any Facility, business or operation;
- (m) no customer or supplier required to be disclosed on <u>Schedule 2.21 (a)</u> or <u>Schedule 2.21(b)</u> has canceled, terminated or otherwise altered (including any reduction in the rate or amount of sales or purchases or change to the supply or credit terms, as the case may be) or notified a Company of any intention to do any of the foregoing or otherwise threatened in writing to cancel, terminate or materially alter (including any reduction in the rate or amount of sales or purchases, as the case may be) its relationship with any Company;
- (n) no insurer (i) has questioned, denied or disputed (or otherwise reserved its rights with respect to) the coverage of any claim pending under any Liability Policy or (ii) has provided any notice of cancellation or any other indication that it plans to cancel any Liability Policy or raise the premiums or materially alter the coverage under any Liability Policy;
 - (o) no Company has adopted any Employee Plan or increased any benefits under any Employee Plan;
- (p) no Company has written off as uncollectible any Accounts Receivable, modified or cancelled any material third-party Debt or written up or written down any of its material Assets or revalued its Inventory;
- (q) no Company has failed to make a scheduled capital expenditure or investment that exceeds one hundred thousand dollars (\$100,000) or failed to pay any trade account payable or any other Liability when due;
 - (r) no Company has failed to reasonably maintain or properly repair any of its material Assets;
- (s) no Company has acquired or agreed to acquire by merging or consolidating with, or by purchasing a substantial portion of the assets of, or by any other manner, any business of any Person or acquired any capital asset or related capital assets with a fair market value in excess of One Hundred Thousand Dollars (\$100,000);
 - (t) no Company has threatened, commenced or settled any Action; and
 - (u) no Company has entered into any Contractual Obligation to do any of the things referred to elsewhere in this Section 2.8.

2.9 Assets.

2.9.1 Ownership of Assets. Except as set forth in Schedule 2.9.1, the Companies have sole and exclusive, good and marketable title to, or, in the case of property held under a lease or other Contractual Obligation, a sole and exclusive, Enforceable leasehold interest in, or right to use, all of their properties, rights and assets, whether real or personal and whether tangible or intangible, including all assets reflected in the Most Recent Balance Sheet or Most Recent ECS Aviation Balance Sheet, as applicable, or acquired after the Most Recent Balance Sheet Date (except for such assets which have been sold or otherwise disposed of since the Most Recent Balance Sheet Date in the Ordinary Course of Business) (collectively, the "Assets"). Except as disclosed on Schedule 2.9.1, none of the Assets is subject to any Encumbrance other than Permitted Encumbrances.

- 2.9.2 <u>Sufficiency of Assets</u>. Except as set forth in <u>Schedule 2.9.2</u>, the Assets comprise all of the assets, properties and rights of every type and description, whether real or personal, tangible or intangible, used or necessary to conduct the Companies' Businesses as now conducted and contemplated to be conducted, including to perform the work required for the Backlog.
- 2.10 Accounts Receivable. Except as set forth in Schedule 2.10, all Accounts Receivable, unbilled invoices, costs in excess of billings, work in process, retainage and other amounts ("Receivables") reflected on the Most Recent Financials Balance Sheet and in the records and books of account of the Companies since the Most Recent Financials Balance Sheet Date through the Closing Date as being due to the Companies have arisen in the Ordinary Course of Business, represent legal, valid, binding and enforceable obligations to the respective Company and, subject only to consistently recorded reserves for bad debts established as of a date prior to the Closing Date in a manner consistent with past practice, have been, or will be, collected or are, or will be, collectible in the aggregate recorded amounts thereof in accordance with their terms (and in no event later than the 90 th day following the Closing Date) and are not and will not be subject to any contests, claims, counterclaims or setoffs. Except as set forth in Schedule 2.10, there has been no material adverse change since (i) the Most Recent Financials Balance Sheet Date in the amount or collectability of the Receivables due to the Companies or the related provisions or reserves from that reflected in the Most Recent Balance Sheet or the Most Recent ECS Aviation Balance Sheet, as applicable, and (ii) the Most Recent Financials Balance Sheet Date in the amount or collectability of the Receivables due to the Companies or reserves from that reflected in the Most Recent Financials Balance Sheet. Except as set forth on Schedule 2.10, (i) no account debtor or note debtor is delinquent for payments in excess of \$50,000 or for more than ninety (90) days, (ii) no account debtor or note debtor has refused or threatened to refuse to pay its obligations to the Companies for any reason, or has otherwise made a claim to set-off or similar claim, and (iii) to ECS's Knowledge no account debtor or note debtor is insolvent or bankrupt.
- 2.11 <u>Backlog. Schedule 2.11</u> sets forth a complete and accurate list of, on a contract-by-contract basis, the Contractual Obligations underlying the Companies' Backlog, including the specific amount of Backlog attributable to each such Contractual Obligation, together with a description of the remaining work to be performed under each such uncompleted Contractual Obligation. The Backlog amounts set forth on <u>Schedule 2.11</u>, both individually and in the aggregate, constitute the Companies' best reasonable estimates of such amounts. Immediately following the date hereof, since the Most Recent Balance Sheet Date, the aggregate amount of Backlog, as adjusted for reductions resulting from the performance of work in the Ordinary Course of Business pursuant to such Contractual Obligations has not been reduced by 5% or more. As of the Closing Date, since the Most Recent Financials Balance Sheet Date, except as set forth on <u>Schedule 2.11</u>, the aggregate amount of Backlog has not been reduced by 5% or more as adjusted for reductions resulting from the performance of work in the Ordinary Course of Business pursuant to such Contractual Obligations. Contained in the Data Room are true, accurate and complete copies of each written Contractual Obligation listed on <u>Schedule 2.11</u>, in each case, as amended or otherwise modified and in effect.

2.12 Real Property.

- 2.12.1 The Companies own no real property and do not have any right, title or interest in and to any real property except as specifically set forth on Schedule 2.12. Schedule 2.12 describes each leasehold interest in real property leased, subleased by, licensed or with respect to which a right to use or occupy has been granted to or by a Company (such leased real property is referred to as the "Real Property"), and specifies the lessor(s) of such Real Property, and identifies each lease or any other Contractual Obligation under which such Real Property is leased by any of the Companies (the "Real Property Leases"). Except as described on Schedule 2.12 there are no written or oral subleases, licenses, concessions, occupancy agreements or other Contractual Obligations granting to any other Person the right of use or occupancy of the Real Property, and there is no Person (other than the Companies and/or any lessee(s) of the Real Property specifically identified on Schedule 2.12 in possession of the Real Property. With respect to each Real Property Lease that is a sublease: (a) to ECS's Knowledge, the representations and warranties set forth in Sections 2.19.2 and 2.19.3 are true and correct with respect to the underlying lease(s) (the "Underlying Leases"); and (b) Schedule 2.12 describes each of the Underlying Leases, if any. The Companies have valid leasehold interests in and to the Real Property and any and all Facilities located thereon, free and clear of all Encumbrances except for Permitted Encumbrances and the Encumbrances set forth on Schedule 2.12.
- 2.12.2 The Real Property Leases do not impose restrictions on any portion of any Business that interfere with such Business. No Company is obligated to pay any leasing or real estate brokerage commission as a result of the Contemplated Transactions. There is no pending or, to ECS's Knowledge, threatened eminent domain taking affecting any of the Real Property. Contained in the Data Room are true, correct and complete copies of all of the Real Property Leases and Underlying Leases, including all amendments, modifications, guaranties, notices, memoranda of lease, estoppel certificates, instruments of title, plats, surveys, title abstracts, title insurance policies, and subordinations, non-disturbance and attornment agreements related thereto. No consents or approvals are required to be obtained under the Real Property Leases and/or the Underlying Leases in connection with the Contemplated Transactions, including from the landlord(s) thereunder, or if such consents or approvals are required, (i) they have been obtained and copies of which are contained in the Data Room, (ii) in the case of third party lenders to Affiliates of any Member of the Company, that are disclosed on Schedule 2.5(b), or (ii) in the case of third party landlord(s) that are not Affiliates of any Member or the Company, that are disclosed on Schedule 2.5(b).
- 2.12.3 No Facility currently existing on the Real Property encroaches upon, and no Facility under construction on the Real Property will encroach upon, the real property or easement estate of any other Person. No facility of any other Person encroaches

upon the Real Property or any easement estate appurtenant to the Real Property. Each Facility is supplied with utilities and other services (including gas, electricity, water, drainage, sanitary sewer, storm sewer, fire protection and telephone) necessary for the operation of such Facility as the same is currently operated or proposed to be operated, and all installation charges for such utilities and other services have been fully paid, and all service charges due prior to the Closing Date for such utilities and other services have been paid when due. Each Facility is in good repair and operating condition, normal wear and tear excepted, with all required maintenance having been regularly performed and, to ECS's Knowledge, free from structural defects. Each parcel of Real Property abuts on, and has direct vehicular access to, a public road, or has access to a public road via a permanent, irrevocable appurtenant easement benefiting the parcel of Real Property, in each case, to the extent necessary for the conduct of the Businesses.

- 2.12.4 All Permits necessary in connection with the construction upon, and present use and operation of, the Real Property and the lawful occupancy thereof have been issued by the appropriate Governmental Authorities. Except as set forth in Schedule 2.12, the current use of the Real Property is in accordance with the certificates of occupancy relating thereto and the terms of any such Permits. All such Permits will continue in full force and effect immediately after giving effect to the Contemplated Transactions. The Real Property and its current use, occupancy and operation by any of the Companies and the Facilities located thereon do not (a) constitute a nonconforming use under any applicable building, zoning, subdivision or other land use or similar Legal Requirement, or (b) otherwise violate or conflict with any covenant, condition, restriction or other Contractual Obligation, including the requirements of any applicable Encumbrances thereto. Except as set forth on Schedule 2.12, none of the Companies nor any Predecessor thereof: (x) is in violation of any Legal Requirement relating to Real Property, including setback requirements, zoning restrictions and ordinances, building, life, access, safety, health and fire codes and ordinances affecting the Real Property; (y) has received notice of any eminent domain, condemnation or similar proceeding pending or, to ECS's Knowledge, threatened, or any Governmental Order relating thereto; or (z) has received notice of any Taxes or assessments (other than ordinary real estate Taxes pending and not yet due and payable) against the Real Property or any contingencies existing under which any assessment for real estate Taxes may be retroactively filed against the Real Property. The Real Property is not located within any flood plain (except in the case of Real Property leased by any Company within a flood plain where such Company's use is permitted by such Company under applicable Legal Requirements) or subject to any similar type of restriction for which any Permits nec
- 2.13 <u>Tangible Personal Property</u>. Except as set forth on <u>Schedule 2.13</u>, all of the fixtures and other improvements to the Real Property included in the Assets (including any Facilities) and all of the tangible personal property other than inventory included in the Assets (the "<u>Equipment</u>") (a) are adequate and suitable for their present and intended uses, (b) are in good working order, operating condition and state of repair, reasonable wear and tear excepted, (c) have no defects (whether patent or latent) that interfere with their intended use and (d) have been maintained in all material respects in accordance with the commercially reasonable standards of any manufacturer or any other governmental or regulatory entities.

2.14 Intellectual Property.

- 2.14.1 Schedule 2.14 identifies all Intellectual Property owned or used by the Companies in their respective Businesses, including all Intellectual Property with respect to Company Technology, and lists: (a) all registered Intellectual Property which has been issued to or is otherwise owned by the Companies or that relates to or is otherwise used by the Companies in the Businesses; and (b) each Contractual Obligation pursuant to which any of the Companies have granted or have been granted rights to any Intellectual Property or to which any of them is otherwise bound to any third party, excluding however, "shrink wrap," "click through" or other standard agreements for commercially available, unmodified third party Software in each case having a purchase price of \$2,500 or less ("Commercial Software"). Except as set forth on Schedule 2.14, No Company has any pending application for registration which such Company has made or otherwise owns with respect to any Intellectual Property. True, accurate and complete copies of all such Contractual Obligations, as amended or otherwise modified and in effect, are contained in the Data Room, together with true, accurate and complete copies of all other written documentation evidencing ownership and prosecution (if applicable) of each such item. Except as set forth in Schedule 2.14, all such Contractual Obligations are in full force and effect. Except as set forth in Schedule 2.14, to ECS's Knowledge, there does not exist any claim, allegation, or basis for any claim or allegation, that any Intellectual Property owned or otherwise used by the Companies is invalid or unenforceable or that the Companies' rights with respect thereto are subject to claims or defenses that would impair or preclude enforcement of such rights, including misuse, laches, acquiescence, statute of limitations, abandonment or fraudulent registration. Schedule 2.14 also identifies each trade name, trade dress and unregistered trademark or service mark used by any of the Companies or in connection with th
- 2.14.2 The Companies either individually or collectively are the sole owners of all rights, title and interest in and to all Intellectual Property with respect to, or have the right to use as specified on Schedule 2.14, all the Company Technology free and clear of any Encumbrances (other than Permitted Encumbrances). Each Company has the right to use all other Technology and Intellectual Property used in its respective Businesses as currently conducted and as currently contemplated to be conducted in the future. The Intellectual Property identified on Schedule 2.14 constitutes all Intellectual Property that is used in the Businesses and all Intellectual Property necessary for the conduct of the Businesses as currently conducted and as contemplated to be conducted in the future. Except as set forth on Schedule 2.14, no Member has any ownership or other interest in any Intellectual Property used in the Businesses.

- 2.14.3 No Company nor any Predecessor thereof (a) has, nor has the conduct of the Businesses, interfered with, infringed upon, misappropriated, diluted or otherwise violated or come into conflict with any Intellectual Property rights of third parties or (b) has received any charge, complaint, claim, demand, or notice alleging any such interference, infringement, misappropriation, dilution or other violation or conflict (including any claim that it must obtain a license or refrain from using any Intellectual Property rights of any third party in connection with the conduct of the Businesses or the use of the Company Technology, or, with respect to any other Technology used in the Businesses, that it must obtain a license that it does not already possess or refrain from using same). To ECS's Knowledge, there does not exist any basis for any such claim or allegation. To ECS's Knowledge, no third party has interfered with, infringed upon, misappropriated, diluted or otherwise violated or come into conflict with any Company Technology or any of the Companies' other Intellectual Property. Except as set forth in Schedule 2.14, no claim or legal proceeding involving any Intellectual Property by or against the Companies is pending or, to ECS's Knowledge, has been threatened. Except as otherwise specified on Schedule 2.14 and except for licenses to Commercial Software, no Company is bound by any Contractual Obligation containing any covenant or other provision relating to Intellectual Property that in any way limits or restricts the Companies' ability to use, exploit, assert, or enforce any of their Intellectual Property or conduct their Businesses as currently conducted and as currently contemplated to be conducted in the future.
- 2.14.4 Except for that Software described on <u>Schedule 2.14</u>, the Companies' Software constitutes all the Software used in or necessary to conduct the Businesses as currently conducted by the Companies and as contemplated to be conducted in the future.
- 2.14.5 Except as set forth in <u>Schedule 2.14</u>, with respect to each item of the Company Technology or other Technology or Intellectual Property used in the Businesses:
 - (a) the Companies possess all right, title, and interest in and to such item, or have the valid and Enforceable right to use same, free and clear of any Encumbrance (other than Permitted Encumbrances);
 - (b) to ECS's Knowledge, such item is not subject to any outstanding Governmental Order, and no Action is pending or threatened, which challenges the legality, validity, enforceability, use or ownership of such item, nor, to ECS's Knowledge, is there any basis for such a challenge; and
 - (c) no Company has agreed to nor otherwise has a Contractual Obligation to indemnify any Person for or against any interference, infringement, misappropriation or other conflict with respect to such item.
- 2.14.6 Schedule 2.14 identifies each item of Technology that any Person other than the Companies owns and that is used by any of the Companies pursuant to any license, sublicense or other Contractual Obligation except for agreements for Commercial Software (the "Licenses"). Except as disclosed on Schedule 2.14 and except for the Licenses, there are no royalties or other amounts payable for the use of any such Technology. Contained in the Data Room are true, accurate and complete copies of all of the Licenses, in each case, as amended or otherwise modified and in effect. Except as disclosed on Schedule 2.14, each License is in full force and effect, is valid and Enforceable and no party thereto is in breach of any of the terms thereof. With respect to each such item identified on Schedule 2.14: (a) to ECS's Knowledge, such item is not subject to any outstanding Governmental Order, and no Action is pending or threatened which challenges the legality, validity or enforceability of such item and (b) no Company has granted any sublicense or similar right with respect to any License covering such item.
- 2.14.7 The Companies have taken reasonable and customary measures and precautions necessary to protect and maintain the confidentiality of all Trade Secrets in which any of the Companies have any right, title or interest and otherwise to maintain and protect all such Trade Secrets. All current employees, consultants and independent contractors of the Companies who are or have been involved in, or who have contributed to, the creation or development of the Companies' owned intellectual property have executed and delivered to such Company an agreement (containing, except as disclosed on Schedule 2.14, no exceptions to or exclusions from the scope of its coverage) that protects proprietary information and assigns to such Company any and all such Intellectual Property or have validly waived or otherwise conveyed the benefit of any rights therein to such Company. Except as disclosed on Schedule 2.14, none of the Companies own any Software or possess any source code for any licensed Company Software.
- 2.14.8 All of the computer systems, including the Software, firmware, hardware, networks, interfaces, platforms and related systems owned or used by the Companies (collectively, the "Company Systems"): (i) are in reasonably satisfactory working order and are scalable to meet current and reasonably anticipated capacity, including the ability to process current and anticipated peak volumes in a timely manner; (ii) have reasonably appropriate security, back ups, disaster recovery arrangements, and hardware and software support and maintenance to minimize the risk of material error, breakdown, failure or security breach occurring and to ensure if such event does occur it does not cause a material disruption to any portion of the Businesses; (iii) are configured and maintained to minimize the effects of viruses; and, (iv) except as disclosed on Schedule 2.14, to ECS's Knowledge, do not contain viruses, Trojan horses, back doors or other malicious code and have not, in the last eighteen (18) months, suffered any failures, breakdowns, continued substandard performance or other adverse events that have caused or could reasonably be expected to result in the substantial disruption or interruption in or to the use of such Company Systems and/or the conduct of the Businesses.

2.14.9 The Companies and their respective employees have complied at all times with all applicable privacy Laws regarding the collection, processing, disclosure and use of all data consisting of personally identifiable information that is or is capable of being associated with specific individuals.

2.15 Legal Compliance; Illegal Payments; Permits.

- 2.15.1 Compliance. The Companies are and have been since inception in compliance with:
- (a) their respective Organizational Documents (including, with respect to ECS, immediately after the date hereof the New Operating Agreement) and there is no basis which could constitute a failure to comply; and
- (b) except as set forth in <u>Schedule 2.15.1</u>, all Legal Requirements and, to ECS's Knowledge, there is no basis which could constitute a failure to comply.
- 2.15.2 Anti-Corruption. No agent, Affiliate, employee or other Person associated with or acting on behalf of any Company directly or indirectly, has (a) made any unlawful contribution, gift, bribe, rebate, payoff, influence payment, kickback, or other payment to any Person, private or public, regardless of form, whether in money, property, or services (i) to unlawfully obtain favorable treatment in securing business, (ii) to unlawfully pay for favorable treatment for business secured, (iii) to unlawfully obtain special concessions or for special concessions already obtained or (iv) for any other illegal or improper purpose or (b) established or maintained any fund or asset for the benefit of any Company that has not been recorded in the books and records of the Companies.
- 2.15.3 <u>Permits</u>. Except as set forth in <u>Schedule 2.15.3</u>, the Companies have been duly granted all Permits under all Legal Requirements necessary for the conduct of the Businesses as presently conducted or contemplated to be conducted in the future. <u>Schedule 2.15.3</u> describes each Permit affecting, or relating to, the Assets or any Business, together with the Governmental Authority or other Person responsible for issuing such Permit. Except as disclosed on <u>Schedule 2.15.3</u>, (a) the Permits are valid and in full force and effect, (b) no Company is in material breach or violation of, or default under, any such Permit, and, to ECS's Knowledge, no basis exists which, with notice or lapse of time or both, would constitute any such breach, violation or default, (c) the Permits will continue to be valid and in full force and effect, on identical terms following the consummation of the Contemplated Transactions, and (d) the Companies have properly and validly completed in all material respects all filings and registrations that are required for the operation of the Businesses. Except as set forth in <u>Schedule 2.15.3</u>, there is no investigation or Action pending or, to ECS's Knowledge, threatened that would result in the termination, revocation, suspension or restriction of any Permit or the imposition of any fine, penalty or other sanctions for violation of any Legal Requirement relating to any Permit.

2.15.4 Additional Compliance Representations.

- (a) No petition under the federal bankruptcy laws or any state insolvency law has been filed by or against, or a receiver, fiscal agent or similar officer appointed by a court for the business or property of, (i) any Company, (ii) any Member, (iii) except as set forth in Schedule 2.15.4, any partnership in which any Company or Member was a general partner at or within two years before the time of such filing, or (iv) except as set forth in Schedule 2.15.4, any corporation or business association of which any Non-Entity Member or Indirect Member was an executive officer at or within two years before the time of such filing.
- (b) No Company or Member, except as set forth in <u>Schedule 2.15.4</u>, within the last ten (10) years, has been convicted in a criminal proceeding (excluding traffic violations and other minor offenses) or is or has been a named subject or target of a pending criminal proceeding or investigation.
- (c) No Company or Member has been the subject of any Governmental Order not subsequently reversed, suspended or vacated, permanently or temporarily enjoining any Company or Member from, or otherwise limiting any Company's or Member's involvement or right to engage in, or association with Persons engaged in, any of the following activities:
 - (i) acting as a futures commission merchant, introducing broker, commodity trading advisor, commodity pool operator, floor broker, leverage transaction merchant, any other Person regulated by the Commodity Futures Trading Commission, or as an associated Person of any of the foregoing, or as an investment adviser, underwriter, broker or dealer in securities, or as an affiliated Person, director or employee of any investment company, bank, savings and loan association or insurance company, or engaging in or continuing any conduct or practice in connection with such activity;
 - (ii) engaging in any type of business practice; or

- (iii) engaging in any activity in connection with the purchase or sale of any security or commodity, or in connection with any violation of federal or state securities laws or federal commodities laws.
- (d) No Company or Member has been found by a Governmental Authority in a civil action or by the Securities and Exchange Commission ("SEC") to have violated any securities Law, where the judgment in such civil action or finding by the SEC has not been subsequently reversed, suspended or vacated.
- (e) No Company or Member has been found by a Governmental Authority in a civil action or by the Commodities Futures Trading Commission to have violated any federal commodities law, where the judgment in such civil action or finding by the Commodity Futures Trading Commission has not been subsequently reversed, suspended or vacated.
- (f) No Company or Member has been the subject of, or a party to, any Governmental Order, not subsequently reversed, suspended or vacated, relating to an alleged violation of:
 - (i) Any federal or state securities or commodities Law; or
 - (ii) Any Law respecting financial institutions or insurance companies including, but not limited to, a temporary or permanent injunction, order of disgorgement or restitution, civil money penalty or temporary or permanent cease-and-desist order, or removal or prohibition order; or
 - (iii) Any Law prohibiting mail or wire fraud or fraud in connection with any business entity.
- (g) No Company or Member has been the subject of, or a party to, any sanction or order, not subsequently reversed, suspended or vacated, of any self-regulatory organization (as defined in Section 3(a)(26) of the Exchange Act), any registered entity (as defined in Section 1(a)(29) of the Commodity Exchange Act), or any equivalent exchange, association, entity or organization that has disciplinary authority over its members or persons associated with a member.
- (h) No Company or Member has been involved in any of the following matters: (a) making any political contribution that is or would be illegal under any Law; (b) the disbursement or receipt of funds of the Companies outside the normal system of accountability; (c) unlawful payments, whether direct or indirect, to or from foreign or domestic governments, officials, employees or agents for purposes other than the satisfaction of lawful obligations, or any transaction which has as its intended effect the transfer of assets of any of the Companies or the Members for the purpose of effecting such payment; (d) the intentionally improper or inaccurate recording of payments and receipts on the books of any Company or Member; or (e) any other matters of a similar nature involving unlawful disbursements of assets of any Company or Member.
- 2.16 <u>Inventories</u>. The Companies maintain sufficient Inventory in the Ordinary Course of Business to conduct their respective Businesses consistent with past practices. The Inventories of the Companies reflected on the Most Recent Financials Balance Sheet and the Most Recent Financials and in the records and books of account of the Companies since the Most Recent Balance Sheet Date are of a quality and a quantity usable or saleable, as the case may be, in the Ordinary Course of Business. All Inventory has been valued at cost, and all unmarketable, returned, rejected, damaged, slow moving or obsolete Inventory has been written off or written down to net realizable value in the appropriate Company's books and records and in the Most Recent Financials Balance Sheet and Most Recent Financials. Each Company has sole custody and control of and maintains adequate insurance coverage on all materials, supplies, parts or other assets delivered to such Company by or on behalf of its customers for use in connection with projects such Company is undertaking for such customers (the "<u>Customer Assets</u>"); except as set forth in <u>Schedule 2.16</u>, no material Customer Asset has been damaged, lost, stolen, or otherwise suffered a material diminution in value from the time of receipt by such Company; and no Company has received notice of any claim, loss, or damage related to the Customer Assets.

2.17 Employee Benefit Plans.

(a) Schedule 2.17(a) contains a true and complete list of each "employee benefit plan" (within the meaning of Section 3(3) of ERISA, including multiemployer plans within the meaning of Section 3(37) of ERISA, each a "Multiemployer Plan"), and all stock purchase, stock option, severance, employment, change-in-control, fringe benefit, collective bargaining, bonus, incentive, deferred compensation, employee loan and all other employee benefit plans, agreements, programs, policies or other arrangements, whether or not subject to ERISA (including any funding mechanism therefor now in effect or required in the future as a result of the Contemplated Transactions or otherwise), whether formal or informal, oral or written, legally binding or not, under which (i) any current or former employee,

director or consultant of any Company (the "Company Employees") has any present or future right to benefits and which are contributed to, sponsored by or maintained by any Company or (ii) any Company has had, has or may have any actual or contingent present or future liability or obligation. All such plans, agreements, programs, policies and arrangements shall be collectively referred to as the "Employee Plans", provided, however, that Schedule 2.17(a) sets forth only the Multiemployer Plans of which ECS has Knowledge.

- (b) Except as set forth in Schedule 2.17(b), with respect to each Employee Plan, other than Multiemployer Plans, ECS has made available in the Data Room a current, accurate and complete copy (or, to the extent no such copy exists, an accurate description) thereof and, to the extent applicable: (i) any contracts or agreements relating to any Employee Plan, other than Multiemployer Plans, including all trust agreements, insurance or annuity contracts, investment management agreements, record keeping agreements and other material documents or instruments related thereto; (ii) the most recent determination letter, if applicable; (iii) any summary plan description and other written communications (or a description of any oral communications) by any Company to the Company Employees concerning the extent of the benefits provided under a Employee Plan, (iv) a summary of any proposed amendments or changes anticipated to be made to the Employee Plans, at any time within the twelve months immediately following the date hereof; and (v) for the three most recent years (A) the Form 5500 and attached schedules, (B) reviewed financial statements, if any, (C) actuarial valuation reports and (D) any non-discrimination testing results.
- (c) (i) Each Employee Plan, other than Multiemployer Plans, has been established and administered in accordance with its terms, and in compliance with the applicable provisions of ERISA, the Code, the Health Insurance Portability and Accountability Act of 1996 and other applicable Laws; (ii) each Employee Plan, which is intended to be qualified within the meaning of Section 401(a) of the Code is so qualified and has received a favorable determination letter, opinion letter, or advisory letter from the Internal Revenue Service upon which it may rely as to its qualification, and nothing has occurred, whether by action or failure to act, that could reasonably be expected to cause the loss of such qualification; (iii) no event has occurred and no condition exists that would subject any Company, either directly or by reason of its affiliation with any member of its "Controlled Group" (defined as any organization which is a member of a controlled group of organizations within the meaning of Sections 414(b), (c), (m) or (o) of the Code), to any Tax, fine, Encumbrance, penalty or other Liability imposed by ERISA, the Code or other applicable Laws; (iv) for each Employee Plan, other than Multiemployer Plans, with respect to which a Form 5500 has been filed, no material change has occurred with respect to the matters covered by the most recent Form since the date thereof; (v) no "reportable event" (as such term is defined in Section 4043 of the Code) that could reasonably be expected to result in Liability has occurred with respect to any Employee Plan, other than Multiemployer Plans, (vi) no nonexempt "prohibited transaction" (as such term is defined in Section 406 of ERISA and Section 4975 of the Code) or "accumulated funding deficiency" (as such term is defined in Section 302 of ERISA and Section 412 of the Code (whether or not waived)) has occurred with respect to any Employee Plan, other than Multiemployer Plans,; (vii) no Employee Plan, other than Multiemployer Plans, has failed or is expected to fail to satisfy the minimum funding standards of Section 302 of ERISA or 412 of the Code or is or is expected to be in "at risk status" within the meaning of Section 430(i)(4) of the Code, and no Employee Plan, other than Multiemployer Plans, has, or is expected to have, an "adjusted funding target attainment percentage", as defined in Section 436 of the Code, that is less than 80%; (viii) there is no present intention that any Employee Plan, other than Multiemployer Plans, be materially amended, suspended or terminated, or otherwise modified to change benefits (or the levels thereof) under any Employee Plan, other than Multiemployer Plans, at any time within the twelve months immediately following the date hereof; (ix) no Employee Plan is a split-dollar life insurance program or otherwise provides for loans to executive officers (within the meaning of the Sarbanes-Oxley Act of 2002); and (x) no Company has incurred any current or projected Liability in respect of post-employment or post-retirement health, medical or life insurance benefits for current, former or retired employees of any Company, except as required to avoid an excise tax under Section 4980B of the Code or otherwise except as may be required pursuant to any other applicable Law.
- (d) Except as set forth in Schedule 2.17(d), no Employee Plan, other than Multiemployer Plans, is an "employee pension benefit plan" (within the meaning of Section 3(2) of ERISA) subject to Title IV of ERISA, and no Company has any obligation to contribute, or has incurred any actual or contingent liability or obligation (including any obligation to make any contribution) to or in respect of any such plan, other than Multiemployer Plans.
- (e) Except as set forth in Schedule 2.17(e), with respect to each Employee Plan, other than Multiemployer Plans, (i) no actions, suits or claims (other than routine claims for benefits in the ordinary course) are pending or to ECS's Knowledge threatened; (ii) to ECS's Knowledge, no facts or circumstances exist that could give rise to any such actions, suits or claims; (iii) no written or oral communication has been received by the Companies from the Pension Benefit Guaranty Corporation (the "PBGC") in respect of any Employee Plan, subject to Title IV of ERISA concerning the funded status of any such plan or any transfer of assets and liabilities from any such plan in connection with the Contemplated Transactions; and (iv) no administrative investigation, audit or other administrative

proceeding by the Department of Labor, the PBGC, the Internal Revenue Service or other governmental agencies is pending, in progress or to ECS's Knowledge threatened (including any routine requests for information from the PBGC).

- (f) Except as set forth in Schedule 2.17(f), no Employee Plan or Legal Requirement exists that, as a result of the execution of this Agreement, approval of this Agreement by the Companies' respective shareholders, members, directors and/or managers, or the Contemplated Transactions (whether alone or in connection with any subsequent event(s)), could (i) result in severance pay, termination indemnity or any similar payment or any increase in severance pay, termination indemnity or any similar payment, (ii) accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or result in any other material obligation pursuant to, any Employee Plan, (iii) limit or restrict the right of any Company to merge, amend or terminate any Employee Plan, (iv) cause any Company to record additional compensation expense on its income statement with respect to any outstanding stock option or other equity-based award, or (v) result in payments under any Employee Plan which would not be deductible under Section 280G of the Code.
- (g) Except as set forth in <u>Schedule 2.17(g)</u>, there has been no amendment to, written interpretation of or announcement (whether or not written) by any Company relating to, or any change in employee participation or coverage under, any Employee Plan that would increase the expense of maintaining such Employee Plan above the level of expense incurred in respect thereof for the most recent fiscal year ended prior to the date hereof.
- (h) No compensation under any Employee Plan subject to Section 409A of the Code is or has been required to be includible in the gross income of any participant or beneficiary by reason of Section 409A(a)(i)(A) of the Code or is or has been subject to any additional tax under Section 409A(a)(i)(B) of the Code, no amounts are or have been includible in the gross income of any participants or beneficiaries by reason of Section 409A(b) of the Code, and the consummation of the Contemplated Transactions with the lapse of time or otherwise will not result in any violation of Section 409A of the Code with respect to any Employee Plan.
- (i) Except as specified on Schedule 2.17(i), with respect to each Multiemployer Plan: (i) all contributions required to be made with respect to employees of the relevant Company have been timely paid; (ii) to ECS's Knowledge, the relevant Company has not and is not expected to incur, directly or indirectly, any withdrawal liability under ERISA with respect to any such plan (whether by reason of the Contemplated Transactions or otherwise); (iii) to ECS's Knowledge, the maximum aggregate amount of any withdrawal liability under ERISA with respect to the Multiemployer Plans (determined as if the relevant Company and members of its Controlled Group incurred a complete withdrawal from all Multiemployer Plans immediately following the Closing Date) would not be in excess of \$10,000; (iv) to ECS's Knowledge, no such plan is (or is expected to be) insolvent or in reorganization and no accumulated funding deficiency (as defined in Section 302 of ERISA and Section 431 of the Code), whether or not waived, exists or is expected to exist with respect to any such plan and no such plan is or is expected to be in "endangered status" or "critical status" within the meaning of Section 432 of the Code; (v) neither the relevant Company, nor any member of its Controlled Group has withdrawn, partially withdrawn, or received any notice of any claim or demand for withdrawal liability or partial withdrawal liability against any of them; and (vi) to ECS's Knowledge, neither the relevant Company, nor any member of its Controlled Group has any actual or contingent liability under Section 4204 of ERISA.
- (j) No employment agreement or other agreement required to be disclosed on <u>Schedule 2.17(a)</u> hereof, would require that the service provider under that agreement and/or his or her spouse or dependants have the right to continue to participate in and remain eligible for the health insurance benefits (as in effect from time to time) offered by the Companies, as applicable, for more than one year, except as otherwise provided pursuant to COBRA or other applicable law.

2.18 Environmental Matters. Except as set forth on Schedule 2.18:

2.18.1 No Company nor any Predecessor has at any time engaged in or permitted any operations or activities upon the Real Property or any portion thereof, for the purpose of or in any way involving the manufacture, treatment, remediation, removal, generation, Release, discharge, disposal, refining or dumping of any Contaminant (whether legal or illegal, accidental or intentional) or the illegal or improper handling, storage, treatment or use of any Contaminant, nor has any Company caused any Contaminant to be constructed, deposited, Released, stored, handled, treated, disposed, leached or otherwise located on or under the Real Property, or any previously owned, leased or operated properties, onto or underneath other properties, nor has any Contaminant migrated or threatened to migrate from other properties onto or under the Real Property, or any previously owned, leased or operated properties;

- 2.18.2 neither the Real Property, nor any previously owned, leased or operated property, is listed or proposed for listing on the NPL pursuant to CERCLA, or listed on the CERCLIS list or any similar state list of sites, and there do not exist any conditions at the Real Property or any previously owned, leased or operated property which, if known to a Governmental Authority, would qualify the Real Property or any previously owned, leased or operated property for inclusion on any such list;
- 2.18.3 all of the third Persons with which any Company or Predecessor has arranged, engaged or contracted to accept, treat, transport, store, dispose of or remove any Contaminant generated or present at the Real Property, or any previously owned, leased or operated property, were properly permitted at the relevant time to perform the foregoing activities or conduct;
- 2.18.4 no Company or Predecessor has transported or arranged for the transport of any Contaminant to any facility or site for the purpose of storage, handling, treatment or disposal which (i) is or was, at the time of disposal, subject to a Remedial Action requirement (other than routine, anticipated, closure-related corrective action obligations affecting closed solid waste management units at such facility) issued under any solid or hazardous waste regulatory Law, (ii) at the time of the disposal had received a notice of violation or was otherwise subject to an enforcement action with respect to alleged violations of any Environmental Law or (iii) is listed or proposed for listing on the NPL pursuant to CERCLA, or listed on the CERCLIS list or any similar state list of sites, and there do not exist any conditions at such facility or site for inclusion on such list;
 - 2.18.5 no Environmental Lien has attached to the Real Property or any previously owned, leased or operated properties;
- 2.18.6 there is not constructed, placed, deposited, stored, disposed of nor located on the Real Property or any previously owned, leased or operated properties any asbestos in any form which has become or threatens to become friable;
- 2.18.7 no underground or surface structures, including storage tanks, landfills, impoundments, waste piles, sumps, or gas or oil wells, or associated piping, are or have ever been located on the Real Property or any previously owned, leased or operated properties;
- 2.18.8 there is not constructed, placed, deposited, released, stored, disposed, leaching nor located on the Real Property or any previously owned, leased or operated properties any polychlorinated biphenyls ("PCBs") or transformers, capacitors, ballasts, or other equipment which contain dielectric fluid containing PCBs, or any other insulating material containing urea formaldehydes;
- 2.18.9 no Company or Predecessor is subject to any pending or, to ECS's Knowledge, threatened investigation, judicial or administrative proceeding, notice, order, judgment, decree or settlement, alleging or addressing in connection with the Businesses or the Real Property or any previously owned, leased or operated properties concerning (i) any violation of any Environmental Law, (ii) any Remedial Action, (iii) any Liabilities arising from the Release or threatened Release of any Contaminant at the Real Property or any previously owned, leased or operated properties or any other location, (iv) any Liabilities for personal injury or threatened personal injury, or injury or threatened injury to property or natural resources or (v) any Environmental Liabilities and Costs;
- 2.18.10 each Company complies and is not in violation of or Liable for any failure to comply, and it and its Predecessors have since inception complied, not violated or become Liable for a failure to comply, with all applicable Environmental Laws, including with respect to the conduct of the Businesses at jobsites and other locations;
 - 2.18.11 there have been no Releases of any Contaminants at, to, or from the Real Property or any previously owned, leased or operated properties;
 - 2.18.12 no Company or Predecessor has any contingent Liability in connection with any Release or threatened Release;
 - 2.18.13 each Company and its Predecessors has reported all violations of Environmental Laws in accordance with applicable Laws;
- 2.18.14 each Company and its Predecessors has corrected all potential violations of Environmental Laws caused by such Company or any Predecessor, including at jobsites and other third Person locations; and
- 2.18.15 ECS and the Members have made available to Parent in the Data Room true and complete copies of, and listed on Schedule 2.18, all Environmental Reports in the possession or control of ECSD, any Company or Member that have ever been in the control or possession of ECSD, any Member or Company or that were prepared for or at the request of ECSD, any Company or Member relating to the Real Property.

2.19 Contracts; Work-in-Process.

- 2.19.1 Contracts. Except as disclosed on Schedule 2.19, no Company is bound by or a party to:
- (a) any Contractual Obligation (or group of related Contractual Obligations) for the purchase or sale of inventory, raw materials, commodities, supplies, goods, products, equipment or other personal property, or for the furnishing or receipt of services, in each case, the performance of which will extend over a period of more than one year or which provides for either an aggregate or annual payments to or by the Companies in the aggregate in excess of fifty thousand dollars (\$50,000);
- (b) (i) any capital lease or (ii) any other lease or other Contractual Obligation relating to the Equipment providing for annual rental payments in excess of fifty thousand dollars (\$50,000), under which any Equipment is held or used by the Companies;
- (c) any Contractual Obligation, other than Real Property Leases or leases relating to the Equipment, relating to the lease or license of any Asset, including Technology and Intellectual Property (and including all customer license and maintenance agreements) that is not included on Schedule 2.14;
- (d) any Contractual Obligation relating to the acquisition or disposition of (i) any business (whether by merger, consolidation or other business combination, sale of securities, sale of assets or otherwise) or (ii) any asset other than in the Ordinary Course of Business;
- (e) any Contractual Obligation under which any Company is, or may become, obligated to pay any amount in respect of indemnification obligations, purchase price adjustment or otherwise in connection with any (i) acquisition or disposition of assets or securities (other than the sale of inventory in the Ordinary Course of Business), (ii) merger, consolidation or other business combination or (iii) series or group of related transactions or events of the type specified in clauses (i) and (ii) above.
- (f) any Contractual Obligation concerning or consisting of a partnership, limited liability company or joint venture agreement or any other relationship involving the sharing of profits, losses or costs;
- (g) any Contractual Obligation (or group of related Contractual Obligations) (i) under which any Company has created, incurred, assumed or guaranteed any Debt or (ii) under which any Company has permitted any Asset to become subject to an Encumbrance;
 - (h) any Contractual Obligation under which any other Person has guaranteed any Debt of the Companies;
- (i) any Contractual Obligation to purchase goods or services exclusively from a given Person or Persons or purchase a minimum amount of goods or services from a given Person or Persons, or all or a portion of the supply of certain goods or services utilized by any Company from a given Person or Persons;
- (j) any Contractual Obligation involving any obligation on the part of any Company or any of their Affiliates to refrain from competing with any Person, from soliciting any employees, independent contractors or customers of any Person or from conducting any other lawful commercial activity (including in any geographic region) or any such Contractual Obligation for any Company's benefit from any other Person(s);
- (k) any Contractual Obligation under which any Company is, or may become, obligated to incur any severance pay or special Compensation obligations which would become payable by reason of this Agreement or the Contemplated Transactions;
- (l) any Contractual Obligation under which any Company is, or may, have any Liability to any investment bank, broker, financial advisor, finder or other similar Person (including an obligation to pay any legal, accounting, brokerage, finder's, or similar fees or expenses in connection with this Agreement or the Contemplated Transactions);
- (m) any profit sharing, equity option, equity purchase, equity appreciation, deferred compensation, severance or other plan or arrangement for the benefit of any Company's current or former managers, members, officers or employees, consultants or independent contractors;

- (n) any Contractual Obligation providing for the employment or consultancy (including on an independent contractor basis) with an individual (or in the case of a consultant or independent contractor, an entity) on a full-time, part-time, consulting or other basis or otherwise providing Compensation or other benefits to any officer, member, manager, employee or consultant (other than an Employee Plan);
 - (o) any agency, dealer, distributor, sales representative, marketing, handler, third party service provider, or other similar agreement;
 - (p) any Contractual Obligation under which any Company has advanced or loaned an amount to any of its Affiliates or employees;
 - (q) any settlement, conciliation or similar Contractual Obligations imposing an obligation on any Company after the Closing Date;
- (r) any Contractual Obligation that limits the ability of any Company or any of its Affiliates to incur any Debt or to Guarantee any Debt or other obligation of any Person, or that limits the amount of any Debt that any Company may incur or Guarantee, or prohibits it from granting any Encumbrance on any Asset to secure any Debt incurred or Guaranteed; or
- (s) any Contractual Obligation not otherwise disclosed on Schedule 2.19 and (A) pursuant to which the Companies have an aggregate future liability to any Person in excess of fifty thousand dollars (\$50,000), (B) entered into other than in the Ordinary Course of Business or other than on arms-length terms, or (C) that is material to the conduct or operation of any Company's Businesses.

Contained in the Data Room are true, accurate and complete copies of each written Contractual Obligation (or to the extent no such copy exists, an accurate description) listed on <u>Schedule 2.19</u>, in each case, as amended or otherwise modified and in effect.

- 2.19.2 Enforceability, etc. Each Contractual Obligation required to be disclosed on Schedule 2.26 (Debt), Schedule 2.12 (Real Property Leases), Schedule 2.11 (Backlog), Schedule 2.14 (Intellectual Property), Schedule 2.17(a) (Employee Benefit Plans), Schedule 2.19 (Contracts), Schedule 2.19.5 (Work-in-Process), Schedule 2.21(a) or Schedule 2.21(b) (Customers and Suppliers), Schedule 2.27 (Insurance), or Schedule 2.31.1 (Surety Bonds) (each, a "Disclosed Contract") is Enforceable against the parties thereto, and is in full force and effect, and will continue to be so Enforceable and in full force and effect on identical terms following the Closing, and (i) except for payments due in connection with the filings disclosed on Schedule 2.4, the consummation of the Contemplated Transactions (other than the transactions contemplated by this Agreement and Option Agreement) shall not (either alone or upon the occurrence of additional acts or events) result in any payment or payments being due from any Company to any Person and (ii) except for payments due in connection with the filings disclosed on Schedule 2.4, subject to obtaining the consents set forth in Schedule 2.5(b), the consummation of the transactions contemplated by this Agreement and Option Agreement shall not (either alone or upon the occurrence of additional acts or events) result in any payment or payments being due from any Company to any Person.
- 2.19.3 <u>Breach, etc.</u> Except as set forth in <u>Schedule 2.19.3</u>, none of the Companies or the Members and, to ECS's Knowledge, none of the other parties to any Disclosed Contract is in material breach or violation of, or default under, or has repudiated any provision of, any Disclosed Contract (including all Surety Bonds, warranty obligations or otherwise), nor has any event occurred which, with the passage of time or the giving of notice, or both, would constitute a material breach or violation of, or default under, any Disclosed Contract (including all Surety Bonds, warranty obligations or otherwise). Except as set forth in <u>Schedule 2.19.3</u>, no Company has received notice from any other party to any Disclosed Contract or, to ECS's Knowledge, has any reason to believe that such party intends to terminate such Disclosed Contract or alter in any way the relationship of the parties under such Disclosed Contract. Except as set forth in <u>Schedule 2.19.3</u>, no party to any Disclosed Contract has given any Company notice of any action to terminate, cancel, rescind or procure a judicial reformation thereof, nor, to ECS's Knowledge, are there any circumstances existing which would lead to any of the foregoing.
- 2.19.4 <u>Bids</u>. <u>Schedule 2.19.4</u> sets forth a complete and accurate list of each outstanding bid or proposal for business submitted by any Company.
- 2.19.5 Except as set forth in Schedule 2.19.5, (i) all work-in-process and contracts underway ("Work-In-Process") constitute work performed pursuant to fully executed written contracts or sales orders taken in the Ordinary Course of Business, (ii) neither a Company nor any customer has been declared to be in default or is in breach of the terms of any obligation to the other, and no valid grounds exist for any set-off of amounts billable to such customers on the completion of orders to which Work-In-Process relates; (iii) all Work-In-Process is of a quality ordinarily produced in accordance with the requirements of the orders to which such Work-In-Process is identified, and will require no rework with respect to services performed prior to Closing except in the Ordinary Course of Business; (iv) all Work-In-Process is being conducted pursuant to fully executed written contracts, orders and change orders issued within the terms of the relationship pursuant to which such Work-In-Process is being conducted; and (v) to ECS's Knowledge, all Work-In-Process could be completed in compliance with the contracts to which each such Work-In-Process relates if managed in the Ordinary Course of Business (and in compliance with construction industry standards and good construction practices) without being performed at or resulting in a loss to the Companies. No Company has any oral agreements for work-in-process or contracts underway.

- 2.20 Affiliate Transactions. Except for the matters disclosed on Schedule 2.20, neither the Companies nor the Members or any Affiliate of any Company or Member is, directly or indirectly, an officer, director, member, manager, employee, consultant, competitor, creditor, debtor, customer, distributor, supplier or vendor of, or is a party to any Contractual Obligation with, any Company. Except as disclosed on Schedule 2.20, no Member or Affiliate of any Member owns or has any ownership interest in any asset used in, or necessary to, the Businesses as conducted or contemplated to be conducted.
- 2.21 Customers and Suppliers. Schedule 2.21(a) sets forth a complete and accurate list of (a) the ten largest customers of each of the Companies (measured by aggregate billings) for the fiscal year ended December 31, 2009 and for the nine months ended on the Most Recent Balance Sheet Date, indicating the existing Contractual Obligations with each such customer by product or service provided and (b) the ten largest suppliers of materials, products or services to each Company (measured by the aggregate amount purchased by the Companies) for the fiscal year ended December 31, 2009 and for the nine months ended on the Most Recent Balance Sheet Date, indicating the Contractual Obligations for continued supply from each such supplier. Schedule 2.21(b) sets forth a complete and accurate list of (a) the ten largest customers of each of the Companies (measured by aggregate billings) for the fiscal year of the Most Recent Audited Financials and for the period commencing immediately after the fiscal year covered by the Most Recent Audited Financials and ended on the date of the Most Recent Financials, indicating the existing Contractual Obligations with each such customer by product or service provided and (b) the ten largest suppliers of materials, products or services to each Company (measured by the aggregate amount purchased by the Companies) for the fiscal year of the Most Recent Audited Financials and for the period commencing immediately after the fiscal year covered by the Most Recent Audited Financials and ended on the date of the Most Recent Financials, indicating the Contractual Obligations for continued supply from each such supplier. Except as set forth on Schedule 2.21(a) or Schedule 2.21(b), the Company's relationships with the customers and the suppliers required to be listed in Schedule 2.21(a) and Schedule 2.21(b) are arms length commercial working relationships and no such customer or supplier has canceled, terminated or otherwise altered (including any reduction in the rate or amount of sales or purchases, material change in the prices charged or paid, or material change to the supply or credit terms, as the case may be) or notified any Company of any intention to do any of the foregoing or otherwise threatened to cancel, terminate or alter (including any reduction in the rate or amount of sales or purchases, in the prices charged or paid, or material change to the supply or credit terms, as the case may be) its relationship with any Company. Except as set forth in Schedule 2.21(a) or Schedule 2.21(b), to ECS's Knowledge, there is no reason to believe that there could be any change in the relationships of the Companies with any of their customers or suppliers as a result of the Contemplated Transactions. Except as set forth in Schedule 2.21(a) or Schedule 2.21(b), the Companies have no notice of any facts or circumstances that have resulted or could result in a change in the relationship that the Companies have with any of their customers and suppliers.
- 2.22 <u>Customer Warranties.</u> Except as set forth in <u>Schedule 2.22</u>, there have been no pending, nor to ECS's Knowledge, threatened, claims under or pursuant to any warranty, whether expressed or implied, on the products or services sold prior to the Closing Date by the Companies that are not disclosed or referred to in the most recent Financials and that are not fully reserved against in the Companies' books and records in accordance with GAAP. Except as set forth in <u>Schedule 2.22</u>, all of the services rendered by the Companies (whether directly or indirectly through independent contractors) have been performed in conformity with all express warranties and, in all material respects, with all applicable Contractual Obligations, and no Company has or shall have any Liability for replacement or repair or for other damages relating to or arising from any such services, except for amounts incurred in the Ordinary Course of Business which are immaterial individually and in the aggregate and not required by GAAP to be disclosed in the Financials. Except as set forth on <u>Schedule 2.22</u>, no Company's Contractual Obligations with any customer contains any unusual warranty provisions that would impose material Liability on any Company. Set forth on <u>Schedule 2.22</u> is the aggregate amount of warranty claims incurred by the Companies during each of the last three completed fiscal years. Except as set forth in <u>Schedule 2.22</u>, to ECS's Knowledge, there is no reason to expect an increase in the amount of warranty claims as a percentage of sales in the future.
- 2.23 <u>Product Liability</u>. Except as disclosed in <u>Schedule 2.23</u>, there is not any claim, or, to ECS's Knowledge, the basis of any claim, against any Company for injury to person or property of employees or any third Persons suffered as a result of the manufacture, sale or distribution of any product or the performance of any service by any Company, including claims arising out of the allegedly defective or unsafe nature of structures built or otherwise worked on by any Company. Except as disclosed in <u>Schedule 2.23</u>, there is no pending or, to ECS's Knowledge threatened investigation of the construction of any structure where construction services were provided by any Company. Except as disclosed in <u>Schedule 2.23</u>, To ECS's Knowledge, there are no Liabilities or threatened claims for warranty obligations other than those arising in the Ordinary Course of Business.
- 2.24 <u>Capital Expenditures and Investments</u>. Each Company's budget for capital expenditures and Investments is set forth on <u>Schedule 2.24</u>, which includes a schedule of all monies disbursed on account of capital expenditures and Investments made by the Companies since the Most Recent Financials Balance Sheet Date.

2.25 Employees.

- 2.25.1 All employees of each Company earning on an annual basis an amount in excess of \$50,000 are, including employer, job title, location and Compensation, listed on <u>Schedule 2.25</u>. Except as set forth on <u>Schedule 2.25</u>, no present or past employee of any Company is bound by a non-competition agreement. Except as set forth on <u>Schedule 2.25</u>, no Company is a party to any Contractual Obligation with any employee or former employee.
- 2.25.2 Except as disclosed on Schedule 2.25, there are no labor troubles (including any work slowdown, lockout, stoppage, picketing or strike) pending, or to ECS's Knowledge, threatened between any Company, on the one hand, and its employees, on the other hand. Except as disclosed on Schedule 2.25, (a) no employee of any Company is represented by a labor union, association or representative body, (b) no Company is a party to, or otherwise subject to, any collective bargaining agreement or other labor union, association or representative body Contractual Obligation, (c) during the past five years there have been no strikes, slowdowns, work stoppages, disputes, lockouts, or to ECS's Knowledge threats thereof, by or with respect to any employees of any Company, (d) no petition has been filed or proceedings instituted by an employee or group of employees of any Company with any labor relations board seeking recognition of a bargaining representative and (e) there is no organizational effort currently being made or threatened by, or on behalf of, any labor union, association or representative body to organize employees of any Company and no demand for recognition of employees of any Company has been made by, or on behalf of, any labor union, association or representative body. No Company is a party to, or otherwise bound by, any consent decree with, or citation or other Governmental Order relating to employees or employment practices. The Companies are in compliance with applicable Legal Requirements, Contractual Obligations, and policies relating to employment, employment practices, wages, hours, and terms and conditions of employment, including the obligations of the Fair Labor Standards Act ("FLSA") and the Worker Adjustment and Retraining Notification Act of 1988 ("WARN"), and all other notification and bargaining obligations arising under any collective bargaining agreement, by Legal Requirement or otherwise. Except as disclosed on Schedule 2.25, no Company has effectuated a "plant closing" or "mass layoff" as those terms are defined in WARN, affecting in whole or in part any site of employment, facility, operating unit or employee of any Company without complying with all provisions of WARN, or implemented any early retirement, separation or window program within the past five years, nor has any Company planned or announced any such action or program for the future. Except as disclosed on Schedule 2.25, no executive officer's or other key employee's employment with any Company has been terminated for any reason nor has any such officer or employee notified the applicable Company of his or her intention to resign or retire.
- 2.25.3 Except as set forth on Schedule 2.25, T&D and its Predecessors are and have been since inception in sole and exclusive control of the labor relations and employment policies affecting their employees; do not and have not since inception shared common supervision with the other Companies; their employees do not also work and have not worked for any of the other Companies; and they are and have been since inception in full compliance with their obligations under each Outside Agreement between Southwestern Line Constructors, NECA and IBEW Local 769. Except as set forth on Schedule 2.25, since inception there have been no grievances, demands for arbitration, lawsuits or unfair labor practice charges filed alleging that T&D or any Predecessor is a single employer, joint employer or alter ego with respect to any entity, and since inception there have been no attempts by any labor organization to enforce Section 2.10 (dual shop) or Section 2.11(a) (work preservation) of the Outside Agreement between Southwestern Line Constructors, NECA and IBEW Local Union 769. Except as set forth on Schedule 2.25, no Company other than T&D has ever had a collective bargaining relationship with a labor union, association or representative body, nor has any labor union, association or representative body ever engaged in an organizational effort to represent any employees of such Companies or made a demand for recognition or filed a petition with any labor relations board seeking to represent any employees of such Companies.
- 2.25.4 No Company is delinquent in payments to any of its employees or consultants for any wages, salaries, overtime pay, commissions, bonuses, vacation pay, sick pay, severance and termination pay, benefits or other Compensation for any services or otherwise arising under any policy, practice, Contractual Obligation, plan, program or Legal Requirement. Except as set forth on Schedule 2.25, no Company's policies or practices are currently being audited or, to ECS's Knowledge investigated by any Governmental Authority. Except as set forth on Schedule 2.25, there is no pending or, to ECS's Knowledge, threatened Action, unfair labor practice charge, grievance, or other charge or inquiry against any Company brought by or on behalf of any employee, prospective employee, former employee, retiree, labor organization or other representative of any Company's employees, or other individual or any Governmental Authority with respect to employment practices brought by or before any Governmental Authority.
- 2.25.5 The Companies are and have been since inception in compliance in all material respects with all applicable Legal Requirements relating to the employment of labor, including applicable laws respecting employment and employment practices, terms and conditions of employment, wages and hours, payroll documents, equal opportunity, immigration compliance, occupational health and safety, termination or discharge, plant closing and mass layoff requirements, affirmative action, workers' compensation, disability, unemployment compensation, whistleblower laws, collective bargaining, the payment of all applicable Taxes including the full payment of all required social security contributions and other required withholdings, and the Companies are not in violation of any applicable Legal Requirements concerning classification or retention of independent contractors.
- 2.25.6 Except as set forth on <u>Schedule 2.25</u>, no Member is, and to ECS's Knowledge, no other employee, officer, manager, contractor or consultant of any Company is obligated under any applicable Legal Requirement or under any Contractual

Obligation of any nature, or is subject to any judgment, decree or Governmental Order that would interfere with the use of such Member's, employee's, officer's, manager's, contractor's or consultant's best efforts to promote the interests of the applicable Company or that would conflict with any Business. The conduct of the Companies' Businesses has not and the contemplated operation of the Companies' Businesses, and the consummation of the Contemplated Transactions, will not, conflict with or result in a breach of the terms, conditions or provisions, or constitute a default under any Contractual Obligation under which any member, employee, officer, director, manager, contractor or consultant of any of the Companies or any of their Predecessors is or was bound (other than the Contractual Obligations listed under item 1 on Exhibit G). The conduct of the Companies' Businesses, the contemplated operation of the Companies' Businesses and the consummation of the Contemplated Transactions, including any prior or future contemplated actions of any member, employee, officer, director, manager, contractor or consultant of any of the Companies or any of their Predecessors in connection therewith, have not conflicted with and will not conflict with or result in a breach of any of the terms, conditions or provisions of, or constitute a default under, any Contractual Obligation listed under item 1 on Exhibit G.

2.26 Litigation; Governmental Orders.

- 2.26.1 <u>Litigation</u>. Except as disclosed on <u>Schedule 2.26</u>, there is no Action to which any Member (related to the Businesses) or any Company is a party (either as plaintiff or defendant) or to which its assets are subject pending, or to ECS's Knowledge, threatened. There is no Action to which any Company or Member is a party (either as plaintiff or defendant) or to which its assets are subject pending, or to ECS's Knowledge, threatened, which (a) in any manner challenges or seeks the rescission of, or seeks to prevent, enjoin, alter or delay the consummation of, or otherwise relates to, this Agreement or the Contemplated Transactions, or (b) may result in any change in the current equity ownership of the Companies, nor, to ECS's Knowledge, is there any basis for any of the foregoing. Except as disclosed on <u>Schedule 2.26</u>, there is no Action that any Member (related to the Businesses) or any Company presently intends to initiate. No Company or Member has any notice of any facts or circumstances that could result in any Action to which it would be a party (either as plaintiff or defendant) or to which its assets would be subject.
- 2.26.2 Governmental Orders. Except as disclosed on Schedule 2.26, no Governmental Order has been issued that is applicable to, or otherwise affects, any Company, or its assets or Businesses.
- 2.27 Insurance. Schedule 2.27 sets forth a true, correct and complete list of all insurance policies by which the Companies, or any of their Assets, employees, officers, managers or directors or the Businesses have been insured (including fire, theft, casualty, comprehensive, general liability, worker's compensation, business interruption, environmental, product liability, automobile, vehicle and equipment insurance policies), or otherwise the beneficiary of coverage of, since December 31, 2007 (the "Liability Policies"). The list includes for each Liability Policy the type of policy, form of coverage, policy number, name of insurer and expiration date. The Companies have made available in the Data Room true, accurate and complete copies of all Liability Policies, in each case, as amended or otherwise modified and in effect. Schedule 2.27 describes any self-insurance arrangements affecting any Company. The Companies have since December 31, 2007 maintained in full force and effect with, to ECS's Knowledge, financially sound and reputable insurers insurance with respect to their Assets and the Businesses, in such amounts and against such losses and risks as is customarily carried by Persons engaged in the same or similar business and as is required under the terms of any applicable Real Property Leases or other Contractual Obligations. Except as disclosed on Schedule 2.27, no insurer (a) has questioned, denied or disputed (or otherwise reserved its rights with respect to) the coverage of any claim pending under any Liability Policy or (b) has provided any notice of cancellation or any other indication and the Companies have no reason to believe that any insurer plans to cancel any Liability Policy or materially raise the premiums or alter the coverage under any Liability Policy. All premiums due and payable under all Liability Policies of the Companies have been paid and, except as set forth in Schedule 2.27 the Companies are not liable for any retroactive premium or similar adjustment. Schedule 2.27 identifies a
- 2.28 <u>Banking Facilities</u>. <u>Schedule 2.28</u> sets forth a true, correct and complete list of: (a) each bank, savings and loan or similar financial institution with which any Company has an account or safety deposit box or other arrangement, and any numbers or other identifying codes of such accounts, safety deposit boxes or such other arrangements maintained by any Company thereat; (b) the names of all Persons authorized to draw on any such account or to have access to any such safety deposit box facility or such other arrangement; and (c) any outstanding powers of attorney executed by or on behalf of any Company.
- 2.29 <u>Powers of Attorney</u>. Except as set forth on <u>Schedule 2.29</u>, no Company has any general or special powers of attorney outstanding (whether as grantor or grantee thereof).
- 2.30 No Brokers. Except for the fees of FBR Capital Markets disclosed on Schedule 2.30(a), no Company or Member has a Liability of any kind to, or is subject to any claim of, any broker, finder or agent in connection with the transactions contemplated by the Company Agreements (other than this Agreement and the Option Agreement). Except for the fees of FBR Capital Markets disclosed on Schedule 2.30(b), no Company or Member has a Liability of any kind to, or is subject to any claim of, any broker, finder or agent in connection with the transactions contemplated by this Agreement or the Option Agreement.

2.31 Surety Bonds.

- 2.31.1 The Companies have posted all deposits, letters of credit, trust funds, bid bonds, performance bonds, reclamation bonds, surety bonds and all such similar undertakings (collectively, "Surety Bonds") required to be posted in connection with their operations. Schedule 2.31.1 contains a true and complete list of all Surety Bonds posted by the Companies in connection with their operations including the name of each surety and the Companies' reasonably best estimated cost of completion for the project or job secured by such Surety Bond.
- 2.31.2 Except as disclosed on <u>Schedule 2.31.2</u>, the Companies are in compliance with all Surety Bonds applicable to each of them, and the operation of their Businesses and the state of reclamation with respect to the Surety Bonds are "current" or in "deferred status" regarding reclamation obligations and otherwise are in compliance with all applicable reclamation and other applicable Legal Requirements. Except as set forth on <u>Schedule 2.31.2</u>, no Company is nor will be required to obtain a substitute Surety Bond with respect to any of its Surety Bonds and the Surety Bonds will remain in effect on identical terms immediately following the Closing.
- 2.32 <u>Reorganizations for Federal Income Tax Purposes</u>. In connection with the qualification of the Merger as a reorganization within the meaning of Sections 368(a)(1)(A) and 368(a)(2)(D) of the Code:
 - (a) The fair market value of the Parent Shares and other consideration received by the Unit Equivalent Holders will be approximately equal to the fair market value of the Unit Equivalents surrendered in the exchange.
 - (b) The Companies and the Members will pay their respective expenses, if any, incurred in connection with the transaction and will not pay any of the expenses of Parent, Merger Subsidiary or Sister Subsidiary incurred in connection with the transaction.
 - (c) There is no intercorporate indebtedness existing between Parent and the Company, between Merger Subsidiary and the Company or between Sister Subsidiary and the Company that was issued, acquired or will be settled at a discount.
 - (d) No Company is an investment company as defined in Section 368(a)(2)(F)(iii) and (iv) of the Code.
 - (e) No Company is under the jurisdiction of a court in a Title 11 or similar case within the meaning of Section 368(a)(3)(A) of the Code.
 - (f) None of the compensation received by any Non-Entity Member—employee of ECS will be separate consideration for, or allocable to, any of his Unit Equivalents; none of the Parent Shares received by any Non-Entity-Member—employee will be separate consideration for, or allocable to, any employment agreement; and the compensation paid to any Non-Entity Member—employee will be for services actually rendered and will be commensurate with amounts paid to third parties bargaining at arm's—length for similar services.
 - (g) Pursuant to the Merger, immediately after the Effective Time, Merger Sub will hold at least ninety percent (90%) of the fair market value of the net assets and at least seventy percent (70%) of the fair market value of the gross assets held by ECS immediately prior to the Effective Time. For purposes of this representation, amounts paid by ECS to its dissenters, amounts paid by ECS to holders of its Equity Securities who receive cash or other property, amounts used by ECS to pay reorganization expenses, and all redemptions and distributions (except for regular, normal dividends) made by ECS will be included as assets of such Company, respectively, immediately prior to the Merger.
- 2.33 No Material Adverse Effect. Except as set forth in Schedule 2.33, since the date of the Most Recent Balance Sheet Date, no event or circumstance has occurred which has had or would reasonably be expected to have a Material Adverse Effect.

2.34 Securities Law Matters.

- 2.34.1 The Parent Shares that may be issued in connection with this Agreement are being acquired by the Members for their own accounts, for investment and not with a view to, or intention of, distribution thereof in violation of the 1933 Act, or any applicable state securities laws, and the Parent Shares will not be disposed of by any Member in contravention of the 1933 Act or any applicable state securities laws.
- 2.34.2 Each Member is an "accredited investor" as defined in Rule 501(a) under the 1933 Act, as amended by the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, possessing such knowledge and experience in financial and business matters and sophistication, as an investor, adequate for the evaluation of the merits and risks of its investment in the Parent Shares and to make an informed investment decision.

- 2.34.3 Each Member has had an opportunity to ask questions and receive answers concerning Parent and the Parent Shares and has had full access to such other information concerning Parent and the Parent Shares as such Member has requested or which has been filed by Parent with the SEC.
- 2.34.4 Each Member has discussed with and relied upon the advice of its independent legal counsel, tax and financial advisors with regard to the meaning and legal consequences of such Member's representations and warranties contained herein and the considerations involved in making an investment in the Parent Shares, and such Member understands that Parent is relying on such Member's representations set forth herein in respect of the offer and sale of the Parent Shares.
- 2.34.5 Each Member acknowledges that the Parent Shares are "restricted securities" as defined by Rule 144 promulgated under the 1933 Act, are not registered under the 1933 Act or any applicable state securities laws and are being issued by Parent to the Members in reliance upon the Section 4(2) private placement exemption contained in the 1933 Act.
- 2.34.6 Each Member understands that he or it must bear the economic risk of his or its acquisition of the Parent Shares for an indefinite period of time because (i) the acquisition of Parent Shares pursuant to this Agreement has not been registered under the 1933 Act and applicable state securities laws; and (ii) the Parent Shares may therefore not be sold, transferred, pledged, or otherwise disposed of unless subsequently so registered or, in the opinion of counsel (reasonably satisfactory to Parent), registration under the 1933 Act or any applicable state securities laws is not required.
 - 2.34.7 Legends. It is understood that the certificates evidencing the Parent Shares may bear one or all of the following legends:
 - (a) "These securities have not been registered under the Securities Act of 1933, as amended. They may not be sold, offered for sale, pledged or hypothecated in the absence of a registration statement in effect with respect to the securities under such Act or an opinion of counsel reasonably satisfactory to the Company that such registration is not required or unless sold pursuant to Rule 144 of such Act."
 - (b) "These securities are additionally subject to the transfer restrictions set forth in that certain Agreement and Plan of Merger, dated November 16, 2010, as amended or otherwise modified (the "Agreement"), by and among MasTec, Inc., ECS Acquisition Merger Subsidiary I, Inc., ECS Acquisition Merger Subsidiary II, LLC, EC Source Services, LLC, Force Capital Partners, LLC, FCP Investments, LLC, Jon Maslonka, Sidney Strauss, Justin Campbell, Casey Maslonka, and Martin Maslonka."
 - (c) Any legend required by Law, including any legend required by the Business Corporation Act of the State of Florida.

3. REPRESENTATIONS AND WARRANTIES OF PARENT

In order to induce ECS and the Members to enter into and perform this Agreement and to consummate the Contemplated Transactions, except with respect to the representations and warranties contained in <u>Sections 3.6</u> and <u>3.7</u> which are being made by Parent solely as of the date hereof, upon exercise of the Merger Option Parent will represent and warrant to the Company and the Members as follows:

- 3.1 Organization. Parent is duly organized, validly existing and in good standing under the laws of the State of Florida and has the corporate power and authority to own its properties and to carry on its business as it is now being conducted.
- 3.2 <u>Power and Authorization</u>. The execution, delivery and performance by Parent of this Agreement and the consummation of the Contemplated Transactions are within the power and authority of Parent and upon or prior to the exercise of the Merger Option will be duly authorized by all necessary action on the part of Parent, and no other corporate or other action on the part of Parent or any other Person will be necessary to authorize the execution and delivery of this Agreement by Parent or the consummation by Parent of the transactions contemplated by this Agreement. This Agreement (a) has been duly executed and delivered by Parent and (b) is a legal, valid and binding obligation of Parent, Enforceable against Parent in accordance with its terms.
- 3.3 <u>Authorization of Governmental Authorities</u>. Except for compliance with the HSR Act, the filing of the Articles of Merger as provided in <u>Section 1.1</u>, and the filing in connection with the Subsequent Merger of the filings noted in <u>Schedule 2.4</u>, no action by (including any authorization, consent or approval), or in respect of, or filing with, any Governmental Authority is required for, or in connection with, the valid and lawful (a) execution, delivery and performance by Parent of this Agreement or (b) consummation of the Contemplated Transactions by Parent.

- 3.4 <u>Noncontravention</u>. Except as set forth in <u>Schedule 3.4</u>, neither the execution, delivery and performance by Parent of this Agreement nor, if the Merger Option is exercised, the consummation of the Contemplated Transactions will:
 - (a) violate any provision of any Legal Requirement applicable to Parent;
 - (b) result in a breach or violation of, or default under, or give rise to a right for any third-party to terminate or any prepayment penalty under any Contractual Obligation of Parent;
 - (c) require any action by (including any authorization, consent or approval) or in respect of (including notice to), any Person under any Contractual Obligation; or
 - (d) result in a breach or violation of, or default under, Parent's Organizational Documents.
 - 3.5 No Brokers. Parent has no Liability of any kind to any broker, finder or agent with respect to the Contemplated Transactions.
- 3.6 <u>Securities</u>. As of the date hereof, Parent has made all filings with the SEC that it is required to make under the Exchange Act including, and since the date of, the filing of its Annual Report on Form 10-K for the year ended December 31, 2009 (collectively, the "<u>Public Reports</u>"). The Public Reports were prepared and comply in all material respects with the requirements of the Exchange Act.
- 3.7 <u>Parent Financial Statements</u>. As of the date hereof, the financial statements of Parent (the "<u>Parent Financial Statements</u>") included in the Public Reports were prepared in accordance with the books and records of Parent and its consolidated subsidiaries and have been prepared in accordance with GAAP. As of the date hereof, the Parent Financial Statements fairly present in all material respects in accordance with GAAP the financial position of Parent and its consolidated subsidiaries as at the respective dates thereof and the results of the operations of Parent and its consolidated subsidiaries for the respective periods covered thereby.

4. PRE-CLOSING COVENANTS

4.1 Access and Investigation. Between the date of this Agreement and the Closing Date, ECS will, and ECS and the Members will cause each of their respective Representatives and the Subsidiaries to, (a) afford Parent and its Representatives, upon reasonable advance notice and during regular business hours full and free access to the Companies' personnel (including, for purposes of negotiating post-Closing employment and non-competition arrangements), properties, jobsites, Contractual Obligations, customers, books and records (including all Tax records), and other documents and data, (b) furnish Parent and its Representatives with and upload to the Data Room copies of all such Contractual Obligations, books and records, and other existing documents and data as Parent and its Representatives may reasonably request, and (c) furnish Parent and its Representatives with and upload to the Data Room such additional financial, operating, and other data and information as Parent or its Representatives may reasonably request; provided that (i) no investigation pursuant to this Section 4.1 shall affect or be deemed to modify any representation or warranty made by the ECS and the Members; (ii) such access does not unreasonably interfere with the normal business operations of the Companies; (iii) that all requests for access shall be directed to the Members' Representative, or such other person as the Companies may designate from time to time; and (iv) the Companies shall not be required to provide access to customers until the Companies arrange meetings with such customers; provided that the Companies shall use their commercially reasonable efforts to arrange such meetings.

4.2 Operation of the Companies and their Businesses.

4.2.1 Ordinary Course. Except as expressly permitted, and in compliance with all the terms and conditions required, by the New Operating Agreement or as expressly consented to in writing by Parent between the date of this Agreement and the Closing Date (the "Pre-Closing Period"), ECS shall, and shall cause the Subsidiaries to, and the Members shall cause the Companies to, act and carry on their respective Businesses solely in the Ordinary Course of Business, consistent with all the terms and conditions of the New Operating Agreement, and shall use commercially reasonable efforts to (i) maintain and preserve their business organization, assets, Governmental Authorizations and properties, (ii) preserve their business relationships with customers, strategic partners, suppliers, distributors and others having business dealings with them, (iii) continue to perform under all Contractual Obligations, (iv) maintain all Liability Policies and (v) keep available the services of their present officers, employees and consultants.

4.2.2 Pre-Closing Negative Covenants.

(a) To the extent permitted by applicable laws, without limiting the generality of the foregoing, except (i) as expressly consented to in writing by Parent, or (ii) as contemplated by this Agreement, during the Pre-Closing Period: ECS and the Members shall not, ECS shall cause the Subsidiaries not to, and the Members shall cause the Companies not to, take any actions that would or would reasonably be expected to result in a breach of any of ECS' and/or any Member's representations, warranties, covenants and other agreements contained in the New Operating Agreement.

- (b) During the Pre-Closing Period, except as consented to in writing by Parent, no Member shall directly or indirectly do any of the following:
 - (i) sell, lease, pledge, or otherwise transfer, dispose of or otherwise encumber or subject (or allow to become subject) to any Encumbrance any Common Units; provided however, that Parent shall not unreasonably withhold its consent to any such transfer by an ECS Member to his or its Affiliate resulting from death or bona fide estate planning purposes; or
 - (ii) authorize or enter into an agreement to do anything prohibited by the foregoing.
- 4.2.3 Option Pre-Closing Period Covenants. To the extent permitted by applicable laws, without limiting the generality of the foregoing, (i) except as expressly consented to in writing by Parent, or (ii) as contemplated by this Agreement, from the date on which Parent exercises the Merger Option to the Closing (the "Option Pre-Closing Period"):
 - (a) ECS and the Members shall not, ECS shall cause the Subsidiaries not to, and the Members shall cause the Companies not to, directly or indirectly do any of the following:
 - (i) declare, set aside or pay any cash or non-cash distributions or dividends (other than cash distributions or dividends from the Subsidiaries to ECS), split, combine or reclassify any Equity Securities or issue or authorize the issuance of any other Equity Securities in respect of, in lieu of or in substitution for its Equity Securities or Debt; or purchase, redeem or otherwise acquire any Equity Securities;
 - (ii) issue, deliver, sell, grant, pledge or otherwise dispose of or encumber any Equity Securities;
 - (iii) amend or adopt any amendments to any Company's or Entity Members' Organizational Documents;
 - (iv) acquire by merging or consolidating with, or by purchasing all or a substantial portion of the assets or any stock of, or by any other manner, any business or any Person or division thereof, or any assets that are material, in the aggregate, to any Company;
 - (v) sell, lease, license, pledge, or otherwise dispose of or otherwise encumber or subject (or allow to become subject) to any Encumbrance, other than Permitted Encumbrances, any of its material properties or Assets, including any capital asset or related capital assets with a fair market value in excess of Ten Thousand Dollars (\$10,000);
 - (vi) knowingly or irrevocably waive any material right of any Company under any Contractual Obligation;
 - (vii) incur any Debt, (B) issue, sell or amend any Debt, (C) Guarantee or otherwise become liable for any Debt of another Person, (D) make any material loans, advances or capital contributions to, or Investment in, any other Person, (E) modify or cancel any material third-party Debt owed to any Company, or (F) enter into any arrangement having the economic effect of the foregoing;
 - (viii) make any single capital expenditure or other expenditures with respect to PP&E in excess of Fifty Thousand Dollars (\$50,000) in the aggregate;
 - (ix) except as required to comply with applicable Laws or the terms of any Employee Plan, (A) adopt, enter into, terminate or amend any Employee Plan, (B) increase the compensation or fringe benefits, or pay any bonus not required by an existing plan, arrangement or agreement to, any officer or employee of any Company, (C) grant any Equity Securities, or (D) take any action other than in the Ordinary Course of Business to fund or in any other way secure the payment of compensation or benefits under any Employee Plan;
 - (x) except as may be required to comply with applicable Tax Laws, make, revoke, amend or change any election in respect of Taxes, file any amendment to a Tax Return, settle any claim or assessment in respect of Taxes, or consent to any extension or waiver of the limitation period applicable to any claim or assessment in respect of Taxes;

- (xi) threaten, commence, pay, discharge, settle or satisfy any Action;
- (xii) terminate or fail to renew any Governmental Authorization that is required for continued operations;
- (xiii) enter into, amend or modify any collective bargaining agreement or union contract with any labor organization or union or waive, release or assign any rights or claims under any such agreement or contract;
- (xiv) accelerate or defer any material obligation or payment by or to any Company, or not pay any accounts payable or other obligation of any Company when due, unless contested in good faith with full and complete appropriate reserves provided in the relevant Company's books and records;
- (xv) decrease or defer in any material respect, any required maintenance, repair or replacement of any PP&E or the level of costs expended in connection therewith;
- (xvi) fail to maintain insurance at levels at least comparable to current levels or otherwise in a manner inconsistent with past practice;
- (xvii) discontinue any line of business or adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, recapitalization or other similar reorganization;
- (xviii) take any action that could or could reasonably be expected to (A) result in a material breach of any representation or warranty of ECS and/or the Members, (B) result in any of the conditions set forth in Section 6 (other than Section 6.1) or Section 7 not being satisfied (other than (i) to the extent a Non-Entity Member is a Required Employee, such Non-Entity Member's own decision not to enter into an employment agreement or noncompetition agreement with Parent, and (ii) a decision by a Member or its Affiliate not to terminate or amend a Real Property Lease or enter into a new Real Property Lease with any Company or Parent or Affiliate of either), or (C) otherwise prevent or materially delay or materially impair its ability to consummate the Contemplated Transactions on the terms contemplated by this Agreement:
- (xix) at any time within the 90-day period prior to the Closing Date, effectuate a "plant closing" or "mass layoff" as those terms are defined in WARN or any similar state Law, affecting in whole or in part any site of employment, facility, operating unit or employee of any Company;
- (xx) make any change to its accounting methods, principles or practices (including with respect to Taxes) or to the Financials or to the working capital policies applicable to any Company, except as required by GAAP;
- (xxi) except for entering into any non-exclusive license agreements in the Ordinary Course of Business, transfer or grant to any third party any rights with respect to any Intellectual Property;
 - (xxii) form any subsidiary or acquire any Equity Security of any Person;
- (xxiii) make any changes to the Budget (as defined in the New Operating Agreement) or any Company's strategic, financial or risk management plan;
- (xxiv) enter into or amend or otherwise modify any transaction, agreement, or contract with any Member or any Affiliate of any Member;
- (xxv) enter into or amend or otherwise modify any transaction, agreement or contract with any third party with a value expected to exceed \$50,000; or
 - (xxvi) authorize or enter into an agreement to do anything prohibited by the foregoing.

- (b) ECS and the Members shall allow, ECS shall cause the Subsidiaries to allow, and the Members shall cause the Companies to allow, Parent to approve or select, as applicable, any Company's:
 - (i) financial plan, including monthly and quarterly income statements, balance sheets and cash flow statements;
 - (ii) annual audited financial statements;
 - (iii) risk management plan;
 - (iv) corporate counsel;
 - (v) authorized signatories and signing authorities on all bank accounts for check signing, money transfer authorities, etc.;
 - (vi) outside consultants with a contract value in excess of \$10,000 or to whom payments are expected to exceed \$10,000 in the aggregate;
 - (vii) expenditure or transaction involving:
 - 1. the procurement of goods, services or other resources for an amount that would exceed the amount budgeted for such items in the Budget (as defined in the New Operating Agreement) by more than 5%, or which when added to all other expenditures and outlays of the Companies for the then applicable Budget period, would cause total actual expenditures and outlays to exceed by more than 1% the total amount of the Budget for such period;
 - 2. the purchase by any Company of assets in excess of \$10,000 not included in such Company's capital expenditures budget;
 - 3. any leasing or purchasing fleet, equipment or tools not included in such Company's capital expenditures budget;
 - 4. a single purchase of inventory (not included in annual budgeted procurement plans) in excess of \$20,000;
 - 5. opening of any letters of credit or bonding of jobs;
 - 6. any other non-ordinary course payment (including the settlement of litigation or claims) in excess of \$10,000;
 - 7. the hiring, termination, election or appointment of any department head, option holder or officer of any Company;
 - 8. changes to compensation programs of any employees at or above the level of department head and/or any other employees (excluding commissioned salesman) with salaries in excess of \$50,000;
 - 9. any Company-wide salary increases that have an annual cost in excess of the prior year's National Consumer Price Index (all urban wage earners);
 - 10. establishment of new businesses or new product lines; or
 - 11. adjustments to reserve accounts in excess of \$10,000 per month.
 - (c) No Member shall directly or indirectly do any of the following:
 - (i) sell, lease, pledge, or otherwise transfer, dispose of or otherwise encumber or subject (or allow to become subject) to any Encumbrance any Common Units; provided however, that Parent shall not unreasonably withhold its consent to any such transfer by an ECS Member to his or its Affiliate resulting from death or bona fide estate planning purposes;

(ii) take any action that would or would reasonably be expected to (A) result in a material breach of a representation or warranty of ECS and/or the Members, (B) result in any of the conditions set forth in Section 6 (other than Section 6.1) or Section 7 not being satisfied (other than (i) to the extent a Non-Entity Member is a Required Employee, such Non-Entity Member's own decision not to enter into an employment agreement or noncompetition agreement with Parent, and (ii) a decision by a Member or its Affiliate not to terminate or amend a Real Property Lease or enter into a new Real Property Lease with any Company or Parent or Affiliate of either), or (C) otherwise prevent or materially delay or materially impair its ability to consummate the Contemplated Transactions on the terms contemplated by this Agreement; or

(iii) authorize or enter into an agreement to do anything prohibited by the foregoing.

4.3 Commercially Reasonable Efforts; Notification; Hart-Scott-Rodino Act Filings .

4.3.1 (a) Upon the terms and subject to the conditions set forth in this Agreement, if Parent exercises the Merger Option, Parent, Merger Subsidiary, Sister Subsidiary, ECS and the Members agree and ECS and the Members agree to cause the Subsidiaries to use commercially reasonable efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the Contemplated Transactions, including (i) preparing and filing of all forms, registrations and notices required to be filed, obtaining of all necessary actions or nonactions, waivers, consents and approvals from Governmental Authorities and the making of all necessary registrations and filings (including filings with Governmental Authorities) and the taking of all commercially reasonable steps as may be necessary to obtain an approval or waiver from, or to avoid an Action by, any Governmental Authority, (ii) obtaining all necessary consents, approvals or waivers from third parties, (iii) defending of any Actions challenging this Agreement or the consummation of the Contemplated Transactions, including seeking to have any stay or temporary restraining order entered by any court or other Governmental Authority vacated or reversed and (iv) the execution and delivery of any additional instruments reasonably necessary to consummate the Contemplated Transactions.

Notwithstanding the foregoing or any other provision of this Agreement to the contrary, in no event shall Parent be obligated to, and the Companies shall not without the prior written consent of Parent, agree or proffer to divest or hold separate, or enter into any licensing, business restriction or similar arrangement with respect to, any assets (whether tangible or intangible) or any portion of any business of Parent, its Affiliates or any Company.

4.3.2 Without limiting the generality of the foregoing, as soon as may be reasonably practicable if Parent exercises the Merger Option, and in no event later than 10 Business Days after the date of such exercise, if such exercise occurs, Parent and ECS each shall file with the FTC and the Antitrust Division of the DOJ Notification and Report Forms relating to the Contemplated Transactions as required by the HSR Act. Subject to applicable Laws and the preservation of any applicable attorney-client privilege, Parent and ECS each shall promptly (i) supply the other with any information that may be required in order to effectuate such filings, (ii) supply any additional information that reasonably may be required by the FTC or the DOJ, and each will use reasonable efforts to obtain a waiver of the applicable waiting period and (iii) make any further filings pursuant thereto that may be necessary, proper or advisable in connection therewith. Parent, on the one hand, and ECS, on the other, shall share equally all filing fees and other charges for the filing under the HSR Act and shall each be responsible for all costs and expenses for obtaining their own respective consents, approvals or waivers from third parties and for defending any actions against them challenging this Agreement or the consummation of the Contemplated Transactions.

4.4 Acquisition Proposals.

4.4.1 ECS and Members shall not, nor shall they permit or authorize any of their Representatives or the Subsidiaries to, directly or indirectly, initiate, solicit, encourage or otherwise facilitate (including by way of furnishing confidential information) any inquiries or the making of any proposal or offer, with respect to (i) any merger, reorganization, share exchange, business combination, recapitalization, consolidation, liquidation, dissolution or similar transaction involving any Company, (ii) any sale, lease, exchange, mortgage, pledge, transfer or purchase of a significant portion of the Assets or any Asset material to the Businesses (other than the sale of Inventory or obsolete PP&E, which has been replaced by all substitute PP&E necessary to continue to operate the Businesses in the Ordinary Course of Business) or Equity Securities of any Company or Entity Member, (iii) any purchase or sale of, or tender offer or exchange offer for Unit Equivalents (any such proposal or offer being hereinafter referred to as an "Acquisition Proposal"). ECS and the Members shall not, nor shall they permit or authorize any of their Representatives or the Subsidiaries to, directly or indirectly, (a) engage in any negotiations concerning, or provide any confidential information or data to, or have any discussions or conversations with, any Person relating to an Acquisition Proposal, or otherwise facilitate any effort or attempt to make or implement or accept an Acquisition Proposal, or (b) enter into any letter of intent or similar document contemplating, or enter into any agreement with respect to, an Acquisition Proposal. Without limiting the foregoing, it is agreed that any violation of the restrictions set forth in this paragraph by any Representative, whether or not such Person is purporting to act on behalf of any Company or Member or otherwise, shall be deemed to be a breach of this Section 4.4 by ECS and the Members.

- 4.4.2 Members' Representative will promptly (and in any event within two (2) Business Days) notify Parent in writing of the existence of any proposal, discussion, negotiation or inquiry received by any Company, any Member or any of their Representatives with respect to any Acquisition Proposal, and the Members' Representative will immediately communicate to Parent the terms of any proposal, discussion, negotiation or inquiry which it or they may receive (including a copy of any such proposal) and the identity of the Person making such proposal or inquiry or engaging in such discussion or negotiation.
- 4.4.3 The Companies and Members will, and will cause their Representatives to, immediately cease and cause to be terminated any existing activities, discussions or negotiations with any Person(s) conducted heretofore with respect to any Acquisition Proposal. In addition, the Companies shall promptly request that each Person who has heretofore received information in connection with such Person's consideration of an Acquisition Proposal return or destroy all confidential information heretofore furnished to such Person by or on behalf of any Company or any Member. Neither ECS nor the Members shall, and each of which shall cause the Subsidiaries not to, release any third party from, or waive any provision of, any confidentiality or standstill agreement to which any Company or Member is a party. ECS and the Members agree that they will take the necessary steps to promptly inform their Representatives of the obligations undertaken by ECS and Members in this Section 4.4.
- 4.5 <u>Bank Accounts</u>; <u>Powers of Attorney</u>. As of the Closing, to the extent requested by Parent, ECS and the Members shall cause Parent's designees to be added, and the Companies' respective designees to be removed, as signatories with respect to each of the Companies' respective bank accounts and to terminate any powers of attorney.
- 4.6 Notice. Between the date hereof and the Closing Date within thirty (30) days of Parent's request (such request not to be made more than two (2) times in any calendar year) and promptly after Parent's exercise of the Merger Option, the Members' Representative will notify Parent in writing (where appropriate and only with respect to matters occurring after the date hereof, through updates to the Disclosure Schedules) of, and contemporaneously will provide Parent with and upload to the Data Room true, accurate and complete copies of any and all information or documents relating to any event, transaction or circumstance, that existed or occurred on, prior to or after the date of this Agreement, as soon as practicable after it becomes known to ECS or any Member, that causes or will cause any representation, warranty, covenant or agreement of ECS and/or the Members under this Agreement to be breached, that renders or will render untrue any representation or warranty of ECS and/or the Members contained in this Agreement as if the same were made on or as of the date of such event, transaction or circumstance, or that make the timely satisfaction of any condition to Closing impossible or unlikely. Both ECS and the Members, on the one hand, and Parent, on the other, shall have the right until the Closing Date to amend or supplement their respective Disclosure Schedules to the representation and warranties in Section 2 and 3. No notice (or updates to Disclosure Schedules) given pursuant to this Section 4.6 shall have any effect on the representations and warranties made as of the date of this Agreement or any covenants or agreements contained in this Agreement for purposes of determining satisfaction of any condition, whether a breach or default has occurred, or the termination or indemnification rights of the parties provided by this Agreement or otherwise.
- 4.7 <u>Consultation</u>. To the extent permitted by applicable Law, (i) during the Pre-Closing Period, ECS and the Members will consult with management of Parent with a view to informing them as to the operation and management of the Companies, and (ii) and during the Option Pre-Closing Period, ECS will cause its management, and will cause the management of the Companies to, participate in monthly operational reviews with the management of Parent.
- 4.8 Interim Financial Statements. During the Pre-Closing Period, ECS and the Members shall cause the Companies to provide Parent with the "Audited Financials," "Unaudited Quarterly Financials" and "Unaudited Monthly Financials," in each case as such term is defined in the New Operating Agreement and subject to the terms and conditions specified in the New Operating Agreement for such financials; provided however, that for the purposes of this Agreement, ECS and the Members shall have: (i) 120 days after the end of each fiscal year to deliver the Audited Financials, (ii) 30 days after the end of each calendar month to deliver the Unaudited Monthly Financials, provided further that for the purposes of this Agreement the Unaudited Monthly Financials shall be required for every calendar month and shall also be required to contain unaudited consolidated statements of cash flows of the Companies for the period from the beginning of the current fiscal year to the end of such month, in each case, in addition to and in compliance with all of the other terms and conditions that apply to such Unaudited Monthly Financials as set forth in the New Operating Agreement and (iii) 30 days after the end of each fiscal quarter to deliver the Unaudited Quarterly Financials; provided further that for the purposes of this Agreement Unaudited Quarterly Financials shall be required for every fiscal year quarter. Upon delivery of the foregoing financial statements to Parent, the representations and warranties set forth in Section 2.6 and other financial statements are the Most Recent Financials as of the Closing Date.

4.9 Environmental Review

4.9.1 Parent, at its sole cost and expense, may undertake "Phase I" environmental site assessments (the "Phase I ESA(s)") of any Asset that is real property. Each Phase I ESA that Parent elects to undertake shall be conducted by a reputable environmental consultant with appropriate licenses, certificates, experience and training selected by the Parent (the "Environmental Consultant"). The Environmental Consultant shall conduct the Phase I ESAs in accordance with the ASTM E1527-00 Standard Practice for Environmental Site Assessments: Phase I Environmental Site Assessment Process (as such practice may have been updated or revised as of

the date hereof, the "ASTM Phase I ESA Standard") and may include such non-scope considerations reasonably specified by the Parent. The Phase I ESAs, however, shall not include testing of any environmental medium, or any materials, of any structure on any real property without the express written approval of ECS, which approval shall not be unreasonably withheld. Notwithstanding the foregoing sentence, any such testing of any environmental medium, or materials, of any structure on any real property during any Phase I ESA shall not unreasonably interfere with the ability of the Companies to conduct their business or result in any material damage to any such structure.

- 4.9.2 Should any Phase I ESA reveal a "recognized environmental condition," as defined in the ASTM Phase I ESA Standard, for which the Environmental Consultant recommends a Phase II ESA (as defined below), then Parent may cause the Environmental Consultant to undertake a reasonable and appropriate subsurface investigation of any such recognized environmental condition at any such real property in a "Phase II" environmental site assessment (the "Phase II ESA(s)") in accordance with ASTM E1903-97 (2002) (as such practice may have been updated or revised as of the date hereof, the "ASTM Phase II ESA Standard"). The Environmental Consultant shall prepare a separate scope of work and cost estimate for each Phase II ESA of each real property that Parent reasonably deems necessary. The performance of such Phase II ESAs shall not unreasonably interfere with the ability of the Companies to conduct their business.
- 4.9.3 After the conclusion of any Phase I or Phase II ESAs, as set forth above, Parent will be responsible to restore the real property to its original condition and repair any damage resulting from such Phase I or Phase II ESA work.

4.10 Employee Benefit Plans

- 4.10.1 Within 90 days of the date hereof: (i) all assets that are attributable to current or former employees (or the beneficiaries of such current or former employees) of an employer participating in the ECS 401K Plan other than the Companies, shall be spun off into a separate employee benefit plan that is: (x) qualified under Section 401(a) of the Code and (y) sponsored by the employer of such current or former employee or employees and (ii) the Participating Employers Addendum to the Adoption Agreement of the ECS 401K Plan shall be amended to remove ECSD as a participating employer in the ECS 401K Plan.
- 4.10.2 ECS and the Members acknowledge that following the date hereof, the Companies and Parent may be members of a single controlled group of corporations under Section 414(b) of the Code or trades or businesses which are under common control under Section 414(c) of the Code. Accordingly, each Member and ECS hereby agrees with Parent that such Member and ECS and their respective Representatives will not, and that such Member will cause its respective Affiliates not to, at any time on or after the date hereof, directly or indirectly, without the prior written consent of the Parent to (a) perform any action, including, but not limited to significantly changing the coverage of the ECS 401K Plan, that would cause the transition rule set forth in Section 410(b)(6)(C)(i) of the Code to no longer apply to any plans intended to be qualified under Section 401(a) of the Code that are maintained by ECS or Parent during the transition period set forth in Section 410(b)(6)(C)(ii) of the Code, or (b) make any election for purposes of determining highly compensated employees under the ECS 401K Plan that would be inconsistent with any election made by Parent with respect to any plan maintained by Parent that is intended to be qualified under Section 401(a) of the Code, including, but not limited to the top-paid group election or the calendar year data election.
- 4.10.3 Each Member and ECS and their respective Representatives shall, or as applicable shall cause their respective Affiliates to, at all times on and after the date hereof, directly or indirectly, (a) operate the ECS 401K Plan in a manner that satisfies the nondiscrimination testing requirements and any other requirements set forth in the Code and the Treasury Regulations thereunder to insure the continued qualification of the ECS 401K Plan under Section 401(a) of the Code, and (b) following the transition period forth in Section 410(b)(6)(C)(ii) of the Code that results from the Contemplated Transactions, provide Parent with any information reasonably requested by Parent for purposes of determining whether the employees of ECS participating in the ECS 401K Plan satisfy the coverage requirements of Section 410(b) of the Code and to the extent ECS fails to satisfy the coverage requirements of Section 410(b) of the Code when tested with the employees of Parent as members of a single controlled group of corporations under Section 414(b) of the Code or trades or businesses which are under common control under Section 414(c) of the Code, cause ECS to comply with Parent's reasonable instructions to insure that the ECS 401K Plan satisfies the nondiscrimination testing requirements, including, modifying the amount of any employer contributions under the ECS 401K Plan, and any other requirements Parent reasonably deems necessary to ensure the continued qualification of the ECS 401K Plan under Section 401(a) of the Code.
- 4.11 Employment Agreements. Each Member and Parent and their respective Representatives will enter or has entered into amendments, or as applicable will cause or has caused their respective Affiliates to enter into amendments, to the employment or service agreements entered into between ECS and each of James Haunty, Robert Dunbar, Chuck Desrosiers, Jon Maslonka, Justin Campbell, Brian Bratton, and Brian Smaistrla and the employment agreements entered into between T&D and William Wright and EEG and Sid Strauss, substantially in the form set forth on Exhibit J, within sixty (60) days of the date hereof.
- 4.12 <u>Health and Life Insurance Plans</u>. Prior to the Effective Time, all employees of an employer, other than the Companies, must cease participating in any "welfare benefit plan" (within the meaning of Section 3(1) of ERISA) or life insurance plan sponsored or maintained by any Company.

POST-CLOSING COVENANTS

5.1 Confidentiality.

5.1.1 The Members acknowledge that the success of the Companies after the Closing Date depends upon the continued preservation of the confidentiality of certain information possessed by the Members, that the preservation of the confidentiality of such information by the Members is an essential premise of the bargain between the parties hereto, and that Parent would be unwilling to enter into this Agreement in the absence of this Section 5.1.1. Accordingly, each Member hereby agrees with Parent that such Member and its Representatives will not, and that such Member will cause its Affiliates not to, at any time on or after the Closing Date, directly or indirectly, without the prior written consent of Parent, disclose or use, any confidential or proprietary information involving or relating to the Businesses, the Companies or Buyer, including (1) customer and supplier information, including lists of names and addresses of customers and suppliers of the Companies, Buyer or any of their respective Affiliates, (2) business plans and strategies, compensation plans, compensation information, sales plans and strategies, pricing and other terms applicable to transactions between existing and prospective customers, suppliers or business associates, (3) market research and databases, sources of leads and methods of obtaining new business, and methods of purchasing, marketing, selling, performing and pricing products and services employed by the Companies or Buyer, (4) information concerning the configuration and architecture, technical data, networks, methods, practices, standards and capacities of the Companies' information systems, Company Software and Company Technology, (5) information identified as confidential and/or proprietary in internal documents of the Companies or documents provided by Buyer, (6) all information posted in the Data Room to the extent such information was produced by or on behalf of any of the Companies and (7) all information that would be a trade secret under any applicable Law; provided, however, that the information subject to the foregoing provisions of this sentence will not include any information generally available to, or known by, the public (other than as a result of disclosure in violation hereof or any other agreement); and provided, further, that the provisions of this Section 5.1.1 will not prohibit any retention of copies of records or disclosure (a) required by any applicable Legal Requirement so long as reasonable prior notice is given to Parent of such disclosure and a reasonable opportunity is afforded to Parent to contest the same or (b) made in connection with the enforcement of any right or remedy relating to, or the performance of any obligation arising under, this Agreement or the Contemplated Transactions. Each Member agrees that it will be responsible for any breach or violation of the provisions of this Section 5.1.1 by any of its Representatives and Affiliates.

5.1.2 Notwithstanding the foregoing, each Member, and each of their Representatives may disclose to any and all Persons, without limitation of any kind, the Tax treatment and Tax structure of the Contemplated Transactions and all materials of any kind (including opinions or other Tax analyses) that are provided to the Members relating to such Tax treatment and Tax structure.

5.2 Restrictive Covenants.

- 5.2.1 Noncompetition. Each Member (other than Strauss) on his or its own behalf and on behalf of each of his or its Affiliates, covenants that from the Closing Date through the later of the seven-year anniversary of the Closing Date or 24 months after such Member's termination of employment with any of the Companies, Parent, the Surviving Entity or any of their respective Affiliates (provided that a transfer of a Member's employment between such entities shall not be deemed a termination of employment), neither he or it nor his or its Affiliates will for his, its or their own account, jointly with another, or for or on behalf of any Person, directly or indirectly, anywhere within the United States of America, Canada, or Mexico (the "Territory"):
 - (a) own, manage or control, or become engaged or serve as a shareholder, bondholder, creditor, officer, director, partner, member, employee, agent, consultant, advisor, or representative of any Person, business or enterprise that competes directly or indirectly with the Companies in the Businesses (a "Competitive Business") other than Parent, the Surviving Entity or any of the Companies; provided, however, that each Member and his or its Affiliates may passively hold up to 1% of the outstanding publicly-traded securities of a Person engaged in a Competitive Business for investment purposes only and any amount of any class of securities of Parent;
 - (b) recruit, induce, solicit, or employ, or in any manner attempt to recruit, induce, solicit, or employ, any Person that is at such time, or during the previous two year period was an employee, independent contractor, or consultant of any Company;
 - (c) solicit any Person that is at such time, or during the previous two-year period was, (i) a customer, supplier or business associate of any Company, or (ii) a Person from whom any Company solicited business or with whom any Company discussed a potential business relationship, in each case, for the purpose of offering or providing services or products which are competitive with services or products provided by any Company in the Businesses;
 - (d) cause or seek to cause to be terminated or adversely affected, or otherwise interfere with, any agreement or arrangement of any kind to which any Company is a party or from which it benefits; or
 - (e) seek to interfere with or adversely affect the ongoing relationships between any Company, on the one hand, and its suppliers, customers and professional and business contacts, on the other.

- 5.2.2 Acknowledgement. Each Member acknowledges that the Companies conduct the Businesses throughout the Territory; that they each provide services, directly or indirectly, to the Companies throughout the Territory; and that they each are intimately familiar with the Companies' trade secrets, confidential information, and employee, customer and supplier relationships. Accordingly, each Member agrees that the Territory, time limitation, and scope of restrictions set forth in this Section 5.2 are reasonable and necessary to protect the legitimate business interests of Parent. Each Member agrees that any reduction to the Territory, time limitation, or scope of restrictions set forth herein would seriously undermine the efficacy of this Section 5.2 and the protections that it is intended to provide. Each Member acknowledges and agrees that the covenants contained in this Sections 5.1 and 5.2 are essential elements of this Agreement and that but for these covenants Parent would not have agreed to consummate the Merger. Each Member further expressly agrees and acknowledges that (a) the confidentiality, nonsolicitation and non-competition covenants contained in this Agreement (i) are reasonable and necessary for the protection of Parent with respect to the covenants' respective stated purposes, time, scope and geographic area, and are supported by adequate consideration; (ii) are necessary for the protection of Parent's legitimate business interests, including the protection of trade secrets, goodwill, and relationships with customers and suppliers purchased by Parent pursuant to this Agreement; and (iii) are not unduly restrictive of any rights of the Members; and (b) each Member has sufficient employment or business alternatives and sufficient assets, taking into account the consideration received and to be received by such Member, such that such Member does not have to compete with any Company or impermissibly use any Company's confidential, proprietary, or trade secret information described in this Agreement in order to earn a living. The existence of any claim or cause of action against Parent by any Member, whether predicated on Parent's breach of this Agreement or otherwise, shall not constitute a defense to the enforcement by Parent of the covenants contained in Sections 5.1 and 5.2.
- 5.2.3 <u>Injunction</u>. The Members agree that a violation of the terms, provisions, obligations, duties and conditions described in this Agreement will cause irreparable damage to Parent for which money damages or other legal remedies would not be an adequate remedy. Accordingly, each Member acknowledges and hereby agrees that in the event of any breach or threatened breach by any Member of any of their covenants or obligations set forth in this Sections 5.1 and 5.2, Parent shall be entitled, in any court in the United States or otherwise having jurisdiction, to an injunction or injunctions, without the posting of any bond, to prevent or restrain breaches or threatened breaches of Sections 5.1 and 5.2, and to specifically enforce the terms and provisions of this <u>Sections 5.1</u> and <u>5.2</u> to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of this <u>Sections 5.1</u> and 5.2. Each Member hereby agrees not to challenge the enforceability of the restrictive covenants contained in Sections 5.1 and 5.2 or raise any objections to the availability of the equitable remedy of specific performance to prevent or restrain breaches or threatened breaches of Sections 5.1 and 5.2, and to specifically enforce the terms and provisions of Sections 5.1 and 5.2 to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of each Member under this Sections 5.1 and 5.2. The Members further agree that (x) by seeking the remedies provided for in this Section 5.2.3, Parent shall not in any respect waive its right to seek any other form of relief that may be available under this Agreement (including monetary damages), and (y) nothing set forth in this Section 5.2.3 shall require Parent to institute any Action for (or limit Parent's right to institute any Action for) specific performance under this Section 5.2.3 prior or as a condition to exercising any rights under Section 1.9.2(h), Section 9 or otherwise, nor shall the commencement of any Action pursuant to this Section 5.2.3 or anything set forth in this Section 5.2.3 restrict or limit Parent's right to pursue any other remedies under this Agreement or otherwise that may be available to Parent thereafter. The parties acknowledge and agree that, in the event Parent institutes an Action to enforce the restrictive covenants set forth in this Sections 5.1 and 5.2 in a forum other than one of the forum identified in Section 11.15, Parent may elect to pursue all other remedies (including damages) available to it based on the same set of facts and circumstances in the forum chosen for the injunctive relief Action or it may elect to pursue its damages and other non-injunction remedies in a forum set forth in Section 11.15 and in no event shall Parent's election be deemed a waiver of its rights or remedies or allow any Member to institute an action or assert claims or counterclaims in any forum except as specified in Section 11.15.
- 5.2.4 Severability. If any covenant or restriction contained in Sections 5.1 and 5.2, or any part thereof, is hereafter construed or found to be invalid or unenforceable in part or in whole, this shall not effect the remainder of the covenants or restrictions, which shall be given full effect, without regard to the invalid portions, and any court having jurisdiction shall enforce such invalid covenant or restriction to the maximum extent possible under applicable Law, including the maximum permissible time, scope and geographic area for such covenant or restriction.
- 5.3 Cooperation Regarding Financial Statements. The Members shall use their best reasonable efforts to assist and cooperate with the efforts of Parent and its accountants and auditors to prepare and audit any financial statements (including pro forma financial statements) that Parent will be required to prepare, file or furnish pursuant to any applicable securities laws or exchange requirements, including the 1933 Act, the Exchange Act, the rules and regulations of the SEC promulgated thereunder, and any rules or regulations of the New York Stock Exchange or other stock exchange, or under any other applicable Laws.
- 5.4 <u>Litigation Support</u>. If and for so long as Parent or any Company is actively contesting or defending against any Action in connection with any fact, situation, circumstance, status, condition, activity, practice, plan, occurrence, event, incident, action, failure to act or transaction existing or occurring on or prior to the Closing Date involving any Company or any Business, the Members will cooperate in the contest or defense and provide such testimony as may be reasonably necessary in connection with the contest or defense, at the cost and expense of Parent (unless and to the extent Parent is entitled to indemnification therefor hereunder, in which event such costs and expenses shall be borne jointly and severally by the Members).

5.5 Restrictions on Transfer of Parent Shares.

- (a) The Members will not, and the Indirect Members will cause the Entity Members not to, sell, transfer, pledge, assign, hypothecate or otherwise encumber, dispose of or hedge (whether by the purchase of put options, short selling Parent Shares or otherwise) any Parent Shares or enter into any agreement to do any of the foregoing (the "<u>Transfer Restrictions</u>") at any time from the date hereof until the date on which the Transfer Restrictions on the Parent Shares have lapsed in accordance with the provisions of <u>Subsection 5.5(b)</u>; provided however, that if any Non-Entity Member receives Parent Shares pursuant to this Agreement, such Non-Entity Member may, subject to Parent's prior written consent, which consent may not be unreasonably withheld, transfer such Parent Shares to a Member of the Immediate Family of such Non-Entity Member as a result of his death or in connection with bona fide estate planning purposes.
 - (b) The Transfer Restrictions shall lapse as follows:
 - (i) The Transfer Restrictions with respect to 25% of the Parent Shares issued to each ECS Member shall lapse on the first anniversary of the Closing Date;
 - (ii) The Transfer Restrictions with respect to an additional 25% of the Parent Shares issued to each ECS Member shall lapse on the second anniversary of the Closing Date; and
 - (iii) The Transfer Restrictions with respect to the remaining 50% of the Parent Shares issued to each ECS Member shall lapse on the third anniversary of the Closing Date.
- 5.6 Member Suretyship Obligations. From and after Closing, the Companies shall not, and Parent shall cause the Companies not to, incur or permit to be issued any additional bonds, letters of credit or other obligations that are subject to the Member Suretyship Obligations. For purposes of this Section 5.6, "Member Suretyship Obligation" means any obligation by which a Member is liable, contingently or otherwise, as obligor, guarantor or otherwise, or with respect to which obligation a Member assures a creditor against loss (including contingent reimbursement or indemnification obligations with respect to letters of credit or surety bonds), in each case, to the issuer or co-issuer of any surety bond or letter of credit issued at the request of any of the Companies. If the Member Suretyship Obligations are not released at or prior to Closing, the Companies and Parent will indemnify the Members from any Member liability related to or arising out of a Member Suretyship Obligation to the extent such Member liability results from an action or an event that occurred after the Closing, and following Closing shall use commercially reasonable efforts to have the Member Suretyship Obligation released.

6. CONDITIONS PRECEDENT TO PARENT'S OBLIGATION TO CLOSE

Parent's obligation to consummate the Merger and to take the other actions required to be taken by Parent at the Closing is subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any of which may be waived by Parent, in whole or in part):

6.1 Accuracy of Representations and Warranties. The representations and warranties of ECS and the Members contained in this Agreement that are qualified by a reference to materiality, a Material Adverse Effect or any similar qualifier (any such qualification referred to herein as a "Materiality Qualifier") shall be true and correct in all respects as written (including the Materiality Qualifier) when made and (without giving effect to any schedule updates under Section 4.6) on and as of the Closing as if made at and as of the Closing (other than representations and warranties that are qualified by a reference to a Materiality Qualifier which address matters only as of a certain date, which shall have been true and correct as written (including the Materiality Qualifier) as of such certain date) and the representations and warranties of ECS and the Members set forth in this Agreement that are not qualified by a Materiality Qualifier shall be true and correct in all material respects when made and (without giving effect to any schedule updates under Section 4.6) on and as of the Closing Date as if made on and as of such time (except for those representations and warranties that are not so qualified and relate to a particular date, which representations and warranties shall be true and correct in all material respects as of such date).

6.2 ECS's and Members' Performance.

- 6.2.1 <u>Covenants</u>; <u>Etc.</u> All of the covenants and obligations that ECS and/or Members are required to perform or to comply with pursuant to this Agreement at or prior to the Closing must have been performed and complied with.
 - 6.2.2 <u>Documents, Etc.</u> Each document required to be delivered by ECS and/or Members pursuant to <u>Section 1.3.2</u> must have been delivered.

- 6.3 No Actions. Since the date of this Agreement, there must not have been commenced or threatened against Parent, any Company, the Members or against any Person affiliated with Parent, any Company, or the Members, any Action (a) involving any challenge to, or seeking material damages or other material relief in connection with, any of the Contemplated Transactions, or (b) that may have the effect of preventing, making illegal, or otherwise materially interfering with any of the Contemplated Transactions that in each such case of (a) and (b) has not been fully resolved without additional cost or Liability to Parent, which has not been fully paid by the Members.
- 6.4 No Claim Regarding Units or Merger Consideration. There must not have been made or threatened by any Person any claim asserting that such Person (a) is the holder or the beneficial owner of, or has the right to acquire or to obtain beneficial ownership of, any Unit Equivalent of, or any other Equity Securities in, any Company, or (b) is entitled to all or any portion of the Merger consideration payable by Parent pursuant to this Agreement that in each such case of (a) and (b) has not been fully resolved without additional cost or Liability to Parent, which has not been fully paid by Members.
- 6.5 No Prohibition. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered into any Law or Governmental Order and no other legal or regulatory restraint or prohibition shall be in effect, in either case, which has the effect of making the Contemplated Transactions illegal or that otherwise prohibits or requires the payment of any damages as a result of the Contemplated Transactions, and no Action in which any of the foregoing is sought shall be pending.
- 6.6 No Material Adverse Effect. Since the date of this Agreement, no Material Adverse Effect with respect to ECS or any of the Companies shall have occurred.
- 6.7 <u>HSR Act</u>. Any waiting period (and any extension thereof) under the HSR Act applicable to the Contemplated Transactions shall have expired or been terminated, or approval of the relevant Governmental Authority otherwise obtained.
- 6.8 <u>Third-Party Consents</u>. The Companies shall have received and delivered to Parent the third-party consents listed on <u>Schedule 2.5(a)</u> and <u>Schedule 2.5(b)</u>, without additional cost or Liability to Parent, which has not been fully paid by ECS.
 - 6.9 Merger Option. Parent shall have in its sole discretion exercised the Merger Option.
- 6.10 Employment Agreements. Each of the Required Employees shall have duly executed and delivered to Parent an employment agreement in a form that is satisfactory to Parent;
- 6.11 Non-Competition Agreements. Each of the Required Employees shall have duly executed and delivered to Parent a non-competition agreement substantially in a form that is satisfactory to Parent.
- 6.12 Environmental Review. Parent's satisfaction with its environmental review pursuant to Section 4.9 as determined in Parent's sole and absolute discretion.
- 6.13 Member Affiliate Leases. The Companies shall have entered into new Real Property Leases and terminated the old Real Property Leases for all Real Property leased by the Companies that are owned by an Affiliate of any Member, in each case, in a form that is satisfactory to Parent.

7. CONDITIONS PRECEDENT TO ECS'S AND MEMBERS' OBLIGATIONS TO CLOSE

ECS's and Members' obligations to consummate the Merger and to take the other actions required to be taken by ECS and Members at the Closing are subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any of which may be waived by the Members' Representative, in whole or in part):

7.1 Accuracy of Representations and Warranties. The representations and warranties of Parent contained in this Agreement that are qualified by a reference to a Materiality Qualifier shall be true and correct in all respects when made as written (including the Materiality Qualifier) and (without giving effect to any schedule updates) on and as of the Closing as if made at and as of the Closing (other than representations and warranties that are qualified by a reference to a Materiality Qualifier which address matters only as of a certain date, which shall have been true and correct as written (including the Materiality Qualifier) as of such certain date), and the representations and warranties of Parent set forth in this Agreement that are not qualified by a Materiality Qualifier shall be true and correct in all material respects when made and (without giving effect to any schedule updates) on and as of the Closing Date as if made on and as of such time (except for those representations and warranties that are not so qualified and relate to a particular date, which representations and warranties shall be true and correct in all material respects as of such date).

7.2 Parent's Performance.

- 7.2.1 Covenants; Etc. All of the covenants and obligations that Parent is required to perform or to comply with pursuant to this Agreement at or prior to the Closing must have been performed and complied with.
 - 7.2.2 Documents, Etc. Parent must have delivered each of the documents required to be delivered by Parent pursuant to Section 1.3.2(b).
- 7.3 No Prohibition. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered into any Law or Governmental Order and no other legal or regulatory restraint or prohibition shall be in effect, in either case, which has the effect of making the Contemplated Transactions illegal or that otherwise prohibits or requires the payment of any damages as a result of the Contemplated Transactions, and no Action in which any of the foregoing is sought shall be pending.
- 7.4 HSR Act. Any waiting period (and any extension thereof) under the HSR Act applicable to the Contemplated Transactions shall have expired or been terminated, or approval of the relevant Governmental Authority otherwise obtained.

8. TERMINATION

- 8.1 Termination Events. This Agreement may, by written notice given prior to or at the Closing, be terminated:
 - 8.1.1 By mutual written consent of Parent and the Members' Representative;
- 8.1.2 At any time after the two year anniversary of the date of this Agreement, by either the Members' Representative or Parent if Parent has not exercised the Merger Option by such two year anniversary; the right to terminate this Agreement pursuant to this <u>Section 8.1.2</u> may not be exercised by the Members' Representative if ECS or any Member is then in breach of any representation, warranty, covenant, agreement or obligation contained in this Agreement;
 - 8.1.3 At any time by Parent, upon written notice to the Members' Representative, if it determines never to exercise the Merger Option; or
- 8.1.4 At any time after the sixth month after Parent exercises the Merger Option (the "Outside Date") by Parent or Members' Representative if the Closing has not occurred by the Outside Date; the right to terminate this Agreement pursuant to this Section 8.1.4 may not be exercised by the Members' Representative if ECS or any Member is then in breach of any representation, warranty, covenant, agreement or obligation contained in this Agreement
- 8.2 <u>Effect of Termination</u>. If this Agreement is terminated pursuant to <u>Section 8.1</u>, this Agreement shall forthwith become void and of no further force and effect, except for the provisions of this <u>Section 8.2</u>, <u>Section 11</u> and <u>Section 12</u>, which shall survive such termination; <u>provided</u>, <u>however</u>, that nothing herein shall relieve any party from Liability for any breach of any representation, warranty, covenant or agreement contained in this Agreement.

9. INDEMNIFICATION

- 9.1 <u>Indemnification by the Members</u>. Subject to the limitations set forth in this <u>Section 9</u>, from and after Closing, the Members, jointly and severally, will indemnify and hold harmless Parent and each of its Affiliates, and the Representatives and Affiliates of each of the foregoing Persons (each, a "<u>Parent Indemnified Person</u>"), from, against and in respect of any and all Losses whether or not involving a Third Party Claim, incurred or suffered by Parent Indemnified Persons or any of them as a result of, arising out of, or directly or indirectly relating to:
- 9.1.1 any breach of, or inaccuracy in, any representation or warranty made by ECS or Members in Section 2, in each case, when made and (giving effect to any schedule updates under Section 4.6) on and as of the Closing as if made at and as of the Closing (other than representations and warranties that are qualified by a reference to a certain date, which shall have been true and correct as of such certain date);
- 9.1.2 any breach or violation of any covenant or agreement of ECS to the extent required to be performed or complied with by ECS or any Company in or pursuant to this Agreement;
- 9.1.3 any breach or violation of any covenant or agreement of any Member to the extent required to be performed or complied with by any Member in or pursuant to this Agreement;

- 9.1.4 any Liabilities (other than Liabilities under the Contractual Obligations of the Companies, except for Liabilities of the Companies or Members of any kind or nature (i) the existence of which would conflict with or constitute a breach of a representation, warranty or agreement of any of the Members or Companies contained herein or (ii) by reason of or arising out of any default or breach occurring prior to the Closing Date (or that would be a default or breach but for the passage of time or giving of notice, or both) or any unperformed, warranty repair obligation of the Companies that exists prior to the Closing Date under such Contractual Obligations but that is unperformed as of the Closing Date, whether or not such Contractual Obligation is disclosed on the Disclosure Schedules) of any of the Companies or the Members, of any nature whatsoever, whether accrued, absolute, contingent or otherwise, arising from or relating to the operations or activities of the Companies or Members prior to the Closing Date, except to the extent any such liability is reflected in full as a Liability in the Actual Closing Balance Sheet;
- 9.1.5 except to the extent any such Liability (other than Liabilities under the Contractual Obligations of the Companies, except for Liabilities of the Companies or Members of any kind or nature (i) the existence of which would conflict with or constitute a breach of a representation, warranty or agreement of any of the Members or Companies contained herein or (ii) by reason of or arising out of any default or breach occurring prior to the Closing Date (or that would be a default or breach but for the passage of time or giving of notice, or both) or any unperformed warranty repair obligation of the Companies that exists prior to the Closing Date under such Contractual Obligations but that is unperformed as of the Closing Date, whether or not such Contractual Obligation is disclosed on the Disclosure Schedules) is reflected as a Liability in the Actual Closing Balance Sheet, any Disclosed Liabilities;
- 9.1.6 any Action by any Person not a party to this Agreement (including any Governmental Authority) against or affecting Parent or any other Parent Indemnified Person which may give rise to or evidence the existence of or relate to a misrepresentation or breach of any representation or warranty contained in Section 2;
 - 9.1.7 the matters set forth on Exhibit G; or
- 9.1.8 any breach of, or inaccuracy in, any representation or warranty made by ECS or Members in Section 2.17(i) when made and on and as of the Closing as if made at and as of the Closing, without regard and without giving effect to any Knowledge Qualifier contained in such representation or warranty which has the effect of making such representation and warranty less likely to be breached (as if such word or words were deleted from such representation and warranty).

Notwithstanding anything contained herein to the contrary, for purposes of determining whether there has been a breach and the amount of any Losses that are the subject matter of a claim for indemnification or reimbursement hereunder, each representation and warranty in this Agreement and schedules and exhibits hereto shall be read without regard and without giving effect to any Materiality Qualifier contained in such representation or warranty which has the effect of making such representation and warranty less likely to be breached (as if such word or words were deleted from such representation and warranty).

- 9.2 <u>Indemnity by Parent</u>. Subject to the limitations set forth in this <u>Section 9</u>, from and after Closing, Parent will indemnify and hold harmless the Members and their Affiliates, and the Representatives and Affiliates of each of the foregoing Persons (each, a "<u>Company Indemnified Person</u>"), from, against and in respect of any and all Losses incurred or suffered by Company Indemnified Persons or any of them as a result of, arising out of, or directly or indirectly relating to:
 - 9.2.1 any breach of, or inaccuracy in, any representation or warranty made by Parent in Section 3; or
 - 9.2.2 any breach or violation of any covenant or agreement of Parent pursuant to this Agreement.
- 9.3 <u>Time for Claims</u>. No claim may be made or suit instituted seeking indemnification pursuant to <u>Section 9.1.1, 9.1.6 or 9.2.1</u> for any breach of, or inaccuracy in, any representation or warranty unless a written notice describing such breach or inaccuracy in reasonable detail in light of the circumstances then known to the Indemnified Person, is provided to the Indemnifying Person:
- 9.3.1 at any time, in the case of any breach of, or inaccuracy in, the representations and warranties set forth in Sections 2.1 (Organization; Predecessors; Subsidiaries), 2.2 (Capitalization of the Companies), 2.3 (Organization; Power and Authorization), 2.5 (Noncontravention), the penultimate sentence of 2.6.1 (Financial Statements), 2.9.1 (Ownership of Assets), 2.30 (No Brokers), the last sentence of 2.25.6 (Employees), 3.1 (Organization), 3.2 (Power and Authorization), 3.4 (Noncontravention) or 3.5 (No Brokers);
 - 9.3.2 at any time, in the case of any claim or suit based upon fraud, intentional misrepresentation or willful misconduct;
- 9.3.3 at any time prior to the ninetieth (90 $^{\rm th}$) day after the expiration of the applicable statute of limitations (taking into account any tolling periods and other extensions) in the case of any breach of, or inaccuracy in, the representations and warranties set forth in Sections 2.17 (Employee Benefit Plans), or 2.18 (Environmental Matters); and

9.3.4 at any time prior to last day of the eighteenth (18 th) month following the Closing Date, in the case of any breach of, or inaccuracy in, any other representation and warranty or any Loss arising from Liabilities described in Section 9.1.1, 9.1.6 or 9.2.1; provided that if, at any time prior to the expiration of the respective survival period set forth in this Section 9.3 with respect to any particular representation or warranty, any Indemnified Party delivers to any Indemnifying Party a written notice alleging the existence of an inaccuracy in or a breach of such representation or warranty and asserting a claim for Losses pursuant to Section 9.1 or Section 9.2, then the representation or warranty underlying the claim asserted in such notice and all indemnity obligations under this Section 9 related thereto shall survive until such claim is finally and fully resolved in accordance with this Agreement.

Claims for indemnification pursuant to any other provision of <u>Sections 9.1</u> and <u>9.2</u> or pursuant to <u>Section 10</u> are not subject to the limitations set forth in this Section 9.3.

9.4 Indemnification Limits.

9.4.1 The Members shall not be liable for any indemnification obligations under this Agreement with respect to, or to hold any Parent Indemnified Persons harmless from, and Parent shall not be liable for any indemnification obligations under this Agreement with respect to, or to hold any Company Indemnified Person harmless from, any claim or group of related claims, unless such claim or group of related claims involves Losses in excess of \$10,000 (all such claims or group of related claims of an amount equal to \$10,000 or less, a "De Minimis Claim"), nor shall any De Minimis Claim be applied to or considered for purposes of calculating the aggregate amount of the Parent Indemnified Persons' or Company Indemnified Persons' Losses, as applicable; provided that the limitations provided in this sentence shall only apply with respect to an aggregate of up to \$100,000 of De Minimis Claims and if all De Minimis Claims in the aggregate exceed \$100,000 (such excess, "Excess De Minimis Claims"), the Members and Parent shall, as applicable, subject to the other limitations provided in this Agreement, indemnify and hold the Parent Indemnified Persons or Company Indemnified Persons, as applicable, harmless from the amount of such excess over \$100,000 and the amount of such excess shall be applied to and considered for purposes of calculating the aggregate amount of the Parent Indemnified Persons' or Company Indemnified Persons' Losses, as applicable, needed to meet the Deductible and the Liability Cap. In addition, the Members shall not be liable for the indemnification obligations pursuant to Section 9.1.1, Section 9.1.4, Section 9.1.5 or Section 9.1.6, or to hold the Parent Indemnified Persons harmless, and Parent shall not be liable for the indemnification obligations pursuant to Section 9.2.1, or to hold the Company Indemnified Persons harmless, until the aggregate amount of Losses, including Losses with respect to Excess De Minimis Claims, with respect to matters referred to, in the case of the Members, Section 9.1.1, Section 9.1.4, Section 9.1.5 and Section 9.1.6 or, in the case of the Parent, Section 9.2.1 exceeds \$600,000 excluding in calculating such amount any De Minimis Claims but not Excess De Minimis Claims (the "Deductible"), after which Parent, on the one hand, and the Members jointly and severally, on the other hand, will be responsible for all Losses, including Losses with respect to Excess De Minimis Claims, in excess of the Deductible up to a maximum aggregate amount of Losses equal to Thirteen Million Six Hundred and Fifty Dollars (\$13,650,000) (the "Liability Cap"). Notwithstanding anything herein to the contrary, (A) none of the De Minimis Claim limitation, the Deductible or the Liability Cap will apply (i) in the case of fraud, intentional misrepresentation or willful misconduct or (ii) with respect to Members' indemnification obligations pursuant to Section 9.1.2, (B) the Deductible or the Liability Cap will not apply with respect to the Members' indemnification obligations pursuant to Section 9.1.7, and (C) the Liability Cap will not apply (i) with respect to any breach of Sections 2.1 (Organization; Predecessors; Subsidiaries), 2.2 (Capitalization of the Companies), 2.3 (Organization; Power and Authorization), 2.5 (Noncontravention), the penultimate sentence of 2.6.1 (Financial Statements), 2.9.1 (Ownership of Assets), 2.17 (Employee Benefit Plans), 2.18 (Environmental Matters), the last sentence of 2.25.6 (Employees), 2.30 (No Brokers), 3.1 (Organization), 3.2 (Power and Authorization), 3.4 (Noncontravention) or 3.5 (No Brokers), or (ii) with respect to Members' indemnification obligations pursuant to Section 10 or Section 9.1.8.

9.4.2 Notwithstanding anything to the contrary herein, except for claims based on fraud, intentional misrepresentation or willful misconduct:
(i) Strauss shall have no liability for any claim that may be made or suit instituted seeking indemnification pursuant to item 1 of Exhibit G, and (ii) Strauss and Campbell shall have no liability for any claim that may be made or suit instituted seeking indemnification pursuant to Section 9.1.3 except to the extent that such claim relates to a breach or a violation of a covenant or agreement of such Member and then solely with respect to such Member's breach or violation.

9.5 Third Party Claims.

- 9.5.1 Notice of Claim. If any third party notifies an Indemnified Person with respect to any matter (a "Third Party Claim") which may give rise to an Indemnity Claim against an Indemnifying Person under this Section 9, then the Indemnified Person will promptly give written notice to the Indemnifying Person; provided, however, that no delay on the part of the Indemnified Person in notifying the Indemnifying Person will relieve the Indemnifying Person from any obligation under this Section 9 except to the extent that the Indemnifying Person is actually prejudiced by the Indemnified Party's failure to give such notice in such a timely manner.
- 9.5.2 <u>Assumption of Defense, etc.</u> The Indemnifying Person will be entitled to participate in the defense of any Third Party Claim that is the subject of a notice given by the Indemnified Person pursuant to <u>Section 9.5.1</u>. In addition, the Indemnifying Person will have the right to assume the defense of the Indemnified Person against the Third Party Claim with counsel with counsel reasonably satisfactory (taking into account the jurisdiction in which the claim is brought, the materiality of the claim and the experience levels and other qualifications of counsel available to defend such claim) to the Indemnified Person so long as (a) the Indemnifying Person

gives written notice to the Indemnified Person within ten (10) days after the Indemnified Person has given notice of the Third Party Claim that the Indemnifying Person will indemnify the Indemnified Person from and against the entirety of any and all Losses the Indemnified Person may suffer resulting from, arising out of, relating to, in the nature of, or caused by the Third Party Claim, (b) the Indemnifying Person provides the Indemnified Person with evidence reasonably acceptable to the Indemnified Person that the Indemnifying Person will have adequate financial resources to defend against the Third Party Claim and fulfill its indemnification obligations hereunder, (c) the Third Party Claim involves only money damages and does not seek an injunction or other equitable relief against the Indemnified Person, (d) the Indemnified Person has not been advised by counsel that an actual or potential conflict exists between the Indemnified Person and the Indemnifying Person in connection with the defense of the Third Party Claim, (e) the Third Party Claim does not relate to or otherwise arise in connection with Taxes or any criminal or regulatory enforcement Action and (f) the Indemnifying Person conducts the defense of the Third Party Claim actively and diligently. The Indemnified Person may retain separate co-counsel at its sole cost and expense and participate in the defense of the Third Party Claim; provided, however, that the Indemnifying Person will pay the fees and expenses of separate co-counsel retained by the Indemnified Person that are incurred prior to the Indemnifying Person's assumption of control of the defense of the Third Party Claim.

- 9.5.3 <u>Limitations on Indemnifying Person</u>. The Indemnifying Person will not consent to the entry of any judgment or enter into any compromise or settlement with respect to the Third Party Claim without the prior written consent of the Indemnified Person, which consent will not be unreasonably withheld, conditioned or delayed, unless such judgment, compromise or settlement (a) provides for the payment by the Indemnifying Person of money as sole relief for the claimant, (b) results in the full and general release of all Parent Indemnified Persons or Company Indemnified Persons, as applicable, from all Liabilities arising or relating to, or in connection with, the Third Party Claim and (c) involves no finding or admission of any violation of Legal Requirements or the rights of any Person and has no effect on any other claims that may be made against the Indemnified Person.
- 9.5.4 <u>Indemnified Person's Control.</u> If the Indemnifying Person does not deliver the notice contemplated by clause (a), or the evidence contemplated by clause (b), of <u>Section 9.5.2</u> within 10 (ten) days after the Indemnified Person has given notice of the Third Party Claim, or otherwise at any time fails to conduct the defense of the Third Party Claim actively and diligently, the Indemnified Person may defend, and may consent to the entry of any judgment or enter into any compromise or settlement with respect to, the Third Party Claim with the consent of the Indemnifying Person, which consent shall not be unreasonably delayed, conditioned or withheld. If such notice and evidence is given on a timely basis and the Indemnifying Person conducts the defense of the Third Party Claim actively and diligently but any of the other conditions in <u>Section 9.5.2</u> is or becomes unsatisfied, the Indemnified Person may defend, and may consent to the entry of any judgment or enter into any compromise or settlement with respect to, the Third Party Claim with the consent of the Indemnifying Person, which consent shall not be unreasonably delayed, conditioned or withheld. In the event that the Indemnified Person conducts the defense of the Third Party Claim pursuant to this <u>Section 9.5.4</u>, the Indemnifying Person will (a) advance the Indemnified Person promptly and periodically for the costs of defending against the Third Party Claim (including reasonable attorneys' fees and expenses) and (b) remain responsible for any and all other Losses that the Indemnified Person may incur or suffer resulting from, arising out of, relating to, in the nature of or caused by the Third Party Claim to the fullest extent provided in this <u>Section 9</u>.
- 9.5.5 <u>Reasonable Cooperation</u>. The party not in control of the prosecution or defense of a Third Party Claim will reasonably cooperate with the other party in the conduct of the prosecution or defense of such Third Party Claim.
- 9.6 <u>Limitations on Liability</u>. If a breach of the representations, warranties or covenants given by the Members in this Agreement is capable of remedy without Loss to Parent or the Parent Indemnified Persons, the Parent Indemnified Persons shall only be entitled to indemnification if the breach is not remedied within thirty (30) days after the date on which notice is served on the Members' Representative.
- 9.7 Exclusive Remedy. Parent acknowledges and agrees that, should the Closing occur, except with respect to claims based on fraud, intentional misrepresentation or willful misconduct, Parent Indemnified Persons' sole and exclusive remedy with respect to any and all claims relating to this Agreement, whether based on contract, tort, strict liability, other laws or otherwise, including any breach or alleged breach of any representation, warranty, covenant or agreement shall be pursuant to the indemnification provisions set forth in this Section 9 and Section 10 and the provisions set forth in Section 5.2.3 and Section 11.18, and hereby waives, for itself and on behalf of the Parent Indemnified Persons, from and after the Closing, to the fullest extent permitted under applicable Law, any and all other rights, claims and causes of action (other than rights, claims, or causes of action arising from fraud, intentional misrepresentation or willful misconduct) it may have against Members arising under or based upon any Law or otherwise relating to this Agreement. The Members acknowledge and agree that, should the Closing occur, except with respect to claims based on fraud, intentional misrepresentation or willful misconduct, Company Indemnified Persons' sole and exclusive remedy with respect to any and all claims relating to this Agreement, whether based on contract, tort, strict liability, other laws or otherwise, including any breach or alleged breach of any representation, warranty, covenant or agreement shall be pursuant to the indemnification provisions set forth in this Section 9, and hereby waive, for itself or himself and on behalf of the Company Indemnified Persons, from and after the Closing, to the fullest extent permitted under applicable Law, any and all other rights, claims and causes of action (other than rights, claims, or causes of action arising from fraud, intentional misrepresentation or willful misconduct) it or he may have against Parent arising under or based upon any Law or otherwise relating to

- 9.8 Members' Representative. For purposes of this Section 9, it is understood and agreed that the Members' Representative shall act on behalf of all Members.
- 9.9 <u>Knowledge and Investigation</u>. The right of any Parent Indemnified Person or Company Indemnified Person to indemnification pursuant to this <u>Section 9</u> will not be affected by any investigation conducted or knowledge acquired (or capable of being acquired) at any time, whether before or after the execution and delivery of this Agreement or the Closing, with respect to the accuracy of any representation or warranty, performance of or compliance with any covenant or agreement, referred to in <u>Sections 9.1</u> or <u>9.2</u>, or otherwise.
- 9.10 <u>Remedies Cumulative</u>. The rights of each Parent Indemnified Person and Company Indemnified Person under this <u>Section 9</u> are cumulative, and each Parent Indemnified Person and Company Indemnified Person, as the case may be, will have the right in any particular circumstance, in its sole discretion, to enforce any provision of this <u>Section 9</u> without regard to the availability of a remedy under any other provision of this <u>Section 9</u>.
- 9.11 Right of Setoff. Upon notice to the Members specifying in reasonable detail the basis therefore, Parent may set off any amount to which it may be entitled from the Members against amounts otherwise payable, if any, under this Agreement. If such setoff is made against Parent Shares, such shares shall be valued on the day immediately prior to the day such shares were due to be issued, absent the setoff. The exercise of such right of setoff by Parent in good faith, whether or not ultimately determined to be justified, will not constitute an event of default under this Agreement or any other agreement between Parent or any of its Affiliates and the Companies or Members. Neither the exercise or failure to exercise such right of setoff will constitute an election of remedies or limit Parent in any manner in the enforcement of any other remedies that may be available to it. Notwithstanding the foregoing, if Parent sets off an amount pursuant to this Section 9.11 and it is later finally mutually determined by the parties or pursuant to Section 11.15 that such setoff amount was in excess of the amount which any Member was required to indemnify Parent, then Parent shall pay interest to such Member on such excess amount at the Prime Rate as published in the Wall Street Journal on the date such setoff was made.
- 9.12 ECS's Indemnification Obligations. The obligation of ECS to indemnify Parent and the other Parent Indemnified Persons shall terminate in all respects upon the Closing; provided that for the avoidance of doubt the parties agree that the termination of ECS's indemnification obligations pursuant to this Section 9.12 shall have no effect on or in any way limit Parent's or any other Parent Indemnified Person's rights to indemnification from the Members, including Parent's and the other Parent Indemnified Persons' rights to offset Losses that are incurred by or imposed on Parent or any other Parent Indemnified Person against any amounts otherwise payable pursuant to this Agreement. In addition, the Members' rights to seek contribution or payment of any amount from any Company for any indemnification obligations that they are required to make to Parent or any other Parent Indemnified Person shall also terminate in all respects upon the Closing. If any Company suffers, incurs or otherwise becomes subject to any Losses as a result of or in connection with any inaccuracy in or breach of any representation, warranty, covenant or obligation of the Members hereunder, then (without limiting any of the rights of any Company, Parent or any other Parent Indemnified Persons as Indemnified Persons) Parent shall also be deemed, by virtue of its ownership of the Equity Securities of the Companies, to have incurred Losses as a result of and in connection with such inaccuracy or breach but in any case the total amount both Parent and the Companies may recover shall not exceed the amount of Losses.
- 9.13 <u>Time of Payment of Claims</u>. Except as otherwise set forth in this <u>Section 9</u>, any amount owing by any Person pursuant to this <u>Section 9</u> shall be paid within two (2) Business Days after determination of such amount. To the extent any ECS Member holds Parent Shares, such ECS Member may satisfy its indemnification obligations pursuant to <u>Section 9.1.1</u> or <u>Section 10</u> by assignment of Parent Shares at an agreed value of \$13 per share; <u>provided</u>, <u>however</u>, that if Parent issues Parent Shares in connection with any Earn-Out Payment then the "agreed value" for purposes of valuing Parent Shares pursuant to this <u>Section 9.13</u> shall be the volume weighted average price of all Parent Shares issued to such ECS Member.
- 9.14 <u>Closing Balance Sheet</u>. Notwithstanding anything to the contrary set forth in this Agreement, no Parent Indemnified Person shall be entitled to indemnification pursuant to <u>Section 9.1.1</u> or <u>Section 10</u> for any Loss or Taxes or alleged Loss or Taxes to the extent of any Liability, accrual or reserve therefor is included as a current Liability in the Actual Closing Balance Sheet.

10. TAX MATTERS

- 10.1 <u>Representations and Obligations Regarding Taxes.</u> ECS and Members jointly and severally represent and warrant to and agree with Parent as follows, in each case except to the extent set forth on <u>Schedule 10</u> (for purposes of this <u>Section 10</u>, the "<u>Companies</u>" includes any Predecessor and any entity that at any time has been a subsidiary of a Company, and any of the Companies may be referred to individually as a "<u>Company</u>"):
- 10.1.1 Except as set forth on Schedule 10.1.1: (A) each Company has duly and timely filed all Tax Returns it was required to file; (B) all such Tax Returns were true, correct and complete and all Taxes of such Company (whether or not shown on any Tax Return and whether or not any Tax Return was required) have been paid; (C) no Company is the beneficiary of any extension of time within which to file any Tax Return; (D) the unpaid Taxes of each Company did not, as of the dates of the Financials, exceed the reserve for Tax liability (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth

on the face of the balance sheets (rather than in any notes thereto) contained in the Financials; (E) since the date of the Most Recent Balance Sheet Date, none of the Companies has incurred any liability for Taxes outside the Ordinary Course of Business; (F) as of the Closing Date, the unpaid Taxes of the Companies will not exceed the reserve for Tax liability (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the face of the books and records of the Companies (rather than in any notes thereto); (G) no claim has ever been made by a Governmental Authority in a jurisdiction where a Company does not currently file Tax Returns that such Company is or may be subject to taxation by that jurisdiction; and (h) there are no Encumbrances on any of the assets of any Company that arose in connection with any failure (or alleged failure) to pay any Tax, except for Encumbrances for Taxes not yet due.

- 10.1.2 Except as set forth on Schedule 10.1.2, there is no dispute or claim concerning any Tax Liability of any Company either (A) claimed or raised by any authority in writing or (B) to ECS's Knowledge based upon personal contact with any agent of such authority. No Company has received from any Governmental Authority any written notice of proposed adjustment, deficiency, underpayment of Taxes or any other such notice which has not been satisfied by payment or been withdrawn, and no claims have been asserted relating to such Taxes against any Company. There is no dispute or claim concerning any Tax Liability of any Company either claimed or raised by any Governmental Authority in writing.
- 10.1.3 Except as set forth on Schedule 10.1.3, no Tax Return has been audited, or is currently the subject of audit. ECS has made available in the Data Room correct and complete copies of all of the Companies' federal and foreign Tax Returns, examination reports, and statements of deficiencies assessed against, or agreed to by any Company since its formation. No Company has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to any Tax assessment or deficiency.
- 10.1.4 Except as set forth on Schedule 10.1.4, no Company is a party to any joint venture, partnership or other arrangement or contract that could be treated as a partnership for Federal income tax purposes. No Company has entered into any sale leaseback or leveraged lease transaction that fails to satisfy the requirements of Revenue Procedure 2001-28 (or similar provisions of foreign Law) or any safe harbor lease transaction. No Company has acquired nor does it own any assets that directly or indirectly secure any debt the interest on which is tax exempt under Section 103 of the Code.
- 10.1.5 ECS has since its formation been an eligible entity as determined in Section 301.7701-3(a) of the Treasury Regulations. ECS elected to be treated as a corporation for federal, state and local income tax purposes commencing January 1, 2008. The Companies are members of an Affiliated Group as defined in Section 1504 of the Code of which ECS is the common parent.
 - 10.1.6 No Company has engaged in any "reportable transaction" as defined in Section 1.6011-4 of the Treasury Regulations.
- 10.1.7 No Company has received from any Governmental Authority any written notice of proposed adjustment, deficiency, underpayment of Taxes or any other such notice which has not been satisfied by payment or been withdrawn, and no claims have been asserted relating to such Taxes against any Company.
- 10.1.8 Each Company has withheld and paid all Taxes required to have been withheld and paid in connection with amounts, allocable, paid or owing to any employee, independent contractor, creditor, member, partner or other third party. No Company has a non-accountable expense reimbursement arrangement within the meaning of Treasury Regulation Section 1.62-2(c).
- 10.1.9 No Company has been a member of an affiliated group filing a consolidated federal income Tax Return (other than a group the common parent of which is ECS). No Company has any liability for the Taxes of any Person (other than Taxes of the Companies under Treasury regulation Section 1.1502-6 (or any similar provision of state, local or foreign law)).
- 10.1.10 There is no contract, agreement plan or arrangement covering any employee or former employee or independent contractor or former independent contractor of any Company that, individually or collectively, could give rise to the payment by such Company of any amount that would not be deductible by reason of Section 162(a)(1) or Section 280G of the Code.
- 10.1.11 Each Company has paid all estimated Taxes for all Tax periods which it is required to have paid estimated Taxes representing 100% of the estimated Taxes required to have been paid.
- 10.1.12 No Company has or had a permanent establishment in any foreign country or engages of has engaged in a trade or business in any foreign country.
- 10.1.13 No Company has had any Indebtedness that (i) was "corporate acquisition indebtedness" as defined in Section 279 of the Code; (ii) bore interest any portion of which was "disqualified interest" as defined in Section 163(j)(3) of the Code, or (iii) was an "applicable high yield discount obligation" as defined in Section 168(i)(1) of the Code. No Company has deducted any amounts that are required to be capitalized.

- 10.1.14 No Company has distributed the stock of any corporation in a transaction satisfying the requirements of Section 355 of the Code within the last five (5) years, and no stock of a Company has been distributed in a transaction satisfying the requirements of Section 355 of the Code within the last five (5) years.
- 10.1.15 No Company will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion of any taxable period) after the Closing Date as a result of any (i) closing agreement as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or non-U.S. Tax law); (ii) installment sale or open transaction disposition occurring on or prior to the Closing Date; (iii) cash basis method of accounting or percentage of completion method of accounting; (iv) an election under Section 108(i) of the Code; or (v) prepaid amount received on or prior to the Closing Date.
 - 10.2 Indemnification for Taxes. Notwithstanding anything to the contrary in this Agreement (in particular Section 9):
- 10.2.1 Members shall jointly and severally indemnify Parent and its Affiliates (each herein sometimes referred to as an "<u>Indemnified Taxpayer</u>") against, and agree to jointly and severally protect, save and hold harmless each Indemnified Taxpayer from, any and all Losses, resulting from:
 - (a) a claim by any Governmental Authority for any Taxes arising from or occasioned by the Merger or the other Contemplated Transactions:
 - (b) any misrepresentation or breach of any representation, warranty or obligation set forth in this Section 10;
 - (c) any liability for the Taxes of any Company for any period ending on or before the Closing Date and the portion of any Straddle Period ending on the Closing Date; and
 - (d) all liabilities of the Companies as a result of the applicability of Treas. Reg. §1.1502-6 or similar provisions of foreign, state or local Tax law for Taxes of any other corporation affiliated with any Company on or prior to the Closing Date.
- 10.2.2 Upon notice from Parent to the Members' Representative that an Indemnified Taxpayer is entitled to an indemnification payment for a Loss pursuant to Section 10.2.1, the Members thereupon shall jointly and severally pay to the Indemnified Taxpayer an amount that, net of any Taxes imposed on the Indemnified Taxpayer with respect to the payment, will indemnify and hold the Indemnified Taxpayer harmless from the Loss.
- 10.2.3 Notwithstanding anything to the contrary in this Agreement, the indemnification obligations of the Members under this Section 10 shall survive the Closing until the 90 th day following the end of the applicable statutes of limitations.
- 10.2.4 All transfer, documentary, sales, use, stamp, registration and other similar Taxes and fees (including any penalties and interest) incurred in connection with this Agreement shall be jointly and severally paid by Members when due, and Members will, at their expense, file all necessary Tax Returns and other documentation with respect to all the transfer, documentary, sales, use, stamp, registration and other Taxes and fees, and, if required by applicable Law, Parent will, and will cause its Affiliates to, join in the execution of any of those Tax Returns and other documentation.

10.3 Covenants With Respect To Taxes.

- 10.3.1 Parent shall (i) grant to the Members' Representative reasonable access to the Companies' books and records (including tax workpapers and returns and correspondence with tax authorities), including the right to take extracts therefrom and make copies thereof, to the extent that the books and records relate to the operations of the Companies during taxable periods ending on or prior to or that include the Closing Date as may be necessary for Tax Return preparation, audit, or other financial reporting matters, and (ii) otherwise reasonably cooperate with the Members' Representative in connection with the preparation and filing of any Tax Return of the Company that the Members are responsible for and with any audit of Taxes that relates to the Businesses of the Companies prior to the Closing Date, in each case to the extent relevant to the Members' indemnification obligations under Section 10.2.
- 10.3.2 Parent shall be responsible for preparing and filing, or causing the Companies to prepare and file, all Tax Returns of the Companies required to be filed for taxable periods ending after the Closing Date and the Members shall be responsible, at their cost, for filing all Tax Returns of the Companies for taxable periods ending on or before the Closing Date ("Member-Prepared Returns"). The Member shall furnish such Member-Prepared Returns to the Parent for its review, comment and approval (such approval not to be unreasonably withheld, conditioned or delayed) at least 30 days prior to the due date (or extended due date) for filing such Tax Returns, and such Tax Returns shall be prepared in accordance with past practice, except as required by Law. The Members shall pay all Taxes required to be paid with respect to Member-Prepared Returns, except to the extent such Taxes have been previously paid, deposited

or accrued in Actual Net Working Capital on the Actual Closing Balance Sheet. In the case of any Tax Returns other than income Tax Returns (a "Straddle Return") for taxable periods which begin before the Closing Date and end after the Closing Date (a "Straddle Period"), Parent shall furnish such Tax Returns to the Member's Representative for its review, comment and approval (such approval not to be unreasonably withheld, conditioned or delayed) at least 30 days prior to the due date (or extended due date) for filing such Tax Returns, and such Tax Returns shall be prepared in accordance with past practice, except as required by Law. The Members shall jointly and severally pay to Parent within five (5) days after the date on which Taxes are paid with respect to a Straddle Return approved by the Members' Representative (or as to which the refusal of Members' Representative to approve is unreasonable) an amount equal to the portion of those Taxes that relates to the portion of the taxable period ending on the Closing Date as determined pursuant to Section 10.3.3, reduced to the extent such Taxes have been previously paid, deposited or accrued in Actual Net Working Capital on the Actual Closing Balance Sheet.

- 10.3.3 For purposes of this Agreement, other than in the case of income Tax Returns, in the case of any period that begins before the Closing Date and ends after the Closing Date, any Tax based directly or indirectly on gross or net income or receipts or imposed in respect of specific transactions and any credits available with respect to any Tax, shall be allocated based on an interim closing of the books by assuming that the taxable period ended on the Closing Date, and any other Tax shall be allocated based on the number of days in the taxable period ending on the Closing Date divided by the total number of days in the taxable period.
- 10.3.4 Neither Parent nor its Affiliates (including the Companies) may amend or cause the amendment of a Tax Return of the Companies, or file or amend any Tax election concerning the Companies, with respect to any period ending on or prior to the Closing Date or any period that begins before and ends after the Closing Date, if such action could have the effect of increasing the Tax liability or reducing any Tax benefit of the Sellers or any Affiliate of the Sellers, without the written consent of the Members' Representative, which consent shall not be unreasonably withheld, conditioned or delayed, except if such amendment is required by any Governmental Authority.
- 10.3.5 Except to the extent reflected as an asset in the Actual Closing Balance Sheet or to the extent resulting from the carryback of a net operating loss or other Tax attribute from a taxable period (or portion thereof) that begins on or after the Closing Date, any refund of Taxes of the Companies (including any interest with respect thereto) attributable to any period occurring on or before the Closing Date, or as allocated pursuant to Section 10.3.3, shall be the property of the Members, shall be paid promptly to the Members' Representative and, if received by Parent or its Affiliate (including the Companies), shall be payable promptly to the Members' Representative.
- 10.3.6 Parent shall not permit, and shall cause its Affiliates not to permit, the Companies to take any actions on the Closing Date that are out of the ordinary course of business of either Company, except as required or contemplated by this Agreement.

11. MISCELLANEOUS

- 11.1 Notices. All notices, requests, demands, claims and other communications required or permitted to be delivered, given or otherwise provided under this Agreement must be in writing and must be delivered, given or otherwise provided:
 - (a) by hand (in which case, it will be effective upon delivery);
 - (b) by facsimile followed by delivery by first class U.S. mail, postage prepaid (in which case, it will be effective upon receipt of confirmation of good transmission); or
 - (c) by overnight delivery by a nationally recognized courier service (in which case, it will be effective on the Business Day after being deposited with such courier service);

in each case, to the address (or facsimile number) listed below:

If to ECS, Force Capital, or FCP Investments, to:

Zephyr Cove, Nevada 89448 Attn: Mr. Martin J. Maslonka Telephone: (775) 586-9380

P.O. Box 740

Facsimile: (775) 586-9384

with a copy (which shall not constitute notice) to: 6644 East Thomas Road, Suite 203

> Mesa, Arizona 85215 Attn: Mr. Casey J. Maslonka Telephone: (480) 245-7200 Facsimile: (480) 830-8829

with a copy (which shall not constitute notice) to: Lewis and Roca LLP

One South Church Avenue, Suite 700

Tucson, Arizona 85701 Attn: Matthew C. Sweger, Esq. (520) 629-4431 Telephone: Facsimile: (520) 879-4722

If to Jon Maslonka, to: 143 Highway 310 P. O. Box 209

Frannie, Wyoming 82423

Telephone: (307) 664-2707

If to Justin Campbell, to: 891 Road 6

P. O. Box 183

Frannie, Wyoming 82423

Telephone: (307) 664-2707

If to Sid Strauss, to: 7130 East Saddleback Street, #58

Mesa, Arizona 85207

Telephone: (480) 218-9711

If to Martin Maslonka, to: P.O. Box 740

> Zephyr Cove, Nevada 89448 Telephone: (775) 586-9380 Facsimile: (775) 586-9384

EC Source Services, LLC with a copy (which shall not constitute notice) to:

6644 East Thomas Road, Suite 203

Mesa, Arizona 85215

Attn: Mr. Casey J. Maslonka Telephone: (480) 245-7200 Facsimile: (480) 830-8829

Lewis and Roca LLP with a copy (which shall not constitute notice) to:

One South Church Avenue, Suite 700

Tucson, Arizona 85701 Attn: Matthew C. Sweger, Esq. Telephone: (520) 629-4431 Facsimile: (520) 879-4722 If to Casey Maslonka, to: 6644 East Thomas Road, Suite 203

Mesa, Arizona 85215

Telephone: (480) 245-7200 Facsimile: (480) 830-8829

If to Parent, Merger Subsidiary or Sister Subsidiary, to:

MasTec. Inc.

800 S. Douglas Road, 12th Floor

Coral Gables, FL 33134

Telephone (305) 406-1892 Facsimile (305) 406-1900

Pablo A. Alvarez, Executive Vice President, Mergers and Acquisitions Attention:

with a copy (which shall not constitute notice) to: MasTec, Inc.

800 S. Douglas Road, 12th Floor

Coral Gables, FL 33134

Telephone: (305) 406-1849 Facsimile: (305) 406-1947

Attention: Albert de Cardenas, Executive Vice President and General Counsel

with a copy (which shall not constitute notice) to: Greenberg Traurig, P.A.

333 Avenue of the Americas

Miami, FL 33131

Telephone: (305) 579-0724 Facsimile: (305) 961-5724 Attention: David Barkus, Esq.

Each of the parties to this Agreement may specify a different address or facsimile number by giving notice in accordance with this Section 11.1 to each of the other parties hereto.

11.2 Members' Representative.

11.2.1 Appointment. "Members' Representative" means Force Capital provided that either Martin Maslonka or Casey Maslonka, or any of their Affiliates, is the beneficial owner of all of the Equity Securities thereof and materially participates in the management of such entity. The Members hereby irrevocably constitute and appoint the Members' Representative as their attorney-in-fact and agent to act in their names, place and stead in connection with all matters arising from and under this Agreement, each of the Company Agreements and any other agreements, documents or instruments related to the Contemplated Transactions and acknowledge that such appointment is coupled with an interest. Members' Representative hereby accepts such appointment and authorization.

11.2.2 Authority. The Members agree to be bound by all notices received or given by, and all agreements and determinations made by, and all documents executed and delivered by the Members' Representative under this Agreement; authorize the Members' Representative to assert claims, make demands and commence actions on behalf of the Members under this Agreement, dispute or to refrain from disputing any claim made by the Members, negotiate and compromise any dispute that may arise under, and exercise or refrain from exercising remedies available to the Members under, this Agreement, and to sign any releases or other documents with respect to such dispute or remedy (and to bind the Members in so doing), give such instructions and do such other things and refrain from doing such things as the Members' Representative shall deem appropriate to carry out the provisions of this Agreement, give any and all consents and notices under this Agreement, and perform all actions, exercise all powers, receive service of process with respect to any Action under this Agreement, any Company Agreement and any other agreement or instrument in connection with the Contemplated Transactions, agree to, negotiate and authorize payments in connection with the Purchase Price Adjustment and any other payment pursuant to the terms of this Agreement, and fulfill all duties otherwise assigned to the Members' Representative in this Agreement. Each Member hereby expressly acknowledges and agrees that the Members' Representative has the sole and exclusive authority to act on the Members' behalf in respect of all matters arising under or in connection with this Agreement after execution of this Agreement, notwithstanding any dispute or

disagreement among them, and that no Member shall have any authority to act unilaterally or independently of the Members' Representative in respect to any such matter. Parent shall be entitled to rely on any and all actions taken by the Members' Representative under this Agreement without any Liability to, or obligation to inquire of, the Members. All notices, counter notices or other instruments or designations delivered by the Members in regard to this Agreement shall not be effective unless, but shall be effective if, signed by the Members' Representative, and if not, such document shall have no force or effect whatsoever, and Parent and any other Person may proceed without regard to any such document.

- 11.2.3 <u>Change of Representative</u>. The Members' Representative may be changed by the Members upon not less than twenty (20) calendar days prior written notice to Parent; provided, that the Members' Representative may not be removed unless ECS Members holding a majority of the Common Units agree to such removal and to the identity of the substituted agent or agents. The Members' Representative may resign at any time upon not less than thirty (30) calendar days' prior written notice to Parent, but in any event, not prior to the appointment of a substitute Members' Representative. No bond shall be required of the Members' Representative. Notices or communications to or from the Members' Representative shall constitute notice to or from the Members.
- 11.2.4 A decision, act, consent or instruction of the Members' Representative, including an amendment, extension or waiver of this Agreement, shall constitute a decision of the Members and shall be final, binding and conclusive upon the Members; and Parent may conclusively and absolutely, rely, without any inquiry, upon any such decision, act, consent or instruction of the Members' Representative as being the decision, act, consent or instruction of the Members. Parent is hereby relieved from any Liability to any Person, including the Members, for any acts done by it in accordance with or reliance on such decision, act, consent or instruction of the Members' Representative.
- 11.2.5 All notices or other communications required to be made or delivered by Parent to the Members shall be made to the Members' Representative for the benefit of the Members, and any notices so made shall discharge in full all notice requirements of Parent to the Members with respect thereto. All notices or other communications required to be made or delivered by the Members to Parent shall be made by the Members' Representative for the benefit of the Members and any notices so made shall discharge in full all notice requirements of the Members to Parent with respect thereto.
- 11.3 <u>Publicity</u>. No public announcement or disclosure will be made by any party with respect to the subject matter of this Agreement or the Contemplated Transactions without the prior written consent of Parent and the Member's Representative; <u>provided</u>, <u>however</u>, that the provisions of this <u>Section 11.3</u> will not prohibit (a) any disclosure required by any applicable Legal Requirements, including any disclosure necessary or desirable to provide proper disclosure under the securities laws or under any rules or regulations of any securities exchange on which the securities of such party may be listed or traded or (b) any disclosure made in connection with the enforcement of any right or remedy relating to, or the performance of any obligation arising under, this Agreement or the Contemplated Transactions.
- 11.4 <u>Succession and Assignment; No Third-Party Beneficiary</u>. Subject to the immediately following sentence, this Agreement will be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, each of such successors and permitted assigns will be deemed to be a party hereto for all purposes hereof. No party may assign, delegate or otherwise transfer either this Agreement or any of its rights, interests, or obligations hereunder (other than transfers by operation of law) without the prior written approval of the other parties; provided, however, that Parent may (a) assign any or all of its rights and interests hereunder to one or more of its Affiliates and (b) designate one or more of its Affiliates to perform its obligations hereunder, provided that any such assignment or designation shall in no event limit, reduce or release Parent's obligations hereunder. Except as expressly set forth in Section 9 with respect to Indemnified Persons who are not parties to this Agreement, this Agreement is for the sole benefit of the parties hereto and their permitted successors and assignees and nothing herein expressed or implied will give or be construed to give any Person, other than the parties and such successors and assignees, any legal or equitable rights hereunder.
- 11.5 Amendments and Waivers. No amendment or waiver of any provision of this Agreement will be valid and binding unless it is in writing and signed, in the case of an amendment, by the Parent and the Members' Representative, or in the case of a waiver, by the party or parties against whom the waiver is to be effective. No waiver by any party of any breach or violation or, default under or inaccuracy in any representation, warranty or covenant hereunder, whether intentional or not, will be deemed to extend to any prior or subsequent breach, violation, default of, or inaccuracy in, any such representation, warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence. No delay or omission on the part of any party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof.
- 11.6 <u>Further Assurances</u>. From and after the date hereof, upon the request of the Members' Representative or Parent, each party will do, execute, acknowledge and deliver all such further acts, assurances, deeds, assignments, transfers, conveyances and other instruments and papers as may be reasonably required or appropriate to carry out the Contemplated Transactions.
- 11.7 Entire Agreement. This Agreement, the Company Agreements and the other agreements and documents to be executed and delivered pursuant hereto constitute the entire agreement among the parties hereto with respect to the subject matter hereof and supersede any and all prior and contemporaneous discussions, negotiations, proposals, undertakings, understandings and agreements,

whether written or oral, with respect thereto. Notwithstanding the foregoing, some or all of the Members may currently or in the future be party to other restrictive covenant agreements with the Companies, Parent, or others, and nothing in this Agreement is intended to supersede or otherwise affect any such other restrictive covenant agreement.

- 11.8 <u>Disclosure Schedules; Listed Documents, etc.</u> Neither the listing nor description of any item, matter or document in any Schedule hereto nor the furnishing or availability for review of any document will be construed to modify, qualify or disclose an exception to any representation or warranty of any party made herein or in connection herewith, except to the extent that (a) such representation or warranty specifically refers to the disclosures in such section of the Disclosure Schedule and (b) to the extent disclosed in another section of the Disclosure Schedules, to the extent that it is reasonably apparent from a reading of the disclosure that such disclosure would also qualify or apply to such other section. The inclusion of any information in any Disclosure Schedule shall not be deemed to be an admission or evidence of the materiality of such item. In addition, matters reflected in the Disclosure Schedules are not necessarily limited to matters required by this Agreement to be reflected in the Disclosure Schedules. Such additional matters are set forth for informational purposes only and do not necessarily include other matters of a similar nature. The information contained in this Agreement, in the Disclosure Schedules, and exhibits hereto is disclosed solely for purposes of this Agreement, and no information contained herein or therein shall be deemed to be an admission by any party hereto to any Person other than the parties hereto of any matter whatsoever (including any violation of Law or breach of contract).
- 11.9 <u>Counterparts; Execution</u>. This Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which together will constitute but one and the same instrument. This Agreement will become effective when duly executed by each party hereto. Facsimile or other electronically scanned and transmitted signatures shall be deemed originals and shall constitute valid execution and acceptance of this Agreement by the signing/transmitting party.
- 11.10 <u>Survival</u>. The covenants and agreements and, subject to <u>Section 9.3</u>, the representations and warranties set forth in this Agreement shall survive and remain in effect after the Closing.
- 11.11 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If any provision hereof would, under applicable Law, be invalid or unenforceable in any respect, then each party hereto intends that such provision will be construed by modifying or limiting it so as to be valid and enforceable to the maximum extent compatible with, and possible under, applicable Law.
- 11.12 <u>Headings</u>. The headings contained in this Agreement are for convenience purposes only and will not in any way affect the meaning or interpretation hereof.
- 11.13 Construction. The parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the parties and no presumption or burden of proof will arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement. The parties intend that each representation, warranty and covenant contained herein will have independent significance. If any party has breached or violated, or if there is an inaccuracy in, any representation, warranty or covenant relating to the same subject matter (regardless of the relative levels of specificity) which the party has not breached or violated, or in respect of which there is not an inaccuracy, will not detract from or mitigate the fact that the party has breached or violated, or there is an inaccuracy in, the first representation, warranty or covenant.
- 11.14 Governing Law. This Agreement, the rights of the parties and all Actions arising in whole or in part under or in connection herewith, will be governed by and construed in accordance with the laws of the State of Florida, without giving effect to any choice or conflict of law provision or rule that would cause the application of the laws of any other jurisdiction.

11.15 Jurisdiction; Venue; Service of Process.

11.15.1 <u>Jurisdiction</u>. Each party, by its execution hereof, (a) hereby irrevocably submits to the exclusive jurisdiction of the state courts of the State of Florida or the United States District Courts located in the Southern District of Florida for the purpose of any Action between or among the parties (or any of them), including but not limited to any such Action arising or related in whole or in part under or in connection with this Agreement, (b) hereby waives to the extent not prohibited by applicable law, and agrees not to assert, by way of motion, as a defense or otherwise, in any such Action, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that any such Action brought in one of the above-named courts should be dismissed on grounds of *forum non conveniens*, should be transferred or removed to any court other than one of the above-named courts, or should be stayed by reason of the pendency of some other proceeding in any other court other than one of the above-named courts, or that this Agreement or the subject matter hereof may not be enforced in or by such court and (c) hereby agrees not to commence any such Action other than before one of the above-named courts. Notwithstanding the previous sentence, (x) a party may commence any Action in a court other than the above-named courts in accordance with the provisions of Section 5.2.3.

- 11.15.2 <u>Venue</u>. With the exception of Actions brought by Parent in accordance with the provisions of <u>Section 5.2.3</u>, each party agrees that for any Action between or among the parties (or any of them), including but not limited to any such Action arising or related in whole or in part under or in connection with this Agreement, such party will bring Actions only in Miami-Dade County, Florida. Each party further waives any claim and will not assert that venue should properly lie in any other location within the selected jurisdiction.
- 11.15.3 Service of Process. Each party hereby (a) consents to service of process in any Action between or among the parties (or any of them) in any manner permitted by Florida law, (b) agrees that service of process made in accordance with clause (a) or made by registered or certified mail, return receipt requested, at its address specified pursuant to Section 11.1, will constitute good and valid service of process in any such Action and (c) waives and agrees not to assert (by way of motion, as a defense, or otherwise) in any such Action any claim that service of process made in accordance with clause (a) or (b) does not constitute good and valid service of process.
- 11.16 Waiver of Jury Trial. THE PARTIES HEREBY WAIVE, AND COVENANT THAT THEY WILL NOT ASSERT (WHETHER AS PLAINTIFF, DEFENDANT OR OTHERWISE), ANY RIGHT TO TRIAL BY JURY IN ANY ACTION BETWEEN OR AMONG THE PARTIES (OR ANY OF THEM), INCLUDING BUT NOT LIMITED TO ANY SUCH ACTION ARISING OR RELATED IN WHOLE OR IN PART UNDER OR IN CONNECTION WITH THIS AGREEMENT OR ANY OF THE CONTEMPLATED TRANSACTIONS, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE. THE PARTIES AGREE THAT ANY OF THEM MAY FILE A COPY OF THIS PARAGRAPH WITH ANY COURT AS WRITTEN EVIDENCE OF THE KNOWING, VOLUNTARY AND BARGAINED-FOR AGREEMENT AMONG THE PARTIES IRREVOCABLY TO WAIVE THEIR RIGHTS TO TRIAL BY JURY IN ANY ACTION WHATSOEVER BETWEEN OR AMONG THEM, WHICH ACTION WILL INSTEAD BE TRIED IN A COURT OF COMPETENT JURISDICTION BY A JUDGE SITTING WITHOUT A JURY.
- 11.17 Expenses. Except as provided in Sections 1.11.3, 9 or 10, (i) ECS shall pay all of ECS's and the Members' fees and expenses, including legal and accounting fees and expenses and the brokers fees and expenses disclosed on Schedules 2.30(a) and 2.30(b), incurred in connection with this Agreement and the Contemplated Transactions; provided however, that any such fees and expenses that have not been paid on or before the Closing Date and are not reflected as a Liability in the Actual Closing Balance Sheet shall be paid by the Members, and (ii) the Parent shall pay all of its own fees and expenses incurred in connection with this Agreement and the Contemplated Transactions.
- 11.18 Specific Performance. Parent shall have the right and remedy, in addition to any others that may be available, at law or in equity, to have the provisions of this Agreement, and each Company Agreement, specifically enforced through injunctive or other relief, without the necessity of posting a bond, it being acknowledged that each Company is a unique asset, with value to Parent's business not readily quantifiable and any breach by ECS or Members of the provisions of this Agreement or the Company Agreements will cause irreparable injury to Parent, the amount of which will be difficult to determine, and that money damages will not provide an adequate remedy to Parent. ECS and Members covenant and agree that they shall not, and shall not authorize any other Person to, challenge or question the enforceability of any provision of this Section 11.18.
- 11.19 <u>Joint and Several</u>. Except as set forth in <u>Section 9.4.2</u>, the covenants, agreements, obligations and Liabilities of ECS and Members hereunder and under each Company Agreement, shall be joint and several in all respects. Where in this Agreement or any agreement or document referred to in the preceding sentence, provision is made for any action to be taken or not to be taken by any Company or Members, ECS and Members agree to cause such Persons to take or not to take such action and shall be jointly and severally Liable for such Persons taking or not taking any such action, as the case may be.
 - 11.20 Recitals. The recitals to this Agreement are true and correct and by this reference are incorporated herein and made a part of this Agreement.
- 11.21 Representation by Lewis and Roca LLP. Parent hereby acknowledges that Lewis and Roca LLP has served as legal counsel to ECS, Force Capital, FCP Investments, Martin Maslonka, and certain of their Affiliates in connection with this Agreement and the Contemplated Transactions and on other matters prior to the parties commencing discussions regarding the Contemplated Transactions. Parent and ECS acknowledge and agree that Lewis and Roca LLP may continue to serve as legal counsel for Force Capital, FCP Investments and Martin Maslonka from and after the Closing with respect to the Contemplated Transactions.

12. DEFINITIONS; CERTAIN RULES OF CONSTRUCTION.

- 12.1 <u>Definitions</u>. As used herein, the following terms will have the following meanings:
- "1933 Act" means the Securities Act of 1933.
- "AAA" shall mean the American Arbitration Association.
- "Accounts Receivable" means the aggregate amount of accounts, commissions and debts payable to the Companies. For all purposes hereunder, the Accounts Receivable shall be valued at their net realizable value, net of an allowance for bad debts.
- "Action" means any claim, action, cause of action or suit (whether in contract, tort or otherwise), litigation (whether at law or in equity, whether civil or criminal), controversy, assessment, arbitration, investigation, hearing, charge, complaint, demand, notice or proceeding to, from, by or before any Governmental Authority.
- "Affiliate" means with respect to any specified Person, (a) each Person directly or indirectly controlling, controlled by or under direct or indirect common control with such specified Person at such time, (b) each Person who is at such time an officer, limited liability company manager or director of, or direct or indirect beneficial holder of at least 10% of any class of the capital stock of, such specified Person, (c) each Person that is managed by a common group of executive officers, limited liability company managers and/or directors as such specified Person, (d) the Members of the Immediate Family (i) of each officer, director, limited liability company manager or holder described in clause (b) and (ii) if such specified Person is an individual, of such specified Person, and (e) each Person of which such specified Person or an Affiliate (as defined in clauses (a) through (d)) thereof will, directly or indirectly, beneficially own at least 10% of any class of equity interests at such time.
- "Backlog" means the amount of revenue that the Companies, based on their past practices for calculating such amounts, expect to recognize from:
 (i) remaining work to be performed on uncompleted Contractual Obligations in progress and (ii) executed Contractual Obligations on which work has not yet begun.
- "Business Day" means any weekday other than a weekday on which banks in Mesa, Arizona or Coral Gables, Florida are authorized or required to be closed.
- "<u>Businesses</u>" means the businesses of the Companies, including financing, developing, designing, engineering, constructing, maintaining, deploying, project planning and execution of electric transmission and substation projects; and the provision of material procurement, expediting, environmental planning, engineering, testing, commissioning, and construction services for electric transmission and substation projects.
 - "CERCLA" means the Comprehensive Environmental Response Compensation and Liability Act of 1980.
 - "CERCLIS" means the Comprehensive Environmental Response Compensation and Liability Information System.
 - "Code" means the U.S. Internal Revenue Code of 1986.
- "Common Unit Percentage" means, with respect to any ECS Member, the number of Common Units held by such ECS Member immediately prior to the Effective Time of the Merger as set forth on Schedule 1.9.1(b) divided by the total outstanding Common Units immediately prior to the Effective Time of the Merger as set forth on Schedule 1.9.1(b).
 - "Company Software" means Owned Software and Licensed Software.
- "Company Technology" means any and all Technology used in connection with the Businesses and any and all Intellectual Property in any and all such Technology.
- "Compensation" means, with respect to any Person, all salaries, compensation, remuneration, bonuses or benefits of any kind or character whatever, paid or provided directly or indirectly by any Company to such Person or Affiliates of such Person.
- "Contaminant" means any pollutant, hazardous substance, radioactive substance, toxic substance, hazardous waste, medical waste, radioactive waste, special waste, petroleum or petroleum-derived substance or waste, asbestos, polychlorinated biphenyls, or any hazardous or toxic constituent thereof and includes any substance defined in or regulated under any Environmental Law.

"Contemplated Transactions" means the Merger and the other transactions contemplated by this Agreement, the purchase of the Preferred Units pursuant to the Membership Interest Purchase Agreement and all of the other transactions contemplated by the Company Agreements.

"Contractual Obligation" means, with respect to any Person, any contract, agreement, deed, mortgage, lease, license, commitment, promise, undertaking, arrangement, performance bond, warranty obligation or understanding, whether written or oral and whether express or implied, or other document or instrument (including any document or instrument evidencing or otherwise relating to any Debt), to which or by which such Person is a party or otherwise subject or bound or to which or by which any property, business, operation or right of such Person is subject or bound.

"CPA Firm" means, unless otherwise agreed in writing by Members' Representative and Parent, an accountant mutually satisfactory to Members' Representative and Parent who satisfies each of the following requirements (unless otherwise agreed by Members' Representative and Parent): (i) neither the accountant nor the firm that employs the accountant shall have performed any accounting or consulting services for any party or any Affiliate of any party at any time during the three year period prior to the date of this Agreement; (ii) the accountant is not related in any way by blood or marriage to any party or any executive officer or director of any party or any Affiliate of such party; (iii) the accountant has been a certified public accountant duly licensed to practice in the state where he or she has his or her primary office for a period of not less than ten years; and (iv) the accountant is willing to accept engagement as a CPA Firm on the terms and conditions of this Agreement, or, if the Members' Representative and Parent fail or refuse to select a firm within 10 calendar days after written request therefor by the Members' Representative or Parent, such an independent nationally recognized accounting firm shall be selected in accordance with the rules of the Miami, FL office of the AAA.

"Debt" means, with respect to any Person, all obligations (including all obligations in respect of principal, accrued interest, penalties, fees and premiums) of such Person, whether direct or indirect, (a) for borrowed money (including overdraft facilities), (b) for Liabilities secured by any Encumbrance existing on property owned or acquired and subject thereto, (c) evidenced by notes, bonds, debentures or similar Contractual Obligations, (d) for the deferred purchase price of property, goods or services, including in connection with the acquisition of any business or non-competition agreement (other than trade payables or accruals incurred in the Ordinary Course of Business), (e) under capital leases (in accordance with GAAP), (f) in respect of letters of credit and bankers' acceptances, (g) for Contractual Obligations relating to interest rate protection, swap agreements, factoring, hedging and collar agreements, (h) in the nature of premiums (prepayment or otherwise) or penalties in connection with the obligations described in clauses (a) through (g) above, and (h) in the nature of Guarantees of the obligations described in clauses (a) through (g) above of any other Person.

"<u>Disclosed Liabilities</u>" means those Liabilities, matters, benefit plans, Actions, disputes, violations, citations, penalties, claims, and potential or threatened Liabilities, Actions, disputes, violations, citations, penalties, claims disclosed on any of ECS's or the Members' Disclosure Schedules or otherwise disclosed in an update to a Schedule pursuant to Section 4.6.

"Disclosure Schedules" means, together, ECS's and the Members' Disclosure Schedule and Parent's Disclosure Schedule.

"DOJ" means the United States Department of Justice.

"ECS 401K Plan" means the EC Source Services, LLC 401(k) Plan sponsored by ECS with an effective date of December 1, 2009.

"ECS's Knowledge" and "of which ECS has Knowledge" or similar phrases means the knowledge of the Members, Brian Bratton, Christine Vidrine, Brian Smaistrla, and Bill Wright. Such Persons will be deemed to have knowledge of a particular fact, circumstance, event or other matter if(a) any such individual has actual knowledge of such fact, circumstance, event or other matter, (b) such fact, circumstance, event or other matter is reflected in one (1) or more documents (whether written or electronic, including electronic mails sent to or by such individual) in the possession of such individual, including his or her personal files, (c) such fact, circumstance, event or other matter is reflected in one (1) or more documents (whether written or electronic) contained in books and records of such individual that would reasonably be expected to be reviewed by such individual in the customary performance of his or her duties or (d) such fact, circumstance, event or other matter would be known to such individual had he or she made reasonable inquiry of appropriate employees.

"ECS Options" means options to purchase Common Units from ECS pursuant to the Unit Incentive Plan.

"ECSD" means ECSD, LLC, a Arizona limited liability company.

"Encumbrance" means any charge, claim, community or other marital property interest, condition, equitable interest, lien, license, option, pledge, security interest, mortgage, right of way, easement, encroachment, servitude, right of first offer or first refusal, buy/sell agreement and any other restriction or covenant with respect to, or condition governing the use, construction, voting (in the case of any security or equity interest), transfer, receipt of income or exercise of any other attribute of ownership.

"Enforceable" means, with respect to any Contractual Obligation stated to be "Enforceable" by or against any Person, that such Contractual Obligation is a legal, valid and binding obligation of such Person enforceable by or against such Person in accordance with its terms.

"Environmental Laws" means all Legal Requirements relating to or addressing the environment, health or safety, which shall include the use, handling, treatment, storage or disposal of any Contaminant, or workplace or worker safety and health.

"Environmental Lien" means any lien in favor of any Governmental Authority for any (a) Liability under any Environmental Laws, or (b) damages arising from, or costs incurred by, such Governmental Authority in response to a Release or threatened Release of a Contaminant into the environment.

"Environmental Liabilities and Costs" means any and all Liabilities, regardless of when made or asserted or imposed or asserted to be imposed by operation of Law, for all environmental, ecological, health, or safety Liabilities or personal injury, property damage or natural resource damage, claims pertaining to any Business or any Company relating to or arising from time periods or events occurring on or prior to the Closing Date, including (i) any uses of or occurrences on the Real Property (or any real property previously owned, operated or leased by any Business or any Company) whether or not such uses and occurrences were in compliance with then applicable Environmental Laws, (ii) the presence of any Contaminant or any Release or the threat of Release of any Contaminant on, at or from the Real Property (or any real property previously owned, operated or leased by any Business or Company), (iii) the storage, handling, treatment or disposal of any Contaminant transported from the Real Property (or any real property previously owned, operated or leased by any Business or Company) whether or not such transportation, storage, handling, treatment or disposal was in compliance with then applicable Environmental Laws, (iv) human exposure to any Contaminant, (v) non-compliance with any Environmental Laws or permits required under Environmental Laws, (vi) any Remedial Action, and (vii) any reasonable fees and expenses of any environmental engineers and attorneys incurred by Parent in connection with any Environmental Liabilities and Costs or the investigation thereof.

"Environmental Report" means any report, study, assessment, audit, or other similar document that addresses any issue of actual or potential noncompliance with, actual or potential liability under or cost arising out of, or actual or potential impact on business in connection with, any Environmental Law or any proposed or anticipated change in or addition to Environmental Law, or regarding any Contaminant, that may affect ECSD, any Member or any Company, including any Person for which any Company may be liable.

"Equity Security" of any Person means any (i) capital stock, membership or partnership interest or other ownership interest of or in such Person, (ii) securities directly or indirectly convertible into or exchangeable for any for the foregoing; (iii) options, warrants or other rights directly or indirectly to purchase or subscribe for any of the foregoing or securities convertible into or exchangeable for any of the foregoing; or (iv) contracts, commitments, agreements, understandings, arrangements, calls or claims of any kind relating to the issuance of any of the foregoing or giving any Person the right to participate in or receive any payment based on the profits or performance of such Person (including any equity appreciation, phantom equity or similar plan or right).

"ERISA" means the federal Employee Retirement Income Security Act of 1974.

"Exchange Act" means the Securities Exchange Act of 1934.

"Excluded Damages" means any incidental, consequential (including lost profits), punitive, special, indirect or exemplary damages, damages for diminution in value, or damages calculated on a multiple of earnings or similar basis.

"Facilities" means any buildings, plants, improvements or structures located on the Real Property.

"FTC" means the United States Federal Trade Commission.

"GAAP" means generally accepted accounting principles in the United States as in effect from time to time.

"Governmental Authority" means any United States federal, state or local or any foreign government, or political subdivision thereof, or any multinational organization or authority or any authority, agency or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, any court or tribunal (or any department, bureau or division thereof), or any arbitrator or arbitral body.

"Governmental Authorization" means any approval, consent, license, Permit, waiver, or other authorization issued, granted, given, or otherwise made available by or under the authority of any Governmental Authority or pursuant to any Legal Requirement.

"Governmental Order" means any order, writ, judgment, injunction, decree, stipulation, ruling, determination or award entered by or with any Governmental Authority in a judicial or administrative proceeding.

"Guarantee" means, with respect to any Person, (a) any guarantee of the payment or performance of, or any contingent obligation in respect of, any Debt or other Liability of any other Person; (b) any other arrangement whereby credit is extended to any obligor (other than such Person) on the basis of any promise or undertaking of such Person (i) to pay the Debt or other Liability of such obligor, (ii) to purchase any obligation owed by such obligor, (iii) to purchase or lease assets under circumstances that are designed to enable such obligor to discharge one or more of its obligations or (iv) to maintain the capital, working capital, solvency or general financial condition of such obligor; and (c) any liability as a general partner of a partnership or as a venturer in a joint venture in respect of Debt or other obligations of such partnership or venture.

"HSR Act" means the Hart-Scott-Rodino Premerger Notification Act, 15 U.S.C. 18a.

"Indemnified Person" means, with respect to any Indemnity Claim, the Person asserting such claim under Section 9.1 or 9.2, as the case may be.

"Indemnifying Person" means, with respect to any Indemnity Claims, (i) ECS and Members under Section 9.1, or (ii) Parent under 9.2, respectively, against whom such claim is asserted.

"Indemnity Claim" means a claim for indemnity under Section 9.1 or 9.2.

- "Intellectual Property" means the entire right, title and interest in and to all proprietary rights of every kind and nature anywhere, including all rights and interests pertaining to or deriving from:
 - (a) patents, copyrights, mask work rights, Technology, know-how, processes, Trade Secrets, algorithms, inventions, works, proprietary data, databases, formulae, research and development data and Software;
 - (b) trademarks, trade names, service marks, service names, brands, trade dress and logos, and the goodwill and activities associated therewith;
 - (c) domain names, rights of privacy and publicity, moral rights, and proprietary rights of any kind or nature, however denominated, throughout the world in all media now known or hereafter created;
 - (d) any and all registrations, applications, recordings, licenses, common-law rights and Contractual Obligations relating to any of the foregoing; and
 - (e) all Actions and rights to sue at law or in equity for any past, present or future infringement or other impairment of any of the foregoing, including the right to receive all proceeds and damages therefrom, and all rights to obtain renewals, continuations, divisions or other extensions of legal protections pertaining thereto.

"Inventory" means all inventory related to the Businesses, wherever located, including all finished goods whether held at any location or facility of any Company or in transit to any Company.

"Investment" means (a) any direct or indirect ownership, purchase or other acquisition by a Person of any notes, obligations, instruments, Equity Securities (including joint venture interests) of any other Person; and (b) any capital contribution or similar obligation by a Person to any other Person.

"Knowledge Qualifier" means any reference or qualification to ECS's Knowledge, Knowledge of ECS, or any similar phrase regarding the knowledge, belief or understanding of ECS.

"Law" means any United States federal, national, foreign, supranational, state, provincial, local or similar statute, law, standard, resolution, promulgation, ordinance, regulation, rule, code, order, requirement or rule of law (including common law), or any similar provision having the force or effect of law.

"Legal Requirement" means any Law, Governmental Order or Permit.

"<u>Liability</u>" means, with respect to any Person, any liability or obligation of such Person, of any kind, character or description, whether known or unknown, whether asserted or unasserted, whether executory, determined, determinable or otherwise, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, whether disputed or undisputed, whether disclosed or undisclosed, whether incurred or consequential, whether secured or unsecured, joint or several, vested or unvested, whether due or to become due, whether choate or inchoate and whether or not required under GAAP to be accrued on the financial statements of such Person and regardless of whether such debt, duty or liability is immediately due and payable, and including all costs and expenses related thereto.

"<u>Licensed Software</u>" means all Software that is owned by any third party and that is licensed to and used by any Company in the conduct of its respective Businesses.

"Losses" means any loss, liability, claim, damage, expense (including costs of defense and reasonable attorneys' fees under circumstances a party is entitled to incur such costs and fees pursuant to this Agreement); provided, however, the term Losses shall be limited to actual damages and shall exclude Excluded Damages, except (i) with respect to any breach of the representations and warranties set forth in Section 2.6 (Financial Statements) as to which Losses shall include damages for a diminution of value, indirect damages and other special damages, and (ii) in the case of fraud or intentional misrepresentation in which event Losses shall include all actual damages and all Excluded Damages.

"Material Adverse Effect" means any change in, development, event, occurrence or effect on, the Businesses, operations, Assets, prospects, condition (financial or otherwise) or results of operations of the Companies which, whether viewed on a short-term or a long-term basis, when considered either individually or in the aggregate together with all other adverse changes or effects with respect to which such phrase is used in this Agreement, (a) is, or would reasonably be expected to be, materially adverse to the Businesses, operations, Assets, prospects, condition (financial or otherwise) or results of operations of the Companies, or (b) prevents or materially delays ECS's or any Member's ability to perform its obligations hereunder.

"Members of the Immediate Family" means, with respect to any individual, (a) such Person's spouse, (b) each parent, brother, sister or child of such Person or such Person's spouse, (c) the spouse of any Person described in clause (b) above, (d) each child of any Person described in clauses (a), (b) or (c) above, (e) each trust, to the extent controlled by such individual, created solely for the benefit of such individual or one or more of the Persons described in clauses (a) through (d) above and (f) each custodian or guardian of any property of such individual or one or more of the Persons described in clauses (a) through (e) above in his capacity as such custodian or guardian, and (g) the personal representative, executor or administrator of the estate of such individual or one or more of the Persons described in clauses (a) through (e) above in his capacity as such personal representative, executor or administrator.

"Net Working Capital" means as of any particular date (a) the value of all of the Companies' current assets, including cash and cash equivalents and retainage (but only to the extent a current asset as determined in accordance with GAAP), but excluding any Receivables not collected on or prior to the 90 h day following the Closing Date and any inter-Company Receivables, less (b) the amount of all of the Companies' current Liabilities (x) excluding any inter-Company Liabilities and (y) including accrued net current Liabilities not yet due, all as determined in accordance with GAAP.

"New Operating Agreement" means the Fifth Amended and Restated Operating Agreement of ECS, dated as of the date hereof.

"NPL" means the list of national priorities among the known Releases or threatened Releases of Contaminants throughout the United States and its territories.

"Ordinary Course of Business" means an action taken by any Person in the ordinary course of such Person's business which is consistent with the past customs and practices of such Person (including past practice with respect to quality, quantity, amount, magnitude and frequency, standard employment and payroll policies and past practice with respect to management of working capital) which is taken in the ordinary course of the normal day-to-day operations of such Person and does not require the consent of the shareholders, board of directors, members or managers of such Person.

"Organizational Documents" means, with respect to any Person (other than an individual), (a) the certificate or articles of incorporation, formation or organization, and any joint venture, limited liability company, operating or partnership agreement and other similar documents entered into or adopted at any time or filed in connection with the creation, formation or organization of such Person and (b) all by-laws, shareholders' agreements, voting agreements, rights of first refusal and similar documents, instruments or agreements relating to the organization or governance of such Person, in each case, as amended or supplemented.

"Owned Software" means all Software used by each Company in the conduct of its respective Businesses that is owned or purported to be owned by such Company.

"Parent Shares" means shares of common stock, \$.10 par value per share of Parent.

"Per Unit Merger Consideration" means the Per Unit Closing Date Merger Consideration plus the Per Unit Earn-Out Merger Consideration.

"<u>Permits</u>" means, with respect to any Person, any license, franchise, permit, consent, approval, right, privilege, certificate or other similar authorization issued by, or otherwise granted by, any Governmental Authority or any other Person to which or by which such Person is subject or bound or to which or by which any property, business, operation or right of such Person is subject or bound.

"Permitted Encumbrance" means (a) statutory liens for current Taxes, special assessments or other governmental charges not yet due and payable or the amount or validity of which is being contested in good faith by appropriate proceedings and for which appropriate reserves have been established in books and records in accordance with GAAP, (b) mechanics', materialmen's, carriers', workers', repairers' and similar statutory liens arising or incurred in the Ordinary Course of Business which liens are not material in amount and do not interfere with any current or anticipated operation of any Business, (c) zoning, entitlement, building and other land use regulations imposed by governmental agencies having jurisdiction over any Real Property which are not violated by the current or contemplated use and operation of the Real Property, (d) restrictions on the transfer of securities arising under federal and state securities laws and (e) and with respect to any Real Property or any Facility, any patent reservations, covenants, conditions, restrictions, servitudes, liens, reservations, easements, declarations, or other matters of record or to which reference is made in the public records that do not interfere with the use of the Real Property or the Facility for the conduct of the Businesses by the Companies.

"Person" means any individual or corporation, association, partnership, limited liability company, joint venture, joint stock or other company, business trust, trust, organization, Governmental Authority or other entity of any kind.

"PP&E" means property, plant and equipment, including trucks, trailers and other construction equipment.

"Release" means the release, spill, emission, leaking, pumping, injection, deposit, disposal, discharge, dispersal, leaching or migrating into the indoor or outdoor environment of any Contaminant through, in, into or from the air, soil, surface water, groundwater or any property.

"Remedial Action" means actions required to (i) clean up, remove, treat or in any other way address Contaminants in the indoor or outdoor environment; (ii) prevent the Release or threat of Release or minimize the further Release or threat of Release of Contaminants; or (iii) investigate and determine if a remedial response is needed, design such a response and perform post-response investigation, monitoring, operation, maintenance and care.

"Representative" means, with respect to any Person, any director, officer, employee, agent, consultant, advisor, or other representative of such Person, including legal counsel, accountants, and financial advisors.

"Required Employees" means each of Brian Bratton, Casey Maslonka, Jon Maslonka, Brian Smaistrla, Justin Campbell, Bill Wright, Sid Strauss, Debbie Greenacre, and any other officer of the Companies designated by Parent.

"Software" means computer software or firmware in any form, including object code, source code, computer instructions, commands, programs, modules, routines, procedures, rules, libraries, macros, algorithms, tools, and scripts, and all documentation of or for any of the foregoing.

"Tangible Net Worth" means the Companies' total Assets, minus total Liabilities, minus goodwill and other intangible assets. For the avoidance of doubt, Tangible Net Worth shall include the Assets set forth on Exhibit H.

"Tax" or "Taxes" means (1) any federal, state, local, or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental (including taxes under Section 59A of the Code), custom duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, or other tax of any kind whatsoever, including any interest, penalty, or addition thereto, whether disputed or not, and any amounts payable pursuant to the determination or settlement of an audit, and (2) liability of any Company for the payment of any amounts of the type described in clause (1) above as a result of any express or implied obligation to indemnify or otherwise assume or succeed to the liability of any other Person.

"<u>Tax Return</u>" means any return, declaration, report, claim for refund or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Technology" means all inventions, works, discoveries, innovations, know-how, information (including ideas, research and development, formulas, compositions, processes and techniques, data, designs, drawings, specifications, customer and supplier lists, pricing and cost information, business and marketing plans and proposals, documentation and manuals), Software, computer hardware, integrated circuits and integrated circuit masks, electronic, electrical and mechanical equipment and all other forms of technology, including improvements, modifications, works in process, derivatives or changes, whether tangible or intangible, embodied in any form, whether or not protectable or protected by patent, copyright, mask work right, trade secret law or otherwise, and all documents and other materials recording any of the foregoing.

"Trade Secrets" means trade secrets, know how and confidential business information and any other information, however documented, that is a trade secret within the meaning of the applicable trade secret protection Laws, including the Uniform Trade Secrets Act.

"Treasury Regulations" means the regulations promulgated under the Code.

"<u>Unit Equivalent Holder</u>" means each of the Persons listed on <u>Schedule 1.9.1(b)</u>, which shall include all holders (or, for clarity, after the Effective Time previous holders) of Unit Equivalents.

"<u>Unit Equivalent Percentage</u>" means, with respect to any Unit Equivalent Holder, the number of Unit Equivalents held by such Unit Equivalent Holder immediately prior to the Effective Time of the Merger as set forth on <u>Schedule 1.9.1(b)</u> divided by the total outstanding Unit Equivalents immediately prior to the Effective Time of the Merger as set forth on <u>Schedule 1.9.1(b)</u>.

"Unit Equivalents" shall mean the number of outstanding Units (other than the Preferred Units) and the number of Units issuable upon exercise of the ECS Options.

"Unit Incentive Plan" means that certain Unit Incentive Plan of ECS dated January 2, 2008, as amended on the date hereof.

"Unpaid Preferred Dividend" has the meaning set forth in the New Operating Agreement.

12.2 Glossary of Other Defined Terms. The following sets forth the location of definitions of capitalized terms defined in the body of this Agreement:

<u>Term</u>	Location
"Acquisition Proposal"	Section 4.4.1
"Actual Closing Balance Sheet"	Section 1.11.4
"Actual Net Working Capital"	Section 1.11.1
"Actual PP&E"	Section 1.11.1
"Actual Tangible Net Worth"	Section 1.11.1
"Agreement"	<u>Preamble</u>
"Articles of Organization"	Section 1.5.1
"Articles of Merger"	Section 1.1
"Assets"	Section 2.9.1
"ASTM Phase I ESA Standard"	Section 4.9.1
"ASTM Phase II ESA Standard"	Section 4.9.2
"ATP"	Recitals
"Audited Financials"	Section 4.8
"Campbell"	<u>Preamble</u>
"Certificates"	Section 1.8.2
"Closing"	Section 1.3.1
"Closing Balance Sheet"	Section 1.11.1
"Closing Certificate"	Section 1.10.1
"Closing Date"	<u>Section 1.3.1</u>
"Commercial Software"	Section 2.14.1
"Common Units"	<u>Recitals</u>
"Company" and "Companies"	Recitals
"Company Agreements"	Section 2.3
"Company Employees"	Section 2.17(a)
"Company Indemnified Person"	Section 9.2
"Company Systems"	<u>Section 2.14.8</u>
"Competitive Business"	<u>Section 5.2.1(a)</u>
"Controlled Group"	Section 2.17(c)
"Customer Assets"	Section 2.16
"CPA Firm's Position"	<u>Section 1.11.3</u>
"Data Room"	Section 1.3.2(a)(iv)
"Deductible"	<u>Section 9.4.1</u>
"Default Notice"	<u>Section 1.9.2(h)</u>
"Defaulting Member"	<u>Section 1.9.2(h)</u>
"De Minimis Claim"	<u>Section 9.4.1</u>
"Disclosed Contract"	<u>Section 2.19.2</u>
"Earn-Out Dispute Expenses"	<u>Section 1.9.2(e)</u>
"Earn-Out Objection"	<u>Section 1.9.2(e)</u>
"Earn-Out Payment"	<u>Section 1.9.2(a)</u>
"Earn-Out Period"	Section 1.9.2(b)(vii)

Term Location "Earn-Out Period Share Value" Section 1.9.2(c) "Earn-Out Review Period" Section 1.9.2(e) "EBITDA" Section 1.9.2(b)(x)"EBITDA Threshold" Section 1.9.2(b)(viii) "ECS" Preamble "ECS Audited Financials" Section 2.6.1 "ECS Aviation" Recitals "ECS Aviation Financials" Section 2.6.1 Section 2.2.5 "ECS Award Agreements" "ECS Members" **Preamble** "ECS Member's Objection" Section 1.11.2 "EEG" Recitals "Effective Time" Section 1.2 "Employee Plans" Section 2.17(a) "Entity Members" Preamble "Environmental Consultant" Section 4.9.1 "Equipment" Section 2.13 "Estimated Closing Balance Sheet" Section 1.10.1 "Estimated Net Working Capital" Section 1.10.1(a) "Estimated PP&E" Section 1.10.1(c) "Estimated Tangible Net Worth" Section 1.10.1(b) "Excess Cash" Section 1.10.4 "Excess De Minimis Claim" Section 9.4.1 "Exchange Agent" Section 1.12.1 Section 1.9.2(b)(x)"Existing Operations" "FCP Investments" **Preamble** "Final Closing Statement" Section 1.11.1 "Financials" Section 2.6.1 "First Earn-Out Period" Section 1.9.2(b)(i) "FBCA" Recitals "FLLCA" Recitals Section 2.25.2 "FLSA" "Force Capital **Preamble** "Fifth Earn-Out Period" Section 1.9.2(b)(v) "Fourth Earn-Out Period" Section 1.9.2(b)(iv) "General Release" Section 1.3.2(a)(v) "Indemnified Taxpayer" Section 10.2.1 "Indirect Members" **Preamble** "Interim Financials" <u>Section 2.6.1</u> "Jon" **Preamble** "Letter of Transmittal" and "Letters of Transmittal" Section 1.12.2(a) "Liability Cap" Section 9.4.1 "Liability Policies" Section 2.27 "Licenses" Section 2.14.6 "Lockup Obligation" Section 1.9.2(h) "Materiality Qualifier" Section 6.1 "Member Suretyship Obligation" Section 5.6 Section 10.3.2 "Member-Prepared Returns" "Members' Representative's Earn-Out Position" Section 1.9.2(e) "Members' Representative's Position" Section 1.11.3 "Members' Representative" Section 11.2.1 "Members" Preamble "Membership Interest Purchase Agreement" Recitals "Merger" Recitals "Merger Consideration Dispute Expense" Section 1.11.3 "Merger Option" Recitals "Merger Subsidiary" **Preamble** "Most Recent Audited Financials" Section 2.7 "Most Recent Balance Sheet" <u>Section 2.6.1</u> "Most Recent Balance Sheet Date" Section 2.6.1

Term Location "Most Recent ECS Aviation Balance Sheet" Section 2.6.1 "Most Recent Financials" Section 2.7 "Most Recent Financials Balance Sheet" Section 2.7 "Most Recent Financials Balance Sheet Date" Section 2.7 "Multiemployer Plan" Section 2.17(a) "Net EBITDA" Section 1.9.2(b)(ix) "Net Working Capital Excess" Section 1.10.4 "Net Working Capital Target Amount" Section 1.10.3(b) Recitals "Non-Competition Obligation" Section 1.9.2(h) "Non-Entity Members" Preamble "Objection Notice" Section 1.9.2(h) "Operating Agreement" Section 1.5.2 "Option Agreement" Recitals "Option Consideration" Section 1.9.1(b) "Option Holders" Section 1.9.1(b) "Option Pre-Closing Period" Section 4.2.3 "Option Value" Section 1.15 "Outside Date" Section 8.1.4 "Parent" Preamble "Parent Indemnified Person" Section 9.1 "Parent's Earn-Out Position" Section 1.9.2(e) "Parent Financial Statements" Section 3.7 "Parent's Position" Section 1.11.3 "PBGC" Section 2.17(e) "PCBs" Section 2.18.8 "Per Unit Closing Date Merger Consideration" Section 1.9.1(a) "Per Unit Earn-Out Merger Consideration" Section 1.9.2(a) "Phase I ESA(s)" Section 4.9.1 "Phase II ESA(s)" Section 4.9.2 "PP&E Excess" Section 1.10.4 "PP&E Target Amount" Section 1.10.3(a) "Predecessor" Section 2.1.2 "Pre-Closing Period" Section 4.2.1 "Preferred Units" Recitals "Public Reports" Section 3.6 "Purchase Price Adjustment" Section 1.11.4(a) "Real Property" Section 2.12.1 "Real Property Leases" Section 2.12.1 "Receivables" Section 2.10 "SEC" Section 2.15.4(d) "Second Earn-Out Period" Section 1.9.2(b)(ii) "Sister Subsidiary" <u>Preamble</u> "Sixth Earn-Out Period" Section 1.9.2(b)(vi) "Straddle Period" Section 10.3.2 "Straddle Return" Section 10.3.2 "Strauss" Preamble "Subsequent Merger" Recitals "Subsidiary" and "Subsidiaries" Recitals "Surety Bonds" Section 2.31.1 "Surviving Entity" Section 1.1 "Tangible Net Worth Excess" Section 1.10.4 Section 1.10.3(c) "Tangible Net Worth Target Amount" "T&D" Recitals Section 5.2.1 "Territory" "Third Earn-Out Period" Section 1.9.2(b)(iii) "Third Party Claim" Section 9.5.1 "Total Parent Shares" Section 1.9.1(a)(i) "Transfer Restrictions" Section 5.5(a) "Unaudited Monthly Financials" Section 4.8

<u>Term</u>	Location
"Unaudited Quarterly Financials"	Section 4.8
"Underlying Leases"	<u>Section 2.12.1</u>
"Units"	<u>Recitals</u>
"WARN"	<u>Section 2.25.2</u>
"Work-In-Process"	Section 2.19.5

12.3 Rules of Construction. Except as otherwise explicitly specified to the contrary, (a) each reference to a Section, Exhibit or Schedule means a Section of, or Schedule or Exhibit to this Agreement, unless another agreement is specified, (b) the word "including" will be construed as "including without limitation," (c) references to a particular statute or regulation include all rules and regulations thereunder and any predecessor or successor statute, rules or regulation, in each case as amended or otherwise modified from time to time, (d) words in the singular or plural form include the plural and singular form, respectively, (e) references to a particular Person include such Person's successors and assigns to the extent not prohibited by this Agreement and (f) all pronouns and any variations thereof refer to the masculine, feminine or neuter singular or plural as the identity of the Person or Persons may require. The terms "hereof", "herein", "hereunder", "hereto" and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement and not to any particular provision of this Agreement. The word "or" shall not be exclusive. All references herein to "dollars" or "\$" are to United States dollars. Any accounting term used in this Agreement shall have, unless otherwise specifically provided herein, the meaning customarily given such term in accordance with GAAP and all financial computations hereunder will be computed, unless otherwise specifically provided herein, in accordance with GAAP consistently applied. All references herein to any period of days shall mean the relevant number of calendar days unless otherwise specified. Whenever any action must be taken hereunder on or by a day that is not a Business Day, then such action may be validly taken on or by the next day that is a Business Day. All references herein to a "party" or "parties" are to a party or parties to this Agreement unless otherwise specified. The phrases "date of this Agreement," "date hereof" an

[signature pages follow]

IN WITNESS WHEREOF, each of the undersigned has executed and delivered this Agreement, or caused this Agreement to be duly executed on its behalf by its officer thereunto duly authorized, as of the date first set forth above.

MASTEC, INC.

By: /s/ Jose R. Mas Name: Jose R. Mas Title: CEO

ECS Acquisition Merger Subsidiary I, Inc.

By: $\frac{\text{/s/ Jose R. Mas}}{\text{Jose R. Mas}}$

Title:

ECS Acquisition Merger Subsidiary II, LLC

By: /s/ Jose R. Mas
Name: Jose R. Mas

Title:

EC SOURCE SERVICES, LLC

By: Force Capital Partners, LLC

By: /s/ Casey Maslonka

Name: Casey Maslonka
Title: Manager

FORCE CAPITAL PARTNERS, LLC

By: /s/ Casey Maslonka

Name: Casey Maslonka

Title: Manager

FCP INVESTMENTS, LLC

By: /s/ Casey Maslonka

Name: <u>Casey Maslonka</u>
Title: <u>Manager</u>

/s/ Jon Maslonka

Jon Maslonka

/s/ Justin Campbell

Justin Campbell

/s/ Martin Maslonka

Martin Maslonka

Signature page to Merger Agreement

By your signature below, you represent that you are not legally married as of the date first	st set forth above.
	/s/ Casey Maslonka
	Casey Maslonka

\(\frac{/s/\text{ Sidney Straus}}{\text{Sidney Strauss}}\)

Signature page to Merger Agreement

EXHIBIT G

InfraSource and Option Holder Indemnification

- 1. Any Loss related to any relationship between or among any member, director, consultant or employee of any Company or any of their respective Affiliates with InfraSource Transmission Services Co., an Arizona corporation ("InfraSource Co."), InfraSource Transmission Services, Inc., a Delaware corporation ("InfraSource Inc.") or any of Infra Source Co.'s or Infra Source Inc.'s respective Affiliates, including pursuant to or in connection with any of the following:
 - (a) Noncompetition Agreement, dated January 27, 2004, between Jon and Infrasource, Inc., as amended by Amendment No.1 to Noncompetition Agreement, dated January 6, 2006, and the Amended and Restated Non-Competition and Management Agreement, effective February 23, 2007, between Jon, InfraSource Co. and InfraSource, Inc.
 - (b) Employment Agreement, dated July 22, 2002, between Strauss and InfraSource Co., as amended by the Amended and Restated Employment and Management Agreement, effective February 23, 2007, between Strauss, InfraSource Co. and InfraSource, Inc.
 - (c) Noncompetition Agreement, dated January 27, 2004, between Campbell and InfraSource Inc., as amended by Amendment No.1 to Noncompetition Agreement, dated January 2006.
 - (d) Noncompetition Agreement, dated January 27, 2004, between Martin Maslonka and InfraSource, Inc., as amended by Amendment No. 1 to Noncompetition Agreement, dated January 6, 2006.
- 2. Any Loss related to any present Option Holder as of the day hereof or future Option Holder (other than the payment of the Option Consideration to such Option Holder at the Effective Time if the Merger occurs) or any former Option Holder (including, Elizabeth Higgins and Jason Combs).

EXHIBIT H

Software Booked as PP&E

- 1. Primavera P6 Enterprise Project Portfolio Management V6.2.1
- 2. Primavera Contract Manager 12.1
- 3. All other Software acquired by any of the Companies from third parties in the ordinary course of business and booked as PP&E

Consent of Independent Valuation Firm

We hereby consent to the references to Houlihan Capital Advisors, LLC ("Houlihan") and to our valuation report dated March 31, 2011 (our "Report") relating to the estimation of fair value of certain auction rate securities held by MasTec, Inc. (the "Company") as of March 31, 2011, including the use of information contained within our Report in the Company's Form 10-Q for the quarter ended March 31, 2011 and to the incorporation by reference of such information from the Company's Form 10-Q for the quarter ended March 31, 2011 in the following Registration Statements: Form S-8 Nos. 333-139996, 333-112010, 333-105781, 333-105516, 333-38932, 333-77823, 333-47003, 333-38940 and 333-30647 and Form S-3 Nos. 333-158502, 333-142083, 333-133252, 333-46067 and Form S-4 No. 333-170834.

We also consent to the reference to our firm under the caption "Experts" in the above-referenced Registration Statements.

/s/ Houlihan Capital Advisors, LLC Houlihan Capital Advisors, LLC

May 2, 2011

CERTIFICATIONS REQUIRED BY SECTION 302(A) OF SARBANES-OXLEY ACT OF 2002

I, Jose R. Mas, certify that:

I have reviewed this quarterly report on Form 10-O of MasTec, Inc.;

Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;

Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;

The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:

- a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
- b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- d) Disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.

The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

- a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 4, 2011

/s/ Jose R. Mas

Jose R. Mas Chief Executive Officer (Principal Executive Officer)

CERTIFICATIONS REQUIRED BY SECTION 302(A) OF SARBANES-OXLEY ACT OF 2002

I, C. Robert Campbell, certify that:

I have reviewed this quarterly report on Form 10-Q of MasTec, Inc.;

Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;

Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;

The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:

- a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
- b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- d) Disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.

The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

- a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 4, 2011

/s/ C. Robert Campbell

C. Robert Campbell
Chief Financial Officer
(Principal Financial and Accounting Officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of MasTec, Inc. (the "Company") on Form 10-Q for the period ended March 31, 2011 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Jose R. Mas, Chief Executive Officer of MasTec, Inc., certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of my knowledge:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 4, 2011

/s/ Jose R. Mas

Jose R. Mas Chief Executive Officer (Principal Executive Officer)

The certification set forth above is being furnished as an exhibit solely pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 and is not being filed as part of the Quarterly Report on Form 10-Q for the period ended March 31, 2011, or as a separate disclosure document of the Company or the certifying officers.

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of MasTec, Inc. (the "Company") on Form 10-Q for the period ended March 31, 2011 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, C. Robert Campbell, Chief Financial Officer of MasTec, Inc., certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of my knowledge:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 4, 2011

/s/ C. Robert Campbell

C. Robert Campbell
Chief Financial Officer
(Principal Financial and Accounting Officer)

The certification set forth above is being furnished as an exhibit solely pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 and is not being filed as part of the Quarterly Report on Form 10-Q for the period ended March 31, 2011, or as a separate disclosure document of the Company or the certifying officers.